Past and Future of Software Architectures for Context-Aware Systems: A Systematic Mapping Study

Cristina Roda¹, Elena Navarro¹, Uwe Zdun², Víctor López-Jaquero¹, Georg Simhandl²

¹LoUISE Research Group, University of Castilla-La Mancha, Spain

{Cristina.Roda, Elena.Navarro}@uclm.es, victor@dsi.uclm.es

²Software Architecture Group, University of Vienna, Austria

{uwe.zdun, georg.simhandl}@univie.ac.at

Abstract. There is a growing interest on context-aware systems in recent years. Context-aware systems are able to change their behaviour depending on new conditions regarding the user, the platform and the environment. These systems are evolving towards interacting with the user in a transparent and ubiquitous manner, especially by means of different types of sensors, which can gather a wide range of data from the user, the platform the user is interacting with, and the environment where such interaction is taking place. It is worth noting that the software architecture of a system is a key artefact during its development and its adaptation process. Hence, the definition of the software architecture becomes essential while developing context-aware systems since it should reflect how the context is tackled for adaptation purposes. With the aim of studying this issue, we have designed and conducted a systematic mapping study to provide an overview about the different architectural approaches used in context-aware systems. One of the main findings of this study is that there are not many software architecture proposals that deal with context-awareness in an explicit way during the adaptation process. It was also detected that there are Human Computer Interaction (HCI) works that focus on context-aware adaptations but neglect partially or completely any possible change in the system architecture during the adaptation process. Due to this, we perceived a need to analyse what research works highlight the use of context and its relationship to the software architecture in existing context-aware systems. Therefore, this mapping study attempts to bridge the gap between Software Architecture and HCI in order to align the adaptation at the architectural level (changes in the configuration of architectural components) and at the HCI level (changes in the interaction modality or the user interface in general).

Keywords: human-computer interaction, user, platform, environment, quality of adaptation, quasi-gold standard.

1 Introduction

In recent years, we can observe a trend aiming at transforming the environment in which the user interacts with a software system into a smart environment so that the user can interact with the system in a transparent and seamless manner. In this sense, *context-awareness* emerges as a suitable approach for developing such systems since it provides them with flexibility, adaptability and capability of acting autonomously on behalf of users [22]. Context-aware systems pay attention to the *context* surrounding the user to adapt their behaviour to new conditions. Such context usually refers to the user, the platform he/she is interacting with, and the environment where such interaction is being done. As stated in [3], these three dimensions (*User, Platform* and *Environment*) represent the main cornerstones for defining the context.

As Perry and Wolf [35] already realized in 1992, the *Software Architecture* (SA) of every software system and its specification should be carefully considered during design. They defined the SA as "the selection of architectural elements, their interactions and the constraints on those elements and their interactions necessary to provide a framework in which to satisfy the requirements and serve as a basis for the design". Therefore, SA becomes a key artefact whenever it is necessary to develop a system that offers complex capabilities.

Given the importance of context and its different dimensions in context-aware systems for adapting its behaviour, as well as the importance of the SA while designing a system, it could be thought that most of the proposals in the area have been defined exploiting SA mechanisms to provide such context-aware adaptations. However, we did not find many software architecture proposals that deal with adaptation based on the context of use. Most of the architectural proposals focus on some specific kind of adaptation, but mostly from the point of view of software architecture and the possible configurations of its components. Consequently, such architectural approaches do not deal with context-awareness in an explicit way during the adaptation process, usually not considering the user at all and hence neglecting the adaptation from the point of view of Human-Computer Interaction (HCI). There are also HCI works that focus on context-aware adaptations, but they do not pay much attention to the required software architecture support.

As a result of these observations, we perceived a need to analyse what research works highlight the use of context and its relationship with the software architecture in existing context-aware systems. To the best of our knowledge, no literature study exists yet that carries out such analysis, so that the execution of a systematic mapping study emerged naturally. As Kitchenham and Charters state [29], these type of studies provide a broad review of primary studies in a specific topic area that aims at identifying what evidence is available on the topic. Thus, by performing such a mapping study, we will be able to provide an overview about the different approaches used in *context-aware systems* in order to align the adaptation at the architectural level (changes in the configuration of architectural components) and at the *HCI level* (changes in the interaction modality or the user interface). Moreover, we will also identify the gaps existing till now with regard to *the deficiencies and weaknesses of systems when dealing with context and its different dimensions*. These findings will help the research community by providing guidelines for modelling, retrieving and managing context and its different dimensions. In addition, guidelines to know how to consider such context by the software architecture will be also provided in order to produce proper adaptation experiences to the final users.

This paper is structured as follows. Section 2 presents the *research methodology* applied to perform this systematic mapping study. Next, Section 3 describes how the *data collection* for this mapping study was carried out. Section 4 shows the *results* obtained from the search process of the mapping study by answering the research questions previously established. Section 5 presents the *discussion* derived from the analysis of the different results. Our main *conclusions* are presented in Section 7. Finally, the *selected papers* of this mapping study are listed in Appendix A.

2 Research Methodology

As aforementioned, this article presents a systematic mapping study that focuses on providing an overview about the state of the art on software architecture concepts when used for developing context-aware systems. A *systematic mapping study* [36] is an evidence-based form of secondary study that provides an overview of a research area, identifying common publication venue types (e.g. conference, journal or workshop), quantitative analyses (e.g. number of published studies per year), and research findings in the investigated research field. Kitchenham and Charters [29] also provide a definition of a systematic mapping study, considering it as broad review of primary studies in a specific topic area that aims to identify what evidence is available on the topic.

This mapping study has followed several guidelines [50][7][29] for conducting systematic studies, especially in the field of software engineering. Moreover, we have also taken into consideration the work by Zhang et al. [55][56] while performing our search strategy, since it was driven by a concept introduced by these authors: the Quasi-Gold Standard. The *Quasi-Gold Standard* [55] refers to a set of known studies extracted from related venues, such as domain-specific conferences and journals perfectly recognized by the research community in the field, and for a given period of time. Our mapping study is composed of the following six stages, being each one of them described separately in the following sections:

- 1. The *rationale* for conducting the mapping study (Section 2.1).
- 2. The establishment of the *Research Questions* (RQs) that this mapping study answers (Section 2.2).
- 3. The search process we have followed relying on the Quasi-Gold Standard approach (Section 2.3).
- 4. The *categorization* we have followed to classify the results obtained (Section 2.4).
- 5. How the *data collection* has been done applying the selection criteria (Section 3).
- 6. The *results* obtained along with the answers to the RQs (Section 4).

2.1 Mapping Study Rationale

The first step while performing a systematic study is to define the main reason that has led to carry it out. In this case, we did not find much evidence in the literature about software architecture proposals that provide support for adaptation based on the context of use. On the one hand, most of the architectural frameworks are centred on supporting some kind of adaptation, but mostly from the point of view of the software architecture and the possible configurations of its components. Such proposals do not manage the *context-awareness* explicitly during the adaptation process, and hence they neglect the user completely as well as the adaptation from the point of view of HCI. On the other hand, there are HCI studies focused on context-aware adaptations which neglect partially or completely any possible change in the system architecture during the adaptation process to support such adaptation. In this sense, we have detected *a need to analyse which research works highlight the use of context and its relationship with the software architecture in existing context-aware systems*.

By doing so, we will be able to provide an overview about the different approaches used in *context-aware systems* in order to align the adaptation at both *architectural level* (changes in the configuration of architectural components) and

HCI level (changes in the interaction modality or the user interface in general). Moreover, we will also identify the existing gaps up to now, i.e. *what are the deficiencies and weaknesses of systems when dealing with context and its different dimensions*. These findings will help the research community to define guidelines for modelling, retrieving and managing context and its different dimensions, as well as to establish how the software architecture may support the context in order to provide the final user with a good adaptation experience. Therefore, the *main objectives* of our study are the following:

- **Objective 1.** Review *software architecture concepts*, such as designs, frameworks, components, styles, patterns, infrastructures and specifications, proposed for designing and building *context-aware systems* with the aim of understanding the relationships between them.
- **Objective 2.** Review how *context-based adaptations* are considered by those software architecture concepts proposed for context-aware systems.
- **Objective 3.** Review the *maturity* of the different software architecture concepts proposed for context-aware systems, as well as the maturity of the methods for assessing context-based adaptations.

2.2 Research Questions

Based on the objectives from Section 2.1, we defined the following *Research Questions* (RQs), which will be answered throughout this paper:

• **RQ 1.** Which *architectural designs, frameworks, components, styles, patterns, infrastructures* and *specifications* are used in the existing context-aware systems? (Related to **Objective 1**)

Rationale. Architectural designs, frameworks, components, styles, patterns, infrastructures and specifications (i.e., *software architecture concepts*) have been used as artefacts to guide the development of software products and establish a certain level of claimed quality. However, there is no overview of primary studies on defining and/or exploiting those software architecture concepts in relation to the existing context-aware systems. In this article, we aim at offering such an overview in order to help researchers and practitioners to analyse which of those architectural topics are the most relevant and the most used.

• **RQ 2.** How do the software architecture concepts proposed for context-aware systems consider the three main *dimensions of context* (User, Platform and Environment) to support adaptation? (Related to **Objective 2**)

Rationale. In the field of context-aware computing, there are many different proposals for modelling the context. However, as stated in [3], there is a consensus that the three main cornerstones for defining the context are: *User*, *Environment* and *Platform*. Therefore, we want to discover how each dimension of context is addressed and taken into account within the different architectural proposals to carry out the adaptations based on these three dimensions of context. In this sense, we propose three different sub-research questions where each one of them refers to a different dimension of context for the sake of clarity.

• **RQ 2.1.** How do the software architecture concepts proposed for context-aware systems consider the *user dimension of context* to support adaptation?

Rationale. The *user* dimension here refers to all characteristics related to the user of the system: preferences regarding the user interface (colour, font size, presentation, content, etc.), profile features (name, age, job, etc.), physical impairments (blind, single-handed, etc.), health condition, physiological data (heart rate, blood pressure, etc.), skills, expertise, contacts, etc. Thus, we are interested in studying the *user* dimension of context as a separate concern to identify how the different software architecture concepts proposed for context-aware systems are addressing it to carry out adaptation.

• **RQ 2.2.** How do the software architecture concepts proposed for context-aware systems tackle the *platform dimension of context* to support adaptation?

Rationale. As stated in [4], this particular dimension of context is related to the software side of a system, such as the operating system, the technology infrastructure represented by a well-defined programming language, applications supported, etc. However, the platform is not only limited to the software, but also it refers to hardware elements, such as the screen size of the monitor, a temperature sensor, or the network characteristics (bandwidth, speed, etc.). Therefore, we want to analyse, separately from the other context dimensions, how the different software architecture concepts proposed for context-aware systems tackle the platform dimension of context to support adaptation.

• **RQ 2.3.** How do the software architecture concepts proposed for context-aware systems take into account the *environment dimension of context* to support adaptation?

Rationale. Along with the user and platform, the *environment* is also considered as one of the main dimensions of context. This dimension includes the ambient noise, light level, temperature, weather, date, nearby people, etc. in the location where the interaction between the user and the system is taking place. Thus, similarly to the other dimensions of context, we are interested in studying this context dimension too with the aim of identifying how the different software architecture concepts proposed for context-aware systems are addressing it to carry out adaptation.

• **RQ 3.** What *evaluations* have been performed so far for validating the different *context-aware adaptations* proposed for context-aware systems? (Related to **Objective 3**)

Rationale. Every software artefact, such as a user interface or a software architecture, should guarantee a certain degree of quality when delivered to prevent the final user from rejecting it. Therefore, context-aware adaptation should also ensure a certain degree of quality whenever they are applied. In this sense, proposals supporting context-aware adaptation should provide some mechanisms to estimate the degree of quality of these adaptations, i.e. to evaluate whether context-aware adaptations are perceived positively by the final user or the entity the adaptation was designed for. With this purpose, we aim to discover to what extent the evaluation of context-aware adaptation in context-aware systems has been tackled in the literature. By doing so, we will be able to estimate the *maturity* of the proposals for adaptation.

• **RQ 4.** What *evaluations* have been performed so far for validating the different *software architecture concepts* proposed for context-aware systems focused on adapting the system using the surrounding context (user, platform and environment)? (Related to **Objective 3**)

Rationale. We want to find out what evaluations have been carried out until now to validate the software architecture concepts proposed for context-aware systems. These evaluations may range from informal evaluations, just showing some examples, to more formal empirical evaluations. In this sense, we want to categorize the different software architecture concepts, depending on the type of evaluation presented. This will enable us to determine the *maturity* of such software architecture concepts for developing context-aware systems.

2.3 Search Process Based on the Quasi-Gold Standard

There have been two main reasons that have led us to use the systematic search strategy presented by Zhang et al. [55][56]: first, i) to identify the most relevant works in the literature with regard to our topic in an objective and systematic way, and second ii) to properly assess the search performance. These authors stated that most of the systematic studies in Software Engineering apply a subjective search process based just on the expertise and knowledge of the team on the research topic and tested on supposedly "well-known" studies previously gathered by such team to assess the search performance. Zhang et al. claimed that this set of "well-known" samples depends on the knowledge of the searchers on the particular topic and, thus, it cannot replace the *Gold Standard* (i.e. the known set of identified primary studies in a collection according to the definition of the Research Questions) for evaluation, since it is not possible to have a complete set of primary studies before the execution of the systematic study. Therefore, these authors introduced the concept *Quasi-Gold Standard* (QGS) [55] that refers to a set of known studies extracted from related venues, such as domain-specific conferences and journals perfectly recognized by the research community in the field, for a given period of time. Thereby, the QGS acts as a Gold Standard substitute, but with the only restriction of *where* (venues) and *when* (during a period of time).

In this sense, with the aim of providing *objectivity* to the search process, Zhang et al. integrate the QGS method into the systematic search process that relies on the analysis of information from the works collected, i.e. from the QGS, instead of using a subjective set of studies which depends on the perceptions of the people who carry out the search. Besides the support for objectivity, another benefit of applying the QGS method is that it can be used to evaluate the search strategy performance, that is, to evaluate the search results in order to guarantee a certain quality level of the automated search. Zhang et al. propose to carry out this assessment by means of the *Sensitivity* which is defined as the proportion of relevant studies retrieved for a given topic:

$$Sensitivity = \frac{Number of relevant studies retrieved}{Total number of relevant studies} \cdot 100\%$$

However, as the gold-standard is unknown, the sensitivity cannot be calculated so that the QGS is used to measure the *Quasi-sensitivity* of the search universe. Fig. 1 depicts the different steps that integrate this systematic search strategy. The first step consists in *identifying the most relevant venues* (both conferences and journals) regarding our research topic in order to *discover the QGS* (Step 2), and also *identifying the engines* (libraries and databases) that will be used for our *automated search* (Step 4). In step 3, *the search strings are built* to use them as input for the automated search. These strings can be defined either by researchers, thanks to their knowledge on the matter or, more objectively, they can be elicited from the studies in the QGS by means of the analysis of word frequency or content. In step 5, *the search results*

from the automatic search are evaluated regarding the quasi-sensitivity and compared with a threshold previously established. If the search performance is found acceptable (usually when quasi-sensitivity \geq 80%), then the results obtained by the automated search are combined with the QGS, finishing the search process. Otherwise, the flow must go back to step 3 to refine the search strings until the search performance reaches an acceptable level. In Section 3.2, each one of these steps are explained in more detail regarding our systematic study on software architectures for context-aware systems.



Fig. 1. Workflow of the systematic search strategy (adapted from [56])

2.4 Categorization of Results

Several well-defined and widespread taxonomies have been used in this work to classify the selected papers obtained from our search process. One of them is the categorization originally proposed by Wieringa et al. [52] for classifying papers regarding the *research type*. Table 1 shows these research types along with its description proposed by Kuhrmann et al. [32]. Note that one result can be categorized as more than one research type.

Table 1. Re	search types	categorization
-------------	--------------	----------------

Research type	Description
Evaluation research	It is shown how the technique is implemented in practice (solution implementation) and which are the consequences of the implementation in terms of benefits and drawbacks (implementation evaluation). This also includes identifying problems in industry. An example for evaluation research is reporting of an industrially hosted case study.
Solution proposal	A solution for a problem is proposed. The solution can be either novel or a significant extension of an existing technique. The potential benefits and the applicability of the solution are shown by a small example or a good line of argumentation.
Validation research	Techniques investigated are novel and have not yet been implemented in practice. Techniques used are, for example, experiments, that is, work done in the lab.

Philosophical papers	These papers present a new way of looking at existing things by structuring the field in the form of a taxonomy or conceptual framework.
Opinion papers	These papers express the personal opinion of somebody whether a certain technique is good or bad or how things should be done. They do not rely on related work or research methodologies.
Experience papers	These papers explain on what and how something has been done in practice. It should be the personal (informal) experience of the author

Another categorization, which has been applied to the selected papers when analysing the RQ 4, is the one presented in Table 2 to classify the *evaluation strategies* presented in the papers. These evaluations include the empirical strategies proposed by Wohlin et al. [54], such as *Survey*, *Case study* and *Experiment*, as well as the evaluation by means of an *Example* when no formal empirical strategy is applied. In this last case, the paper has not been classified neither as a Validation nor Evaluation research because of the lack of formalism and empirical evidence those papers presented.

Table 2. Evaluation strategies categorization

Evaluation strategies	Description
Survey	A <i>survey</i> is a system for collecting information from or about people to describe, compare or explain their knowledge, attitudes and behaviour by means of interviews or questionnaires. Surveys are conducted when the use of a technique or tool already has taken place or before it is introduced.
Case study	A <i>case study</i> is an empirical enquiry that draws on multiple sources of evidence to investigate one instance (or a small number of instances) of a contemporary phenomenon within its real-life context, especially when the boundary between phenomenon and context cannot be clearly specified.
Experiment	An <i>experiment</i> is a formal, rigorous and controlled investigation. Based on randomization, different treatments are applied to or by different subjects, while keeping other variables constant, and measuring the effects on outcome variables. During an experiment, quantitative data is collected and then <i>statistical methods</i> are applied.
Example	When no formal empirical strategy is applied to evaluate the proposal, but there is some kind of illustration.

Finally, we have also classified the results with regard to the following criteria in order to facilitate the later analysis and discussion. Note that it is not possible to predefine the different categories that belong to each one of these criteria before analysing the papers.

- The *country* denotes the country of the institution of the first author of the paper. It is a single value per paper.
- The *venue* indicates whether the paper is a Journal, a Conference or a Workshop. It is a single value per paper.
- The *research topic* represents the main research topic throughout the paper, e.g. context modelling, context inference, service delivery, context-aware applications development, UI adaptation, etc. A paper can be classified according to more than one topic.
- The *application domain* represents the main application domain/s of the proposal, e.g. healthcare, e-learning, smart home, etc. Not every paper is oriented to an application domain, but it is possible to have more than one domain for some papers.
- The question *what is adapted?* indicates which element of the context-aware system is the subject of the adaptation, e.g. services, UI, architecture configuration, content, presentation, navigation, etc. Not every paper expresses what is adapted. In contrast, it is possible to have more than one element that is adapted during the adaptation process in the same work.
- The *software architecture concept* (RQ 1) indicates the type of software architecture concept employed, e.g. serviceoriented-architecture (SOA), multi-agent system, multi-layered architecture, etc. Every paper selected should describe some software architecture concept in the proposal, or even more than one.
- The *user dimension of context* (RQ 2.1) determines the different characteristics of the user considered in the proposal for performing the adaptation of the system, e.g. preferences, profile, physiological data, social profile, abilities,

disabilities, etc. Not every paper deals with this dimension of context. In contrast, one work can present more than one user characteristic.

- The *platform dimension of context* (RQ 2.2) determines the different characteristics of the platform considered in the proposal for performing the adaptation of the system, e.g. network, bandwidth, battery level, screen size, etc. Not every paper deals with this dimension of context. In contrast, one work can present more than one platform characteristic.
- The *environment dimension of context* (RQ 2.3) determines the different characteristics of the environment considered in the proposal for performing the adaptation of the system, e.g. location, noise, light, temperature, weather, date, nearby people, etc. Not every paper deals with this dimension of context. In contrast, one work can be classified according to different characteristics.
- The *other dimensions of context* (RQ 2) determines if the proposal makes use of other dimensions of context different from User, Platform or Environment. Not every paper deals with other dimensions of context. In contrast, one work can be classified according to more than one category for this RQ. For example, some other dimensions of context are the following:
 - *Historical data*, such as the interaction history of the user. It is usually used to know the evolution of the user and hence making changes and adaptations in the system that improve the user experience
 - *User activity*, refers to the activity the user is performing at a specific moment of time, e.g. sleeping, cooking, brushing the teeth, etc.
 - *Interaction task*, refers to the task of interaction the user is making with the system, e. g. opening a new window, clicking a link, typing in the keyboard, etc.

3 Data Collection

In this section, the procedure for collecting all data of this mapping study is presented. Section 3.1 presents the different criteria used to select only the relevant results from the search process. The related venues and engines identified for this study are described in Section 3.2.1. Section 3.2.2 presents the most relevant articles that define the QGS. Section 3.2.3 depicts the definition of the search string for the automated search. Section 3.2.4 presents the outcomes obtained from the automated search. Finally, Section 3.2.5 exhibits the evaluation of the search performance by means of the quasisensitivity calculation.

3.1 Selection Criteria

Generally, the search process in a systematic study returns a great deal of results, but not all of them are usually considered relevant. For this aim, we have defined the following criteria to filter out the papers returned and select only the relevant ones.

The inclusion criteria are:

- **I1**. The study presents any kind of software architecture concept that supports the development, design, etc., of context-aware applications.
- I2. The study reports an example or similar that shows how a particular software architecture concept supports context-aware adaptations.
- **I3**. The study presents some type of evaluation for validating a specific software architecture concept regarding a particular context-aware system.
- I4. The study presents some type of evaluation for assessing the adaptation in context-aware systems considering the supporting architecture.

On the other hand, the *exclusion criteria* are:

- E1. The study is not written in English.
- E2. The study is not related to Computer Science research field at all.
- E3. The study is related to Computer Science research field, but not to Software Engineering or Human-Computer Interaction fields, or to a software architecture concept applied on context-aware systems.
- E4. The study is an older publication from the same authors about the same approach.

- E5. The study is a short one, less than three pages.
- E6. The study is a very large one, e.g. a PhD thesis or a book.
- E7. The study is a secondary or tertiary study (e.g. a systematic mapping study or literature review), not a primary one.
- E8. The study is a non-peer reviewed publication such as a technical report, editorial note, preface, index, introduction, presents a special/theme issue, or it is unreferenced, illegible or inaccessible (e.g., paper is not available through the search engines used).
- **E9**. The study has been published after 2017.

Regarding the last exclusion criterion, note that 2017 was chosen as end date because the search was conducted at the beginning of 2018. The query could return inconsistent results whether it was rerun during the process if 2018 were also considered.

3.2 Quasi-Gold Standard-Based Search Strategy

This section details the steps of the search strategy we have followed to find the primary studies to be used in this mapping study. We would like to remind that we have based our search strategy in the Quasi-Gold Standard concept presented by Zhang et al. [56].

3.2.1 Identifying related venues and engines

The first stage of the search process is related to the identification of the most relevant *related venues* used in the next step for establishing the QGS (see Section 3.2.2), as well as the identification of the *engines* for the automated search in step 4 (see Section 3.2.4).

Regarding the *related venues*, they are usually selected without difficulty considering both the research questions previously established and the expertise and knowledge of the researchers on the subject under study. However, for this mapping study, this selection was challenging because our study is focused not only on one research field, but on two research fields clearly different: Software Engineering (SE) and Human-Computer Interaction (HCI). On the one hand, we are interested in *architecture* concepts, such as designs, frameworks, components, styles, patterns, infrastructures and specifications (derived from the research questions described in Section 2.2), which are obviously connected to the area of Software Architecture (SA). On the other hand, we are only interested in those architectures designed for *context*aware systems, which are directly connected to the Pervasive and Ubiquitous Computing area in the HCI field. Although our first thought was to include conferences and journals related to both research fields, after a quick review of the literature and a thorough discussion, we realized that most of the relevant works related to context-aware systems were published in specialized venues of HCI and Pervasive and Ubiquitous Computing for establishing the QGS. Thereupon, by considering all the aforementioned issues, we finally decided to set as related venues those connected to HCI and Pervasive and Ubiquitous Computing. Moreover, those works published in venues related to SA and SE fields will be discovered during the automatic search stage given that they are also indexed in the selected search engines. Table 3 shows the final set of relevant venues selected for establishing the OGS, along with their associated publisher, digital library or engine. Note that five different venues focused on HCI and Pervasive Computing have been identified, namely the journals Human-Computer Interaction, IEEE¹ Pervasive Computing, Personal and Ubiquitous Computing, and User Modeling and User-Adapted Interaction; as well as the ACM^2 International Joint Conference on Pervasive and Ubiquitous Computing (UbiComp). Note that this last conference, UbiComp, is the result of merging the most renowned conferences in the field during the last few decades, such as PERVASIVE³, UbiComp⁴ and HUC⁵. These five venues have been selected since they are considered the most relevant and referenced ones with regard to HCI and Pervasive Computing.

Table 3. Related venues identified

Type of venue

Venue

Publisher

Digital libraries/Engines

¹ Institute of Electrical and Electronics Engineers

² Association for Computing Machinery

³ International Conference on Pervasive Computing

⁴ International Conference on Ubiquitous Computing

⁵ International Symposium on Handheld and Ubiquitous Computing

Journal	Human-Computer Interaction	Taylor & Francis	ACM Digital Library Scopus
	IEEE Pervasive Computing	IEEE Computer Society	ACM Digital Library IEEE Xplore Scopus
	Personal and Ubiquitous Computing	Springer London	ACM Digital Library Springer Link Scopus
	User Modeling and User- Adapted Interaction	Springer Netherlands	ACM Digital Library Springer Link Scopus
Conference	ACM International Joint Conference on Pervasive and Ubiquitous Computing (UbiComp)	ACM	ACM Digital Library Springer Link Scopus

Regarding the selection of *engines* for the automated search, Zhang et al. [56] stated that it depends on several factors, such as whether they publish the selected the venues or not, the degree of coverage among them, and also the availability and accessibility of the different articles through such engines. Furthermore, they observed that ACM Digital Library and IEEE Xplore constitute a must-have when talking about the engines to be selected for any automatic search for systematic studies in SE. Although our initial intention was choosing the minimum set of search engines for avoiding as much overlapping as possible, we noticed that many works were only indexed by one of these search engines, especially those indexed by Scopus and Springer Link. Therefore, for the sake of completeness, our final set of selected search engines include *ACM Digital Library* and *IEEE Xplore*, as recommended by Zhang et al., as well as *Scopus* and *Springer Link*. Note that when searching with ACM, we have chosen the option *Guide to Computing Literature* instead of *ACM Full-Text Collection* as the former includes a wider set of articles. Table 4 shows the search engines finally chosen for conducting the automated search (see Section 3.2.4).

Table 4. Selected	l engines	for the	automated	search
-------------------	-----------	---------	-----------	--------

Selected search engines	Web reference
ACM Digital Library	[2]
IEEE Xplore	[28]
Scopus	[17]
Springer Link	[48]

Zhang et al. stated [55,56] that authors can also employ other search engines not directly derived from the related venues while performing a systematic study, but, in such a case, the QGS is *only* valid for the assessment of the search performance through those engines inferred from the nominated venues.

3.2.2 Establishing the Quasi-Gold Standard

The second stage of the search process is the establishment of the QGS. Zhang et al. [55,56] recommend a manual search through the related sources identified in the previous step for finding the most relevant and most referenced works regarding our research topic. In this manner, all these relevant papers will be part of our QGS. However, this manual process is very time-consuming since all articles published in such venues have to be analysed one by one. To facilitate such work, we propose a *semi-manual search* which consists in using a generic search engine, such as Google Scholar [19], to identify a good initial set of studies related to our topic, and then applying the snowballing technique, screening one by one the resulting articles, to find the most relevant and most cited works.

The semi-manual search followed here is based on the guidelines for snowballing in systematic studies provided by Wohlin[53]. This author proposes among others the following tips that we have used to identify a *good initial set*, i.e. a set of papers to use for the snowballing procedure:

- i. A good initial set of studies can be identified by using, for example, Google Scholar, with the aim of avoiding bias in favour of a specific publisher.
- ii. If too many papers are found, then the most relevant and highly referenced ones should be identified.
- iii. The initial set should cover different publishers, different years and different authors.

iv. The initial set should be formulated taking into account the keywords used in the RQs.

Therefore, the semi-manual search was conducted by using Google Scholar with an initial search string composed of the keywords *architecture* and *context-aware* extracted from the RQs (see Section 2.2). Seventeen papers were obtained from this first search, ordered by relevance according to Google algorithms, including various PhD dissertations related to relevant authors on the subject, such as Dey's PhD dissertation [15], as well as some technical reports. However, these types of articles, such as PhD theses and technical reports, were not included because of our exclusion criteria (see Section 3.1). Nevertheless, some specific vocabulary and synonyms related to the terms *context* and *context-aware* were gathered. For instance, in Dey's PhD dissertation [15], some context-aware synonyms were found, such as *adaptive* [5], *reactive* [12], *responsive* [16], *situated* [26], *context-sensitive* [40] and *environment-directed* [18]. Some of these terms have been used in the construction of the search string for the automated search (see Section 3.2.3).

Afterwards, the snowballing strategy was applied, looking for articles that cite those included in the initial set (*forward snowballing* [53]), and also looking for articles cited by the works in the initial set, i.e. those which appear in the references list (*backward snowballing* [53]). Thereby, all works found were filtered by their number of references, and also by their publication venue given that we are only interested in those ones published in the venues previously selected (see Section 3.2.1). Finally, we obtained a set of *17 relevant papers* which are indeed focused on architectures for context-aware systems and hence they constitute our QGS. Table 5 presents the distribution of the relevant papers selected from the related venues. Notice that the QGS covers the publication period between *1999* and *2011* so that 1999 was selected as the starting date for the automated search (see Section 3.2.4).

Venues	Papers of the QGS
Human-Computer Interaction	[13][23]
IEEE Pervasive Computing	[20][30][39][43]
Personal and Ubiquitous Computing	[25][41]
User Modeling and User-Adapted Interaction	[9][10][11][21][24][37][49]
ACM Int. Joint Conf. on Pervasive and Ubiquitous Computing (UbiComp)	[1][46]

Table 5. Quasi-Gold Standard distribution

3.2.3 Defining the search string for the automated search

As presented in Fig. 1, the search process at this point can be carried out in two different ways in order to obtain the final search string for the automated search. On the one hand, as Zhang et al. stated [3][4], most systematic studies found in the literature in SE conduct automated search in a subjective manner, i.e. researchers build search strings for automated search based on their knowledge and experience about the research topic. On the other hand, these search strings can also be elicited in a more objective way based on the QGS (dashed lines in Fig. 1). In this sense, text mining techniques are used to discover the most frequently occurring words or sentences within the QGS. By doing so, the search strings can be elicited using such candidate search terms, producing a higher search performance rather than using a subjective search string definition.

In our case, we started with the definition of the search string in a subjective manner, only depending on our knowledge, expertise and manual findings. Subsequently, different keywords and terms repeatedly found during the establishment of the QGS were incorporated to our search string to offer a more objective perspective to the search process. Obviously, the definition of the *search string* should be as useful and accurate as possible with the aim of extracting the most relevant papers related to the scope of the study. Nevertheless, the process of defining an adequate search string is neither an easy task nor a quick one because this study is focused on analysing the different *architectures* proposed for *context-aware* systems. Moreover, as mentioned before in the RQs (Section 2.2), we are especially interested in studying the three dimensions of context: *User, Environment* and *Platform*. Therefore, our final search string should take into account all these keywords, i.e. *architecture, context-aware*, and also the dimensions of context: *user, environment* and *platform*, as well as the most common synonyms for such terms.

With regard to the term *architecture*, some synonyms were retrieved from the RQs, such as: *design*, *framework*, *component*, *style*, *pattern*, *infrastructure* and *specification*. We included all these words in the search string to find as many related works as possible but focusing on any type of architecture concept applied in context-aware systems. As aforementioned, this study deals with two different research areas, *Software Architecture* and *context-awareness*, that address the term of architecture in a different way, using different terminology. For this reason, we have used the following keywords to address the concept architecture in our search string:

architecture OR design OR framework OR component OR style OR pattern OR infrastructure OR specification

Regarding the concept *context-aware*, some synonyms have also been included in the search string as we noticed while establishing the QGS that this term appears in different forms. Some identified synonyms of context-aware were: *adaptive* [5], *reactive* [12], *responsive* [16], *situated* [26], *context-sensitive* [40] and *environment-directed* [18]. However, some of them, such as *adaptive*, *reactive*, *responsive* and *situated*, are too generic to use them as a single word. For solving this problem, we explored the literature in order to find out the most common terms used along with these generic concepts. The term *adaptive* usually is used along with *user interface*, *agent*, *application* and *wearable*; the term *reactive* is often used with *environment*; *responsive* is frequently used with the terms *environment* and *application*; and finally, the word *situated* is broadly used with *communication* and *application*, as well as the composed term *situation aware*. This led us to define the search string related to context-aware as follows:

"context-aware" OR "adaptive user interface" OR "adaptive agent" OR "adaptive application" OR "adaptive wearable" OR "reactive environment" OR "responsive environment" OR "responsive application" OR "situated communication" OR "situated application" OR "situation aware" OR "context-sensitive" OR "environment-directed"

With regard to the dimensions of context, *user*, *environment* and *platform*, some synonyms have also been used to build the search string, including several relevant words commonly used when referring to such dimensions of context:

user OR human OR device OR platform OR environment OR task

Finally, the final search string used for the automated search was defined as follows by gathering the previous ones connected by an AND operator:

	Tuere of Search stang (similar terms are meraded in craenes)
Concept	Search
Architecture	(architecture OR design OR framework OR component OR style OR pattern OR infrastructure OR specification)
Context-aware	AND ("context-aware" OR "adaptive user interface" OR "adaptive agent" OR "adaptive application" OR "adaptive wearable" OR "reactive environment" OR "responsive environment" OR "responsive application" OR "situated communication" OR "situated application" OR "situation aware" OR "context-sensitive" OR "environment-directed")
Dimensions of context	AND (user OR human OR device OR platform OR environment OR task)

Table 6. Search string (similar terms are included in brackets)

Note that we searched for papers published between 1st of January 1999 and 31st of December 2017. On the one hand, the selection of 1999 as our starting date was set during the establishment of the QGS (see Section 3.2.2), because we observed that the first relevant studies related to architectural approaches focused on context-aware systems dated from 1999 [1][46]. On the other hand, the selection of 2017 as end date was chosen because this systematic study started in 2016. Therefore, papers gathered from early 1999 till the end of 2017 may bring a full coverage of existing work on architecture concepts for context-aware systems.

3.2.4 Conducting the automated search

This section presents the steps followed during the search process, depicted in Fig. 2, to obtain the final set of relevant papers for this mapping study. The *automated search using all the four search engines* previously selected (see Section 3.2.1), i.e. ACM Digital Library, IEEE Xplore, Scopus and Springer Link, was firstly conducted using the final search string (see Table 6) and the limitation of the publication years from 1999 to 2017, both included. Note that the search string was differently coded depending on the search engine used, in order to fulfil the syntax requirements and capabilities of each search engine. Furthermore, the search process over each engine was conducted looking up in the *Title* and *Keywords* sections in order to keep the number of results obtained manageable. Table 7 shows the number of articles retrieved from each search engine, as well as the total of papers, 7836, after *merging all the results obtained*.

Table 7. Articles retrieved with the automated search

Search engine	Articles retrieved
ACM Digital Library	645
IEEE Xplore	139
Scopus	2747



Fig. 2. Search process

Next step was to *remove all duplicates* as some papers appeared as a result in more than one search engine. We found 468 papers duplicated, so we continued the search process with 7368 articles. Then, we checked how many articles of the QGS (see Section 3.2.2) appeared in the set of papers provided by the automated search, identifying 14 articles out of 17 papers of the QGS. 3 articles were missing in our search results with respect to QGS. Therefore, they were also included in the set of papers to be used in the following steps.

Then, we applied several *filters* to the 7368 articles selected previously taking into account the exclusion and inclusion criteria defined in Section 3.1. First, we filtered out those papers whose *venue and title* revealed clearly that they were not related at all with the topic being addressed by this study, keeping as result 7368 articles. Second, we filtered out *by abstract, introduction and conclusions*, i.e. we kept those papers that, after reading their abstract, introduction and/or conclusions, we found enough evidence that they were relevant to our study, keeping 629 references. Finally, a filter *by content* was applied so that some more articles were excluded once they were read carefully. Obviously, all papers

included in the QGS were never excluded when applying these type of filters as they must be part of the final selected papers. Note that the most common exclusion criteria at this point were: E3, as some papers were definitely not useful for this work; E4, since some papers were older publications from the same authors about the same approach; E8, because some articles were completely illegible, especially figures detailing software concepts; and E9, as some papers were published after 2017. Table 8 presents the number of papers excluded by exclusion criterion. Note that some of the papers excluded satisfied more than one exclusion criterion. At the end of this search process, we selected *463 references* for being analysed in this mapping study (see Appendix A for further details) as will be presented in Section 4.

Table 8. Number of papers excluded by exclusion criterion

Exclusion criterion	No. of papers excluded
E1	13
E2	275
E3	6140
E4	25
E5	43
E6	10
E7	192
E8	228
E9	5

3.2.5 Evaluating search performance

The completeness and performance of the automated search has been assessed by means of the aforementioned *quasi-sensitivity* concept, whose formula is presented below:

 $Quasi-sensitivity = \frac{Number of QGS studies retrieved from the automted search}{Total number of QGS studies} \cdot 100\%$

A sensitivity threshold greater than 80% can be seen as an acceptable reference for sensitivity evaluation of search performance, as recommended by Zhang et al. [56]. The quasi-sensitivity has been calculated as follows, taking into account that the total number of studies in the QGS is 17 (see Section 3.2.2 for more details) and the number of QGS studies retrieved when performing the automated search was 14:

$$Quasi-sensitivity = \frac{14}{17} \cdot 100 \cong 82,4\%$$

Therefore, with a quasi-sensitivity of 82,4%, we can state that the performance of the automated search proposed for this study is quite acceptable.

4 Results

A data extraction form was prepared that includes several fields to answer each of the RQs. We also collected other information of each paper, namely, keywords, country, venue type, research topic, application domain, research type, main contribution of the paper, and, finally, what elements are adapted during the adaptation process. All these data were collected from all the 463 selected papers (see Appendix A). In the next subsections, first, we present an overview of the selected papers regarding their publication venue (Section 4.1.1), year and country (Section 4.1.2), research topic, application domain and research type (Section 4.1.3), and what is adapted (Section 4.1.4). Afterwards, in Sections 4.2, 4.3, 4.4 and 4.5, some descriptive statistics and frequency analysis are presented to answer each research question introduced in Section 2.2.

4.1 Overview of Selected Papers

This section shows an overview of the selected papers in terms of their publication venue, year, country, research topic, application domain, research type as well as what is adapted by the system. Furthermore, at the end of this section, there is a summary about the different context dimensions covered by the SA concepts found in the selected papers, and the type of evaluation performed regarding those SA concepts.

4.1.1 By publication venue

Table 9 shows the top ten most popular publication venues for papers focused on context-aware systems regarding the number of papers published. As shown in this table, around 17,71% of the 463 selected papers are published in these publication venues: *Personal and Ubiquitous Computing*, *User Modeling and User-Adapted Interaction*, *Wireless Personal Communications*, *World Wide Web* journals, as well as the *International Conference on Embedded and Ubiquitous Computing*.

Ranking	Venue	Туре	No. of papers	% of papers
1	Personal and Ubiquitous Computing	Journal	20	4,32%
2	User Modeling and User-Adapted Interaction	Journal	12	2,59%
3	Wireless Personal Communications	Journal	10	2,16%
4	World Wide Web	Journal	7	1,51%
5	International Conference on Embedded and Ubiquitous Computing	Conference	6	1,30%
6	Procedia Computer Science	Journal	8	1,73%
7	Journal of Systems and Software	Journal	5	1,08%
8	International Conference on Universal Access in Human-Computer Interaction	Conference	5	1,08%
9	Multimedia Tools and Applications	Journal	5	1,08%
10	IEEE Pervasive Computing	Journal	4	0,86%
		Total	82	17,71%

Table 9. Number of papers published in the top ten most popular publication venues

4.1.2 By year and country

Fig. 3 illustrates the distribution of the number of papers published from 1999 to 2017. It can be noticed that there is an increasing interest of the research community in context-aware systems and their architecture artefacts especially since 2007, that it is being sustained since then.



Fig. 3. Number of selected papers by year of publication

Moreover, Fig. 4 depicts the top ten countries that have devoted more effort into research in context-aware systems. Researchers from these ten countries have published a total of 277 papers out of the 463 selected ones, i.e. the 59,83% of

all selected papers. As observed Asian countries, China and South Korea, lead this ranking significantly, followed by Germany, Spain and Italy.





4.1.3 By research topic, application domain and research type

By analysing the content of the selected papers more thoroughly, they can be classified by research topic, application domain and research type. Fig. 5 presents the *research topics* of the selected papers that have been addressed by at least 1% of the analysed papers. The research topics most frequently used are: *context inference, context management, context modelling* and *context gathering*. These terms have been described in a glossary in [42].



Fig. 5. Research topics addressed by the selected papers: number and percentage of papers

Furthermore, the most frequent application domains of context-aware systems were also identified. Fig. 6 shows these *application domains* (at least identified in 1% of the analysed papers) and groups of interest of context-aware systems, being *smart environment* and *healthcare* the most recurrent. It is worth noting the wide range of smart environments that have been considered under such category, such as smart home, smart hospitals, smart office, smart rooms, and even one paper devoted to smart kitchen. Related to healthcare several proposals have been defined for physical rehabilitation and tele-therapy.



Fig. 6. Application domains of selected papers number and percentage of papers

Finally, Fig. 7 shows how the selected papers were classified according their *research types* using Wieringa et al. [52] taxonomy (see Section 2.4 for more details). Most of the selected papers are categorized as *Solution proposals* (424 papers, i.e. 91,5%), followed by *Validation research* (118 papers, i.e. 25,48%), *Experience paper* (10 papers, i.e. 2,1%), *Evaluation research* (3 papers, i.e. 0,64%), and finally *Philosophical paper* (1 paper, i.e. 0,28%). Some papers are categorized as both Solution proposal and Validation research, e.g. P004, P009, P012 or P433, and there are also some papers, e.g. P008 or P035, which are both Solution proposals and Experience papers. Other papers, such as P065 or P411 are just Validation research and the paper P359 is the only one classified as both Validation research and Experience paper. The paper P113 is the only one categorized as both Solution proposal and Evaluation research.

/	424 papers: Solution proposal		
	P024, P025, P026, P030, P031, P032, P034, P037, P039, P040, P041, P043, P044, P045, P046, P047, P048, P050, P051, P052, P056, P058, P059, P052, P056, P059, P052, P056, P059, P051, P057, P079,	3 paper:	Evaluation research
	P080, P081, P082, P083, P083, P083, P084, P085, P086, P087, P088, P089, P090, P091, P093, P094, P095, P096, P097, P098, P099, P100, P101, P103, P106, P107, P110, P111, P114, P115, P117, P119, P120, P123, P124, P126, P128, P129,	P036	P201, P379
	P131, P132, P133, P134, P135, P136, P137, P138, P139, P140, P143, P146, P147, P148, P149, P152, P153, P155, P157, P158, P159, P163, P164, P166, P167, P170, P171, P173, P174, P175, P176, P178, P179, P180, P181, P184,		
	P185, P188, P189, P190, P191, P193, P194, P195, P196, P197, P198, P199, P200, P202, P203, P204, P205, P206, P207, P208, P209, P211, P212, P214, P215, P216, P217, P219, P221, P222, P223, P224, P225, P226, P227, P228,	1 paper:	Philosophical paper
	P230, P231, P232, P234, P235, P236, P237, P238, P239, P240, P241, P242, P243, P244, P245, P246, P247, P248, P250, P252, P254, P255, P256, P259, P261, P262, P263, P264, P265, P269, P270, P271, P273, P274, P275, P277, P274, P275, P274, P275, P274, P275, P274, P275, P274, P275, P277, P274, P275, P274,	P113	
	P278, P279, P280, P282, P283, P284, P285, P284, P285, P284, P287, P288, P289, P291, P291, P292, P293, P294, P295, P296, P299, P301, P302, P303, P306, P307, P309, P310, P311, P312, P314, P316, P317, P318, P319, P320, P321, P322, P323, P325, P325, P325, P326, P327, P333, P337, P338, P339, P342, P344, P344, P346, P347, P349, P350, P352, P353, P354, P356, P357, P358, P357, P358, P357, P358, P357, P358, P357, P358, P357,		1 paper: Vision paper
	P335, P334, P336, P337, P336, P307, P301, P302, P305, P305, P305, P307, P306, P377, P372, P377, P378, P380, P382, P383, P384, P387, P391, P392, P393, P394, P395, P397, P399, P400, P402, P404, P405, P407, P408, P409, P410, P412, P413, P414, P415, P416, P417, P418, P419, P420, P421, P422, P424, P426, P427, P400, P400, P400, P400, P400, P400, P401, P412, P413, P414, P415, P416, P417, P418, P419, P420, P421, P422, P424, P426, P427, P400,		P385
	P429, P430, P431, P432, P434, P435, P436, P437, P438, P440, P441, P442, P443, P447, P449, P451, P452, P455, P456, P457, P458, P459, P460, P461		
	P008, P035, P054, P116	5, P315	10 papers: Experience paper
	118 papers: Validation research P004, P009, P012, P015, P027, P029, P033, P038, P042, P049, P055, P057, P060, P061, P066, P071, P075, P078, P092, P102, P104, P109, P112, P118, P121, P122, P125, P127, P130, P144, P145, P151, P156, P160, P161, P169, P172, P177, P182, P183, P186, P187, P210, P213, P218, P229, P251, P257, P258, P266, P268, P272, P276, P297, P300, P304, P305, P313, P327, P331, P336, P340, P341, P348, P351, P376, P386, P388, P389, P390, P396, P398, P401, P406, P423, P425, P428, P433, P439, P444, P445, P446, P448, P450, P454, P462, P463		P053, P105, P308, P369
	P028, P065, P108, P141, P142, P150, P154, P162, P165, P168, P192, P220, P233, P249, P253, P260, P267, P281, P298, P324, P326, P334, P335, P345, P355, P366, P371, P381, P403, P411, P453		

Fig. 7. Research types of selected papers

4.1.4 By what is adapted

Most of the papers analysed in this mapping study are focused on the adaptation process and, generally, they clearly identify what is adapted by the system. We considered this information very useful and a cornerstone when designing and developing context-aware applications. Therefore, Fig. 8 depicts an overview of the most frequently adapted elements that have been used by at least 1% of the analysed papers. As shown, *services* is the element of context-aware systems most frequently adapted, being used by 61,12% of the analysed paper. UI follows, considered by 47,73% of the papers. Some aspects of the UI that have been considered by at least 1% of the papers have been depicted in a different category, such as *presentation* or *interaction mode*. Moreover, it is worth highlighting that 9,72% of the analysed papers have considered some aspect of the adaptation of the architecture.



Fig. 8. What is adapted in the context-aware systems: number and percentage of papers

4.2 RQ 1 – Software Architecture Concepts

In order to provide an overview of software architecture concepts used for building context-aware systems, after going through all the 463 selected papers, we have identified the following 16 different software architecture concepts and patterns illustrated in Fig. 9 along with the number of selected papers that apply them. Notice that the same work can use more than one architecture concept. It is interesting to highlight that the *multi-layered* approach is the most commonly used one for building context-aware systems, followed by *component-based*, *SOA* and *MAS*.



Fig. 9. Software Architecture concepts

In the following lines, we offer some examples of papers regarding these SA categories in more detail. For example, there are plenty of works that use SOA, among other approaches, such as the papers P178 which proposes the use of SOA along with a multi-layered and middleware approach. P331 also follows a SOA paradigm as it proposes the CAMPH system to provide a number of services for context data acquisition, context storage, context reasoning, service organization and discovery used to facilitate the development of pervasive homecare applications. This work also makes use of a multi-layered architecture, since the middleware infrastructure separates context data and context-aware services in different layers so that they can be used by the applications. P415 also uses SOA along with a P002P approach to provide an optimized context management.

Other works are only focused on applying a SOA approach. For instance, P065 makes use of Web services for designing context-aware applications mainly due to the loosely coupling benefit. P002 and P184 consider SOA too, as well as P108 that integrates services into widgets used for collecting data from sensors and acting like GUI interactors to make changes to the environment. For example, a Light widget can identify and change the light conditions of a specific location.

There are other proposals built using a MAS approach, such as P388 or P213. P177 proposes an agent-based framework along with a multi-layered approach for providing customized services based on context history and users' preferences. The proposed framework consists of the following layers: data gathering layer, context management layer, preference management layer and application layer. Some other papers present a component-based architecture for developing

context-aware systems, e.g. P319 and P222. The architecture presented in P347 also follows the structure of a middleware and it is distributed.

With regard to multi-layered architectures, some examples are P361 and P416 which identify four layers: the *application layer* that consists of context-aware, proactive and mobile applications as well as sensors to provide application-specific context information to such applications; the *Event-Condition-Action (ECA) engine* that is responsible for the application behaviour; the *privacy enforcement layer* that ensures that privacy-sensitive information can only be accessed according to privacy policies; finally, the *infrastructure layer* entails functionality for context management and reasoning. P362 follows a multi-layered approach having several layers, such as the *ontology editor* for modelling ontologies, the *authoring tool* for creating learning objects from existing content and contextualize them with references to the ontology, and the *delivery platform* for context-aware delivery of learning objects and managing the learning progress.

P211 proposes a client-server architecture, called ALICE (Architecture for Location-Independent Computing Environments) to support mobility by providing a range of application level client/server protocols. P061 exploits the publish/subscribe paradigm to receive and redirect messages from sensors to sensor and device models.

4.3 RQ 2 – Dimensions of Context

As aforementioned, three main dimensions of context are widely used in the research community: *User*, *Environment* and *Platform*. The purpose of this research question is to discover how each dimension of context is addressed and taken into account in the different proposals.

		406 papers: Environment P017, P022, P047, P067, P109, P136, P137, P254, P302, P335, P385, P404, P405, P410)	
5 papers: User		P057, P072, P203, P205, P223, P247, P267, P320, P321, P329		
74, P176, P227, P378, P431	208 papers: Others			
	P226, P231, P362	P001, P002, P014, P167, P183, P260, P380, P396, P461	P107, P277, P409	PO
418 papers: Platform P018, P020, P028, P097, P103, P112, P117, P134, P151, P174, P230, P235, P249, P255, P269, P279, P303, P338, P356, P358, P365, P374, P384, P390, P412, P417	P019, P038, P147, P153, P161, P339, P342, P348, P366, P398, P441	P004, P012, P024, P027, P033, P034, P036, P041, P043, P045, P048, P052, P055, P060, P061, P062, P064, P065, P070, P071, P075, P077, P078, P081, P085, P086, P087, P090, P092, P094, P098, P104, P108, P110, P111, P114, P115, P118, P125, P127, P129, P130, P131, P132, P139, P140, P144, P145, P148, P150, P154, P156, P158, P159, P160, P163, P164, P165, P169, P172, P178, P180, P182, P188, P189, P190, P194, P197, P199, P204, P207, P209, P210, P212, P213, P217, P218, P222, P238, P239, P240, P243, P246, P248, P250, P252, P258, P261, P262, P264, P265, P266, P268, P273, P275, P276, P278, P282, P283, P332, P332, P334, P337, P344, P345, P346, P347, P349, P350, P351, P352, P353, P357, P360, P361, P371, P372, P373, P375, P376, P386, P387, P331, P400, P411, P416, P418, P421, P422, P428, P432, P435, P437, P439, P440, P442, P443, P444, P448, P449, P450, P451, P452, P453, P455, P458, P460, P462, P463		
		P003, P005, P006, P008, P010, P013, P015, P016, P023, P029, P032, P035, P039, P042, P044, P049, P050, P053, P054, P056, P058, P059, P066, P068, P076, P079, P080, P083, P084, P088, P089, P095, P096, P100, P101, P102, P105, P106, P116, P119, P120, P122, P123, P124, P133, P135, P141, P142, P146, P149, P152, P155, P162, P166, P168, P170, P171, P173, P175, P177, P179, P185, P186, P187, P191, P192, P193, P195, P202, P206, P208, P211, P214, P215, P216, P219, P220, P221, P225, P229, P232, P233, P234, P237, P241, P242, P244, P245, P253, P256, P257, P259, P263, P270, P271, P274, P280, P281, P284, P285, P287, P230, P306, P307, P309, P310, P313, P316, P318, P322, P324, P327, P330, P336, P341, P343, P354, P359, P368, P369, P370, P377, P379, P381, P382, P388, P389, P392, P393, P399, P401, P402, P403, P406, P408, P414, P415, P419, P420, P423, P424, P426, P427, P429, P433, P436, P445, P447, P456, P457, P459		
P021, P031, P138, P228 P367, P425, P434	, P236, P308, P363,	P007, P009, P025, P030, P063, P069, P073, P082, P091, P113, P121, P126, P181, P196, P198, P201, P224, P251, P272, P291, P295, P312, P325, P383, P394, P395, P407, P430, P438, P446, P4	P200, 54	

Fig. 10. Dimensions of context of the selected papers

Fig. 10 presents an overview of the different dimensions of context considered by each selected paper. Most of the selected papers (418 papers, i.e. 90,2%) consider the *Platform* in some way, followed by the *Environment* (406 papers, i.e. 87,7%), the *User* (385 papers, i.e. 83,1%) and, finally, *Other* dimensions of context (210 papers, i.e. 44,9%).

There are few papers that only consider one dimension of context, such as P074 and P176 (User), P109 (Environment), or P425 (Platform). Others treat two of them, such as P057 (User and Environment), P107 (Environment and Other dimensions), P374 (User and Platform), or P454 (Platform and Environment). On the other hand, some other papers deal with all the three main dimensions of context (User, Platform and Environment), such as P141 or P388, while others also consider extra dimensions of context in addition to the three usual ones, such as P218 or P373. Next, some examples of papers related to all these context dimensions are discussed.

There are many selected papers that deal with the three main dimensions of context, i.e. User, Environment and Platform. For example, P415 takes into account the user needs and preferences, the situation or the available resources, and also the presence and GPS location information for adapting services and applications. P211 revolves around mobile devices and their resources, such as memory, processing power, battery level or network features to adapt the system. It also takes into account the environment, the application characteristics and the user preferences at runtime for dynamic adaptation of services. P177 proposes a data gathering layer in its architecture for collecting and processing the profiles of the users (gender, age, job, hobby, etc.), or for managing data, such as time, location and temperature. By doing so, this proposal is able to predict the user preferences and provide personalized services or products based on those preferences.

Some papers address other dimensions of context in addition to the main ones. For instance, some papers, such as P148 and P347, take into account the *user activity* he/she is doing for guiding the adaptation. P108 considers location of an entity within the environment, current temperature, time, ambient light or noise level, as well as physiological factors of the user, and the activity the person is involved in, such as reading or talking, to carry out the adaptation. It also considers software components attributes, e.g. load of the CPU, state of the files in the file system, or state of the application. P213 in addition to location, time, network characteristics, user preferences, user profile, or system services, also considers the ongoing activities of the users. P416 treats as context for the adaptation the user activity, geo-location, speed and direction, battery level, time, available networks, favourite places, and means of transport. P331 considers many context information, e.g. location of a device, occurrence of an event, room temperature, number of people in a room, medical profile, blood glucose, blood pressure, heart rate or body temperature of a user. It also manages both daily activities of the user being monitored, e.g. eating, and abnormal activities, e.g. falling down, that should be detected in a reasonable time. P065 proposes the following context dimensions: user, device, network, activity, service, location, and resource. P060 presents a framework for improving recommender systems based on the user activity, his/her preferences and profile, information regarding the different devices used, and the location of the user, light, noise, temperature of the environment or time of the day, among others.

Other approaches pay attention to the *historical data* as another dimension of context, that is, past information related to the context of use, e.g. the interaction history of the user. For example, P362 exploits the learning history of the user for adapting his learning activities. P107 employs words used recently, user's previous conversations, current time, date, and user's location to select conversation topics, among others. P075 presents a system that takes into account location, temperature, light, noise or humidity level of the environment, behaviour and preferences of the user, battery level of the devices, among others, as well as the user activity and his historical information, represented as a data set generated using a variety of sensors. By analysing all this information, the system is able to learn user behaviours in an office environment in order to use the inferred rules to populate a user model and support appropriate proactive behaviour, e.g. turning on the fan under appropriate conditions.

Some papers exploit the user's *interaction task* for the adaptation process. For example, P333 presents a software that supports different sensor types, including location and user interaction sensors, to monitor the active application the user is involved in, among others. P107, in addition to historical data, uses the user' interaction task to support a word predictor to foresee which word a user is typing using the letters that have been typed so far. P169 proposes a context-aware system for museums that has the ability of monitoring the continuous changes of the user's interests during his interaction with the system. Finally, P183 presents an Affect and Belief Adaptive Interface System (ABAIS) that exploits a variety of user assessment methods (diagnostic tasks, physiological sensing, etc.) to determine which adaptation approach is the most appropriate.

4.3.1 RQ 2.1 – User dimension of context

We discovered a great variety of terms used for characterizing the *user* of a context-aware system. The terms that are used by at least 1% of the analysed papers are represented in Fig. 11, being among the top most frequently used terms: *profile* of the user including abilities, age, identity, etc.; *preferences*, such as interests; *physiological data*, e.g. heart rate, blood pressure, temperature; interaction considering aspects as user's behaviour or goals while interacting with the system; *health state* refers to any information related to the health conditions or medical data of the user such as his diseases; *emotional state* of the user as, for instance, bored. Note that we have grouped terms according to one of them, whenever they had less than 1%. For instance, in Fig. 11, physiological data is used to group several terms, such as breath rate and oxygen level, that have less than 1% of references. As can be observed, the terms most frequently used are *profile*, with a 41,03% of use and *preferences*, which appears in 31,74% of the selected papers.



Fig. 11. Terms for characterize thing User dimension of context

4.3.2 RQ 2.2 – Platform dimension of context

While characterizing the platform of context-aware systems, we have also detected a great heterogeneity among the terms used in the selected papers. Fig. 12 the top terms that have been used by at least 1% of the analysed papers as well as how many times they have been used. However, there is some information related to the platform more widely used, as Fig. 12 shows: *devices* is the most frequently used; *sensors* considering either software and hardware elements in charge of collecting data from the context is the second most frequently used; *network* and its relevant information, e.g. speed, bandwidth or connectivity; *services* offered by the system, e.g. weather services or traffic information services; *content* offered to the user; *hardware* considered as other elements of the devices, being battery the most frequently used for considering aspects regarding the energy consumption; *storage* to be effectively managed by the system; *screen* and its related features resolution, size and color; *software* considering different elements such as applications developed, resources the system uses, etc.; *UI* offered to the user and different features regarding its interaction modality and format; *actuators*, both software and hardware, that apply changes to the context-aware system; and finally *CPU* being and *cost*.



Fig. 12. Terms for characterizing the Platform dimension of context

4.3.3 RQ 2.3 – Environment dimension of context

Fig. 13 the most frequently used terms for characterizing the environment that have been used by at least by 1% of the analysed papers. Similar to the previous dimensions, we have identified many terms related to the environment. Nevertheless, there are also some of them that are very recurring. First, as identified by the feature *user in the environment*, there are many references that exploits user's information regarding the environment such as users' location regarding the environment, his movement, his orientation, etc. Next, we observed, it is very frequently used information about the *environmental conditions*, such as temperature, light, noise, etc. where the interaction is taking. *Events* happening in the environment is followed by all the terms used to describe the *physical environment* such nearby people, or nearby devices. As can be observed, the term most frequently used is by far the *location*, being used by 82,51% of the selected papers, followed by the *temperature* 19,22%, and *light* 17,71%.



Fig. 13. Terms for characterizing the Environment dimension of context

4.3.4 Other dimensions of context

As aforementioned, we also wanted to determine if some proposals make use of other dimensions of context, in addition to User, Platform or Environment. In this sense, we have identified the following three dimensions of context: Historical data, User activity or Interaction task (see Section 2.4 for more details). Table 10 presents the number of papers that use these additional dimensions. Namely, the *interaction task* the user is doing with the system is used in 22,25% of the selected papers, the *user activity* the user is performing (e.g. walking, sleeping, shopping, cooking, etc.) in 16,41%, and the *historical data* in 11,45%.

Dimension	No. of papers	% of papers
Interaction task	103	22,25%
User activity	76	16,41%
Historical data	53	11,45%
Facts	2	0,43%

Table 10. Other dimensions of context

4.4 RQ 3 – Evaluations of Context-Aware Adaptations

Results presented in this section focus on proposals that apply context-aware adaptations and that provide some mechanism to evaluate these context-aware adaptations, that is, to determine their quality. Therefore, we want to find out to what extent the evaluation of context-aware adaptations in context-aware systems is covered in the literature to estimate its maturity. Fig. 14 shows the different types of evaluation identified to assess the adaptations in context-aware systems: *Automated evaluation, Design space, Satisfaction evaluation, Survey, Ubiquitous Computing Acceptance Model, Usefulness evaluation, User feedback, User-centred Evaluation (UCE), Visual and formal validation with a GUI.* Note that there are quite a few papers, i.e. only the 7,99% of the selected papers that deal with some kind of adaptation evaluation out of the selected papers. This lack will be further discussed in Section 5.



Fig. 14. Adaptation evaluations found in selected papers

P281 presents a framework that consists of, among others, a *design space* for evaluating the adaptation coverage in a reliable way. This design space is used to assess and compare the adaptation levels of different applications based on common criteria. However, other papers focus on evaluating the adaptation process by means of its *usefulness*, such as P139. This paper suggests a user-centred evaluation to verify the usefulness of the adaptations applied.

We have found other types of evaluation that assess the adaptation by means of a Graphical User Interface (GUI) or wizard that shows the user how the adaptation will change the system, so he/she can accept or reject it in a visual manner. We have called this type of evaluation *visual validation*. For example, P186 presents a tool for the generation of context-aware adaptive software system implementations that supports a visual validation of the context-aware adaptive behaviour of the system through a GUI. The user can press the button *Adapt to the Context Information Changes* to see how the context changes impact the system. This evaluation feature enables the detection of missing and incorrect adaptation behaviours in a visual manner. P186 proposes a visual validation of the adaptation too. Its proposal is able to recognize the actual policy rule that caused an incorrect behaviour of the system in order to assist the user when he/she corrects the

adaptation behaviour by means of a wizard. Finally, P388 proposes a widget to support the user while expressing his/her opinion about the current presentation topic. This is carried out by means of two buttons: the *Wow* button that reflects the user feelings whenever he/she is positively impressed by a specific presentation and the *Basta!* button, which is used whenever the user is not interested in the current topic, thus stopping the current presentation.

We have also detected that other articles evaluate the adaptation by using the user *satisfaction*, usually through some type of survey or questionnaire. For instance, P012 evaluates the user's satisfaction regarding the offered context-based recommendations. P181 measures the user's satisfaction exploiting the perceived ease of use or perceived control, among others. P390 estimates the satisfaction degree between the adapted media selected by its proposed Content Adaptation Process and the media requested by the learner. Finally, P398 proposes the evaluation of the user's satisfaction regarding functionality and adaptability by means of a survey.

Lastly, automated adaptation is a possible solution to address evaluation of adaptation in many situations, when the target users are not available, time constraints make impossible making a regular evaluation or there are so many possible adaptations that regular evaluations are not a plausible option. P332 apply as input to the system under evaluation a programmed user model to simulate the behaviour of the user in the long-term more easily and thus draw useful conclusions about the quality of the adaptations relying on some underlying models such as a task model.

4.5 RQ 4 – Evaluations of Software Architecture Concepts

Finally, we wanted to find out what evaluations have been carried out up to now to validate the software architecture concepts proposed for context-aware systems. By doing so, we will be able to discover the degree of *maturity* of the different software architecture concepts. Therefore, we have categorized the different software architecture concepts found in the selected papers according to the type of evaluation presented by using the taxonomy of Wohlin et al. [54] for classifying *Empirical strategies*. We have also added the category *Example* for papers that do not present a formal empirical evaluation (see Section 2.4 for more details about this categorization).

Fig. 15 shows the results from the point of view of the evaluation type presented in each selected paper. The majority of the papers only include an *Example* (135 papers, i.e. 29,15%), followed by *Case Studies* (104 papers, i.e. 22,46%), *Experiments* (50 papers, i.e. 10,80%) and finally *Surveys* (20 papers, i.e. 4,3%). Thus, there are 161 papers among the selected ones (34,7%) that do not have any type of evaluation, validation or example, neither formal nor informal one, for assessing the SA concept proposed. Note that we have found neither formal nor informal evaluation for assessing *aspect-oriented architectures*.

161 papers: All selected papers			135 papers: Example
P002, P006, P013, P014, P016, P017, P020, P023, P068, P069, P072, P080, P081, P087, P088, P089, P132, P133, P137, P138, P139, P143, P146, P148, P178, P179, P181, P184, P185, P191, P194, P195, P214, P215, P219, P223, P224, P225, P228, P230, P252, P254, P261, P262, P265, P269, P270, P273, P306, P307, P309, P311, P315, P316, P318, P320, P356, P357, P362, P363, P364, P367, P369, P370, P408, P409, P414, P415, P418, P419, P421, P424, P	P030, P031 P091, P094 P149, P153 P196, P197 P231, P232 P274, P277 P322, P332 P375, P377 430, P431,	L, P034, P035, P039, P047, P053, P056, P058, P059, H, P096, P099, P100, P106, P110, P113, P116, P124, B, P163, P165, P166, P170, P173, P174, P175, P176, r, P199, P200, P203, P205, P206, P207, P209, P211, L, P235, P236, P240, P241, P242, P243, P248, P250, 7, P280, P283, P285, P286, P288, P292, P296, P302, L, P333, P338, P339, P342, P343, P344, P346, P349, r, P378, P379, P383, P384, P385, P397, P399, P402, P435, P437, P441, P442, P456, P459, P461	P001, P005, P007, P010, P019, P032, P037, P040, P041, P043, P044, P045, P046, P048, P052, P054, P063, P070, P073, P074, P076, P077, P079, P083, P085, P090, P093, P097, P098, P105, P107, P111, P114, P115, P119, P120, P123, P128, P129, P131, P134, P135, P136, P140, P147, P152, P155, P157, P158, P150,
	50 pape	ers: Experiment	P152, P155, P157, P158, P159, P164, P167, P171, P188, P189,
104 papers: Case study			P190, P193, P198, P202, P204,
104 papers. case study	P021,	P003, P012, P018, P022, P024, P028, P033,	P226, P227, P234, P237, P238,
P004, P009, P011, P015, P025, P026, P027,	P420	P050, P051, P084, P086, P101, P112, P117,	P239, P244, P246, P247, P256,
P038, P042, P049, P057, P061, P062, P064,		P145, P154, P168, P192, P201, P216, P218,	P263, P264, P275, P278, P282,
P065, P066, P067, P071, P078, P082, P092,		P220, P245, P253, P260, P267, P279, P287,	P284, P289, P291, P293, P294,
P095, P102, P103, P104, P108, P109, P118,		P290, P297, P303, P335, P340, P345, P348,	P295, P301, P308, P310, P312,
P121, P122, P125, P126, P127, P130, P141,		P358, P366, P368, P376, P396, P398, P412,	P314, P317, P319, P323, P325,
P142, P150, P151, P161, P172, P177, P180,		P417, P436	P328, P329, P330, P337, P347,
PIOZ, PIOS, PIOS, PIOZ, PIZZ, PZZZ,			P350, P352, P353, P360, P361, D265 D272 D273 D274 D290
P273 P276 P281 P298 P299 P300 P304	20 paj	pers: Survey	
P305, P313, P321, P324, P326, P327, P331,		P060, P162, P334, P371	P395 P400 P407 P410 P413
P336, P341, P351, P354, P355, P359, P381,		, . , .	P416. P422. P426. P427. P429.
P386, P388, P389, P390, P394, P404, P405,	D2E4	P008 P029 P036 P055 P075	P432, P438, P440, P443, P447,
P411, P423, P425, P428, P433, P434, P439,	P251	P144 P156 P160 P169 P266	P452, P455, P457, P458, P460
P444, P445, P446, P448, P449, P450, P451,		P268, P401, P403, P406, P463	
P453, P454, P462			

Fig. 15. Type of evaluation presented in each selected paper

With regard to the *Example* category, most of the software architecture concepts use examples to illustrate and validate the proposals in an informal way regarding the following SA concepts: *client-server architecture* (P070 or P128),

component-based architecture (P019, P054 or P373), middleware architecture (P382 or P391), distributed architecture (P347 and P413), model-driven architecture (P090 or P120), multi-layered architecture (P295 or P393), event-triggered architecture (P440), micro-architecture (P129), service-oriented architecture (P007 or P171), multi-agent system (P111 or P263), publish/subscribe architecture (P115), and peer-to-peer architecture (P005 or P284).

We have identified some *Surveys* for assessing the following software architecture concepts: *client-server architecture* (P266), *component-based architecture* (P156 or P160), *distributed architecture* (P251), *model-driven architecture* (P162), *multi-layered architecture* (P401 or P463), *event-triggered architecture* (P036), *service-oriented architecture* (P055 or P144), and *multi-agent system* (P029 and P268).

Some *Case studies* have been also carried out for evaluating the following software architecture concepts: *client-server* architecture (P150 or P454), *blackboard architecture* (P210), *component-based architecture* (P009, P161 or P172), *middleware architecture* (P042 or P102), *distributed architecture* (P251), *model-driven architecture* (P064 or P066), *multi-layered architecture* (P078, P249 or P446), *event-triggered architecture* (P272), *service-oriented architecture* (P065, P151 or P331), *multi-agent system* (P062, P082 or P299), *publish/subscribe architecture* (P061 or P122), *peer-to-peer architecture* (P331), *pipe-and-filter architecture* (P351), and *plug-and-play architecture* (P127).

Finally, we have found a few *Experiments* for the validation of the following software architecture concepts: *component-based architecture* (P112 and P334), *model-driven architecture* (P162), *multi-layered architecture* (P012, P033, P218, P253, P297, P340, P376), *service-oriented architecture* (P371), and *multi-agent system* (P033, P145, P348, P398).

Next, some of the most referenced selected papers are highlighted to provide some insights about the evaluation strategy applied to validate their software architecture proposals.

P416 uses just an *example* to demonstrate and validate its *multi-layered* architecture. It presents a typical use of the infrastructure in a mobile healthcare application scenario. On the other hand, P036 presents a *survey* for assessing the maturity of the *event-triggered* and *multi-layered* architecture proposed. It is the only paper that has applied and evaluated the proposal in a real industrial scenario. Therefore, it has been categorized as *Evaluation research*. First, the system was deployed in a hospital and it went into pilot use. Afterwards, a survey focusing on users' experience while using the system was handed out to the doctors, and over 40 responses were collected.

Other papers have validated their proposals by means of *case studies*. P065 proposes a *service-oriented architecture* which has been evaluated by a case study applied in a medical system scenario and using concrete adaptation rules. P108 also provides a *service-oriented architecture* assessed by a case study. Authors show how the toolkit proposed can be used to design and build context-aware applications and how it supports reusability of components, evolution of applications, and acquisition of complex context. P177 suggests a framework based on the *multi-agent* and *multi-layered* approaches. This framework has been validated by using a case study to show its feasibility. P213 presents a *multi-agent system*. Authors designed and implemented a prototype as a case study to test the applicability, scalability and context effectiveness in everyday activities. Finally, P331 makes use of several paradigms, such as *peer-to-peer*, *multi-layered* and *service-oriented architecture*. To validate the proposal, authors used a case study by deploying a prototype in their research laboratory and using different real scenarios, such as a smart home, an office or a shop.

To conclude, P340 proposes a context-aware system supported by a *multi-layered* infrastructure. Its effectiveness has been proved by conducting *experiments* with a complete implementation of the system with some available sensors and an Android-based handheld device that acted as the host for the main activity recognition module.

5 Discussion

In this section, an analysis of the results described in the previous section regarding the Research Questions is presented, in order to foresee future directions in this research field (Section 5.1). Afterwards, some implications for researchers and practitioners are discussed (Section 5.2).

5.1 Analysis and Synthesis of Results

This section presents an analysis and synthesis of the results described in the previous section.

5.1.1 Publication venues, years and countries

As shown in Table 9, only around 17,71% of the selected papers were published in the top ten publication venues. The rest of the papers were published in a great variety of venues. One of the reasons for this issue is the heterogeneity and number of terms we have considered while performing the automated search. This great amount and variety of words implies a wide number of results, from venues focused on ubiquitous and pervasive computing, to others related to mobile

and wireless communications. We detected a high number of venues related to ubiquitous and pervasive computing; context modelling, especially user modelling; smart environments and ambient intelligence; adaptive systems; and HCI in general. These topics are directly related to context-aware so it is obvious that many papers were published in this type of venues. There were also other venues that revolved around other topics not related to context-aware, but to software architectures. Evidently, the results depended on basically the words used for defining the search string and, as aforementioned, this search string included many terms associated to software architectures, context-aware and context dimensions (see Section 3.2.3).

As Fig. 3 illustrates, a clear evidence has been found regarding the growing interest of the research community in context-aware systems and their supporting architecture. It can be seen that, although this topic emerged many years ago (seminal papers were published in 1999), it has been since 2007, approximately, when the interest has increased significantly until nowadays. One of the main reasons for that interest on context-aware systems is the enormous benefit they provide and the wide availability of sensors at a reduced cost. Context-aware systems are usually implemented in smart environments where users interact through an interface transparent for them. This ubiquitous and transparent way of interaction is becoming a very attractive paradigm, especially when dealing with older or impaired people. Furthermore, there is an increasing attention on healthcare since the worldwide population is aging progressively, among other issues. In this way, context-aware systems have emerged as an appropriate solution for assisting, especially, older people in their daily life. This fact is confirmed as one of the main application domains for context-aware systems is healthcare (see Section 5.1.2).

It can be observed in Fig. 4 that Asian countries China and South Korea lead the ranking of countries more interested in context-aware systems, followed by Germany, Spain and Italy. However, as aforementioned, the advantages of contextaware systems are recognized worldwide so papers about this topic are published all over the world. In conclusion, observing the tendency, context-aware systems will remain one of the main research fields in future years, involving more and more countries worldwide and possibly more and more publication venues.

5.1.2 Research topics, applications domains and research types

The research topics most frequently used, as Fig. 5 depicts, are: *context inference, context management, context modelling* and *context gathering*. These results highlight that most of the selected papers focus on handling the context of use in some way, as well as on developing context-aware applications. This has positive implications. For example, most architectural proposals take care of context in some way, one of the main aims of this study.

Furthermore, we also aimed at identifying the most frequent application domains of context-aware systems (see Fig. 6) that were: *smart environment* and *healthcare*. The vast majority of these systems focus on building smart spaces to assist people in their daily life. In this line, we detected that many papers whose proposals were developed to assist people while they are at home, in a hospital or in a city, even one paper was found regarding smart kitchens. Next, healthcare is the second more important application domain for context-aware systems.

Fig. 16 depicts how many papers are dealing with the five most frequent topics (x axis) and the five most frequent application domains (y axis). When considering both aspects, research topics and application domains, we can notice that all the frequent topic related to smart environment are the most frequent one, followed by that related to healthcare. Context management is the main interest when dealing with applications for smart environments where there are usually a wide variety of sources of context information so that there is a clear need for solving conflicts among such sources. On the other hand, one of the main focus of the research has been related to context inference in applications for smart environment, healthcare and older people. This result is consistent as most of these applications exploit some kind of reasoning mechanisms to infer which kind of activity is carrying out the user using context information in order to provide him with the best assistance. Finally, it can be noticed that less attention has been paid to the definition of proposals for context-aware applications development, that is, on the description of whole frameworks for developing applications of these domains as most of the analysed papers emphasis on an specific aspect, such as context inference or context gathering.



Fig. 16. Relationship between research topics and applications domains

Regarding the research types of the selected papers, they were classified according to Wieringa et al. [52] taxonomy (see Fig. 7). Most of the selected papers were categorized as *Solution proposals* (91,58%), followed by *Validation research* (25,48%), *Experience paper* (2,16%), *Evaluation research* (0,64%), *Philosophical paper* (0,21%) and *opinion paper* (0,21%). These results are quite disturbing as many selected papers were only solution proposals, without any type of validation or evaluation, neither formal nor informal. Only 25,48% of the selected papers offered some kind of laboratory validation which means that the context-aware systems were developed, but not deployed nor evaluated in a more real and industrial setting. P036 was the only one classified as Solution proposal and Evaluation research, since its proposal was developed and evaluated in a real and industrial scenario. First, the system was deployed in a hospital and it went into pilot use. Afterwards, a survey focusing on users' experience while using the system was handed out to the doctors, and over 40 responses were collected.

Thus, we have found that there are many proposals attending to develop context-aware systems for solving a concrete problem. However, we have detected a lack of maturity as many of these works are not validated at all. In this sense, future directions in context-aware computing should go on performing more evaluations on such systems in order to implement them in a real environment and provide additional feedback about the effectiveness, usefulness, etc. these systems can offer.

5.1.3 Context dimensions covered by software architecture concepts

As Fig. 17 illustrates, the three main dimensions of context (User, Platform and Environment) are covered by different software architecture concepts. With respect to other dimensions of context, such as historical data, user activity or interaction task, all the analysed architectural approaches cover some of them, except for the aspect-oriented architecture.



Fig. 17. Relationship between context dimensions and SA concepts

Therefore, it can be claimed that architectural concepts used for developing context-aware systems take into account the context of use in some way. Hence, context is not addressed as an isolated issue, but it is included as a key part of the software architecture. It has been observed that the main purpose was to reflect that some actions would be executed in the system, depending on the current context. Furthermore, it was also detected that most of the systems analysed react to context changes using rules defined following an Event-Condition-Action (ECA) format. Hence, systems perceive a context change by means of an event, then systems check whether the condition related to such event was satisfied in order to react using the corresponding action. For example, in a smart home scenario, an ECA rule could be defined so that, if the system detects that a person is falling down (event), and such subject is an old person (condition), an alarm

call is made to the emergency centre, the caregivers and the family of the person (action). Usually, the necessary infrastructure to exploit the ECA approach, and the process to obtain valuable and high-level context information of data collected from different sensors (e.g. that the user is laying on the floor because of a falling) are usually considered during the design of the software architecture of the context-aware system. This is done by adding one or more architectural components that are specifically in charge of this treatment of the surrounding context.

Regarding the software architecture concepts used to develop context-aware systems, most of the proposals employ, as Fig. 17 depicts, *multi-layered* architectures, *component-based* architectures, *SOAs* and *MASs* since they appeared as the most suitable ones for context-aware systems. The main reason for the popularity of these software architecture concepts is their simplicity, ease of use, and that they provide useful and effective mechanisms to facilitate the adaptation of the system to context conditions. For instance, as seen in Section 4.1.4, many context-aware systems are focused on adapting the services offered to the user so that most of them employ a SOA approach since its main foundation is the support of services. Moreover, the less popular software architecture concepts are aspect-oriented, blackboard, distributed, micro-architecture, pipe-and-filter, and plug-and-play.

It is interesting to highlight that, generally, all these architectural approaches deal with all the dimensions of context, but with some particularities. In general, the platform dimension is the most widely dealt with for adapting the system, followed by the environment, the user and other dimensions. Even though the environment dimension was identified as the second one, there is no significant difference regarding the number of papers between this and the user dimension. Therefore, despite these little differences, we confirm that all context dimensions, including other ones, are more or less considered in a similar way by every software architecture concept.

In conclusion, most of the selected papers, independently of the architecture schema adopted, when they say they take into consideration the current context for adapting the system, such context is referred most of the time to the user, the platform and the environment, and least often to other dimensions of context.

5.1.4 What is adapted?

After analysing all the selected papers, we noticed that they were focused on adapting specific elements of the system so we looked for those elements that are adapted since we considered this information very useful and a cornerstone when designing and developing context-aware applications. As seen in Fig. 8, *services* are overwhelmingly the component of context-aware systems most frequently adapted, followed by the user interface, as well as the *content* of these UIs.

The main reason for this is that many of the context-aware systems analysed were designed following a service-oriented approach. It is interesting to highlight that they provide services as assorted as calendar, weather, location, time, news or traffic information ones. Moreover, since many selected works were focused on the healthcare domain, *alarm calls* appeared here too as a relevant element to be adapted in context-aware systems. An alarm call referred to any type of message delivered to caregivers, doctors or patient's family when an emergency situation was detected. Patient was usually an old person and/or an impaired one who still wants to live alone, but needs to be monitored to prevent them from getting severe injuries.

It is not surprising that UI presentation and navigation are also commonly adapted, since content, presentation and navigation have been traditionally the three aspects considered for adaptive hypermedia, where adaptation has been a key topic for several decades. Some aspects adapted are related to the increasing ubiquity and intelligence of context-aware systems. For instance, recommendations (4,1%) are used in different domain, but it's probably in e-commerce where they probe more useful nowadays. Because of ubiquity, adapting the way people makes an activity (4,97%) has gained also importance in context-aware systems so that these activities actually fit the current context of use.

From these results, we can claim that not many context-aware systems were interested in adapting or reconfiguring the software architecture (1,3%), its components (3,24%) or its configuration (5,18%), compared to other aspects adapted previously discussed. As mentioned before, there is no doubt that context-aware systems should pay attention to its architecture and the way it is designed. In this sense, they could take advantage of the support provided for architectural changes to offer adaptation to the context of use. Therefore, this significant gap in context-aware computing should be covered in future research, requiring more effort in order to align the adaptation at *architectural level* (changes in the configuration of architectural components) and at *HCI level* (changes in the interaction modality or the user interface).

5.1.5 RQ 1 – Software architecture concepts

RQ1 focuses on identifying which architectural concepts were used in existing context-aware systems in order to provide an overview of primary studies aiming at defining and/or exploiting those software architecture concepts. Fig. 9 summarizes the results that answer this research question. One of the problems faced to answer this RQ was that not all the selected papers describe clearly the architecture patterns applied. Sometimes, such software architecture concepts had to be identified from the context of the paper, while sometimes, there was no way to detect a well-defined architecture

approach. In this last case, 55 papers, which represents 11,87% of the total of selected ones, did not describe a software architecture concept. Most of these papers were published in HCI and ubiquitous computing venues, and others were related to mobile networks and wireless communications, but not in venues related to Software Engineering or Software Architecture. These findings demonstrate our first intuition about the existing gap between HCI and SA. Most of the papers focused on context-aware adaptations, but do not consider the software architecture at all since they neglect partially or completely any possible change in the system architecture during the adaptation process.

As illustrated in Fig. 9, *multi-layered* architectures, *component-based* architectures, *SOAs* and *MASs* were the most widely used for context-aware systems. On the contrary, the software architecture concepts less popular were *plug-and-play*, *micro-architecture*, *blackboard*, *pipe-and-filter* and *aspect-oriented* architectures. It is worth noting that many works use at the same time several software architecture concepts for structuring their architectural elements. We found many different combinations, but there were three quite repeated (see Table 11), which involved the most popular ones: multi-layered architecture and MAS; multi-layered architecture and SOA; and, finally, multi-layered and component-based architecture.

Combination of software architecture concepts	No. of selected papers
Multi-layered architecture + MAS	12
Multi-layered architecture + SOA	10
Multi-layered + component-based architecture	10

Table 11. Top three combinations of software architecture concepts

It is worth noting that the application of multi-layered architectures was considerably extended in context-aware systems. The main benefit of using a multi-layered architecture here is the separation of concerns since it enables the creation of different layers, organized hierarchically, where each one is in charge of providing a different functionality of the context-aware system. In this sense, the most popular elements for representing this functionality in the different layers were agents (MAS), services (SOA) and components (component-based architecture), as observed in Table 11.



Fig. 18. Context treatment in context-aware systems

After analysing the selected papers, we found that some of these elements in the architecture of context-aware systems were commonly used to provide specific context-aware functionality. For example, we found elements for adapting the UI, namely for adapting the content information of the UI, and others for handling services associated to the context of use or any other relevant aspect of/for the system.

We found a common schema, illustrated in Fig. 18, for dealing with context in context-aware systems. Some of the steps that make up this process are often turned into components, services or agents in the software architecture responsible for such context handling. Context modelling is the first step and consists in stablishing what the representation of context information will be, e.g. by means of an ontology or a modelling language such as OWL or RDF. Second stage corresponds to *context gathering*, i.e. obtaining context data when required or subscribing to event notifications from multiple sources, usually demanded by services and applications (adapted from [38]). All these context data should be modelled as previously established at the first stage (adapted from [45]). Third, some context information cannot be obtained directly, but by processing low-level context information. This task is performed by the context inference part of the system, usually by applying some reasoning mechanism, as mentioned before. For example, to know whether a person is travelling and which means of transport he is using, a reasoning must be carried out using other context information, such as Wi-Fi access points or calendar information (adapted from [47]). Here, it is interesting to highlight that some context-aware systems present a single component/service/agent in charge of both context gathering and inference, usually called *context discovery*. Once the context inference is executed, and the system has all the necessary context information, then *context management* is performed for solving conflicts among contradictory sources of context information, processing the context information previously gathered (adapted from [8]). Finally, the last step is context delivery, i.e. when an event occurs and matches any rule previously specified, then some context information

is provided to the application or service that requires it, or simply when such application or service requires it (adapted from [14]). It is worth noting that some of this context information can be stored in, e.g. a database, for future context requests. This information stored refers to the historical context of the system.

However, not all the selected papers supported all these stages of the context treatment process, so its software architecture normally reflected only those steps they were focused on. More details about the software architecture concepts and terms related to context treatment are provided in [42], respectively.

5.1.6 RQ 2 – Dimensions of context

This research question revolves around discovering how software architecture concepts proposed for context-aware systems considered the three main dimensions of context (User, Platform and Environment) to support adaptation. We have also included in our study the context dimension *other* since we noticed many authors exploit other information, e.g. historical context, user activity or interaction task. Fig. 10 presents an overview of the different dimensions of context considered in the selected papers. Most of them (90,2%) considered the Platform in some way, followed by the Environment (87,7%), the User (83,1%), and, finally, Other dimensions of context (44,9%). This confirms that the three main dimensions of context are more or less equally dealt with by context-aware systems. This is highly important, since the more dimensions and terms related to context are considered by the system, the more appropriate and correct the adaptations provided will be, because, although context-aware systems will become more complex, they will be able to react to much more events and changes in the surrounding context, avoiding that users refuse to use the system.

The majority of the selected papers lack a description of terms and dimensions of context, and neither presented nor described context concepts with enough clarity those context concepts by using, for instance, a table, a figure or just an enumeration. On the contrary, we notice that these context terms were hidden through the text and distributed all over the paper among the different sections. Another important finding is the heterogeneity of the different context terms. Every author uses his/her own words to refer to the different context terms and dimensions, thus hindering the classification of context dimensions. This led us to define a glossary with different context terms and synonyms that is presented in [42].

We only found some works that concisely presented and modelled such context dimensions. For example, P088 proposed a *Context Dimension Model* for defining at design time the possible contexts of use of a given domain of interest by using a hierarchical structure. This structure is made up of *context variables*, e.g. time, location, current company; and the set of *values* for such context variables, e.g. evening, Spain, with friends. Context variables values perceived at runtime by means of sensors trigger the activation of a given context. P288 also proposed a context model for categorizing context entities using agents, devices, environment, location and time. Location and time were kept separated from other concepts to emphasize the spatial and temporal aspects of the ubiquitous environment. Finally, P127 distinguished the variety of context elements along with how such context information was exploited to adapt the application in the form of a table. For instance, it specified that "when the *battery level* is extremely low, the application terminates in a graceful way to avoid data loss. Also, the interaction with the user is adapted to his/her *preferences*, e.g. vibration on/off".

As seen in Section 4.3.1, we discovered a great variety of terms used for characterizing the *user* of a context-aware system. Fig. 11 presented those results. The *profile* of the user represented the most frequent element that context-aware systems used for the adaptation, along with the *preferences* and information related to the *health state* or *physiological data*, especially the heart rate, which determined the current state of the user. We believe that user preferences have become a key element for adapting the system, since they explicitly exhibit the likings of the user regarding many aspects, e.g. the language, colour, type of content, interaction modality, etc. Therefore, if the system is able to adapt its behaviour depending on such preferences, the success of the adaptation process is almost ensured, i.e. there is a strong likeliness that the user will not reject the adaptation proposed, improving, thus, the quality of adaptation desired for every context-aware system. It is also evident that handling at runtime any information about users' health conditions is quite valuable since healthcare appeared as one of the most important issues worldwide, as previously mentioned.

On the other hand, as seen in Section 4.3.2, we also detected a high heterogeneity of the terms used in the selected papers for characterizing the *platform* of a context-aware system. Fig. 12 showed those results. Nevertheless, *devices* and *sensors* were in the first positions of the ranking of most used concepts. In context-ware systems, devices in general, and sensors in particular, are very important elements. Sensors are essential as they are in charge of monitoring at runtime many context information, from location (hardware sensors) to system logs (software sensors). Furthermore, almost all the selected papers consider them in the design of their architecture, which demonstrates their importance in context-aware systems. But also, other types of devices, such as smartphones, computers, monitors, etc., were considered for adaptation purposes. Many context-aware systems adapt their behaviour attending to network bandwidth or connectivity, screen size, available memory, battery level, or interaction modalities supported by the devices.

Section 4.3.3 presented many features for determining the *environment* of a context-aware system. Fig. 13 depicted those results. Some of the top terms were *location*, *time*, *temperature*, *light* or *noise*. But especially, the location of the user/entity in the interaction place with the 83,38% of appearance rate was the most used one. Many papers include in

their architectural designs components exclusively used to locate elements in the environment, such as GPS or GSM. Furthermore, the majority of those works do not talk about context-aware systems, but only about purely location-aware systems.

To conclude, as mentioned in Section 4.3.4, we also wanted to determine if some proposals use other dimensions of context, different from User, Platform or Environment. We distinguished other three dimensions of context widely used by the context-aware research community: *Historical data, User activity* and *Interaction task*. Table 10 presented those results. Namely, the *interaction task* the user was doing with the system was used in 22,25% of the selected papers, the *user activity* the user was performing (e.g. walking, sleeping, shopping, cooking, etc.) in 16,41%, and *historical data* in 11,45%. Therefore, it is interesting to highlight that more than 40% of all selected papers also paid attention to other dimensions of context for adapting the system, making them more rigorous and providing a higher quality of adaptation than others which do not considered these other dimensions at all. Moreover, in addition to detect a broad consensus regarding the main context dimensions (User, Platform and Environment), we also detected such consensus when considering other dimensions of context. In this way, we were able to classify context data related to Historical data, User activity and Interaction task, but defining them previously in a more formal way (see Section 2.4 for more details).

5.1.7 RQ 3 – Evaluations of context-aware adaptations

This research question focuses on discovering to what extent the evaluation of context-aware adaptations in contextaware systems was covered in the literature. The results of this research question are presented in Fig. 14Fig. 15. Surprisingly, there were only 37 papers with some kind of proposal for assessing the adaptation, i.e. only the 7,99% of the selected papers. After a deep analysis, we classified them into the following categories for assessing the adaptation:

- Design space evaluation. To assess and compare adaptation levels of different applications based on unified criteria.
- Usefulness evaluation. A user-centred evaluation to verify the usefulness of the adaptations applied.
- *Visual validation*. To assess the adaptation by showing how it will change the system, usually by means of a Graphical User Interface (GUI) or wizard.
- *Satisfaction evaluation*. Evaluation of the adaptation by taking into account the satisfaction of the user, usually through a survey or questionnaire.
- User feedback. The user provides some simple feedback, for instance a rating, for the adaptation.
- *User-centred evaluation*. The adaptation was assessed throughout the development of the context-aware system starting from prototypes to the final application, as usually done in user-centred methodologies.
- *Survey*. A survey not intended for user satisfaction assessment, but intended for other adaptation quality factors (performance, etc.).
- Automated evaluation. Adaptation evaluation by using some automated parts. For instance, automating user behaviour simulation as in [P332].
- *Visual and formal validation with a GUI*. Adaptation evaluation by using inspection methods to check heuristics or formal validation rules supported by a GUI (for instance, consistency checks between adaptations).

As aforementioned, every software artefact, such as a user interface or a software architecture, should guarantee a certain degree of quality when delivered in order to prevent the rejection of the final user. Therefore, more concretely, context-aware adaptations should also ensure a certain degree of quality when they are applied to the system at runtime. In this sense, those proposals that apply context-aware adaptations should provide some mechanisms to evaluate them. Nonetheless, we have seen that almost none of the selected papers, except 37 works, considered such evaluation of the quality of adaptation. So, we have confirmed that the maturity of the proposals for adaptation is negligible. Hence, there is a huge gap here to be covered during next years by the research community in order to provide quality context-aware systems.

5.1.8 RQ 4 – Evaluations of software architecture concepts

This RQ was defined to find out what evaluations had been carried out until now to validate the software architecture concepts proposed for context-aware systems. Fig. 15 and Fig. 19 summarize the results to answer this RQ. We categorized the different software architecture concepts found in the selected papers, depending on the type of evaluation presented by using the taxonomy of Wohlin et al. [54]. We also added the category *Example* for papers that did not present a formal empirical strategy but provide some example about the proposal.

As Fig. 19 depicts, *multi-layered architecture, component-based architecture, SOA* and *MAS* were the software architecture concepts having a higher level of maturity. These four architectural concepts were overwhelmingly the most broadly evaluated in the literature. They all were validated by means of examples, surveys, case studies and experiments, but not in the same number. For the multi-layered architecture, examples were the most widely user form of evaluation, followed by case studies, experiments and surveys in the last position. For component-based architecture, the situation was very similar. SOA was evaluated most of the time by means of case studies, followed by examples, surveys and only four experiments. Finally, MAS presented the same results as multi-layered architecture, being example the evaluation strategy more applied, followed by case study, experiment and survey. For these four architectural concepts, there is a good maturity, since we found many papers that used them in their proposals, along with a well-defined validation. Moreover, almost all the experiments identified were done to assess those four software architecture concepts.



SA concepts

Fig. 19. Relationship between SA concepts and evaluation types

Moreover, Fig. 19 illustrates an important fact. For most of the software architecture concepts, there were four or less evaluations, namely for: distributed architecture, P002P, event-triggered architecture, blackboard architecture, microarchitecture, pipe-and-filter and plug-and-play. Aspect-oriented architecture has no evaluation. So, in these cases, we confirm that the maturity is marginal, because of the low number of evaluations and their type, less formal than for other approaches and without experiments at all.

Finally, we also discovered other software architecture concepts with a medium maturity level since they were not enough validated up to now, such as client-server architecture, MDA, middleware architecture and publish/subscribe architecture.

Fig. 15 presents another overview from the point of view of the evaluation type found in each selected paper. Most of the papers only presented an *Example* (29,15%), followed by *Case Study* (22,46%), *Experiment* (10,80%) and finally, *Survey* (4,3%). Thus, the 34,7% of the total of selected papers did not present any type of evaluation or example, neither formal nor informal, for assessing the SA concept proposed.

In conclusion, we have discovered another important gap related to the lack of maturity regarding software architecture concepts when used for developing context-aware systems. We encourage researchers and practitioners to perform more exhaustive and formal validations over these proposals, describing in more detail the participants involved, their number, etc. By doing so, the desired level of maturity will be reached.

5.1.9 Delimitation of context-aware systems from self-adaptive systems

One aspect in our findings that deserves attention is the delimitation of context-aware systems from self-adaptive systems, as the two areas have some overlaps. Context-aware systems are different from self-adaptive systems (and autonomic computing) in a number of respects. A self-adaptive system is a closed-loop system which is able to modify itself automatically at runtime in response to continuous changes of the system and its requirements [34]. Self-* or self-management properties such as self-configuration, self-healing, self-optimization and self-protection aim to reduce efforts to develop and deploy increasingly complex systems [31]. While a self-adaptive system may be able to function without any human intervention, guidance in the form of higher-level objectives is useful and realized in many systems [6]. In contrast to self-adaptive systems, research in context-aware systems is more concerned with how to model, process and manage the context information and less concerned with how a system adapts itself in response to changes in operational context and the adaptation in response to changes of requirements.

On the other hand, studies of context-aware systems and self-adaptive systems reveal common properties like the capability to adapt to changes in their environment [6,33,44]. While context-awareness is a foundational property of both fields, in self-adaptive systems, context-awareness means that the system is aware of its operational environment [31]. In self-adaptive systems, 'context' is more comprehensive and usually extends the set of information that can be used to characterize a situation of an entity. Especially human-centred design and development plays an important role in context-aware systems. Our study clearly indicates that context-aware systems research is predominantly concerned with context information stemming from users and their environment, resulting in systems and applications adapting user's context (e.g. location). The increasing interest in research of recommendation systems, adaptation of user interface or content adaptation emphasizes the differences to self-adaptive systems in general.

To gain an understanding of the differences (boundaries [31]), common features as well as synergies of context-aware systems and self-adaptive systems Hussein et al. [27] investigate context-aware adaptive systems, exemplified by a vehicle routing planning system. While the context information (e.g. vehicle speed) is modelled, processed and managed, to adapt the system to cope with context and requirement changes, the layered architecture also includes monitoring of system-context operational relationships, and a control loop to automatically adapt the system to unanticipated changes (e.g. if the system loses ability to acquire the traffic information).

5.2 Implications for Researchers and Practitioners

This mapping study has revealed that context-aware systems are a growing important research topic, both for HCI and Software Architecture fields. As can be seen in Fig. 3, the interest of the research community in context-aware systems and their architecture has been increasing since 1999 up to now, especially since 2007. We have found 17 relevant papers (as part of the QGS) that propose some kind of framework for context-aware systems (see Section 3.2.2). These important works have influenced others through the years since 1999 until the present day. They have provided a better understanding about context and context-awareness, as well as many advices and recommendations when dealing with context-aware systems in terms of context and user modelling, UI adaptation, and adaptive infrastructures.

This study has shown that there was a real need to analyse which research works highlighted the use of context and its relationship with the software architecture in existing context-aware systems. By doing so, we have provided an overview about the different approaches used in context-aware systems, attending to align the adaptation at both architectural and

HCI levels. In this sense, we have seen the effort invested in improving the use of context in context-aware systems and, especially, by means of their architecture. Moreover, we have also identified gaps existing up to now regarding the deficiencies and weaknesses of the systems when dealing with context and its different dimensions. Therefore, we have documented such effort and gaps by means of this mapping study so researchers and practitioners can make use of the 463 selected papers to improve their research and work. Namely, they can follow the guidelines proposed for modelling, retrieving and managing context and its different dimensions, as well as for knowing how to consider such context in the software architecture in order to provide a good adaptation experience to the final user. We have also offered a list of the most popular publication venues regarding context-aware systems, so researchers may use it for choosing an adequate venue to publish their work. As indicated below, there is a clear trend towards more diversity with regard to application domains and towards more advanced research topics. For practitioners searching for specific papers among our 463 selected papers for a particular problem, we recommend to first follow the requirements of the problem at hand in terms of required dimensions of context (i.e., the answers to RQ2.1-2.3) and if the resulting selection of studies is still too large, we recommend searching among those papers for similar application domains. Our study also indicates clear future trends for researchers coming both from the software architecture and HCI domains, explained in detail in the next section: While early years showed a lot of research on basic aspects of context-aware architectures, the trends are indicating a need for more specialized research into advanced topics and architectures, as well as more application-specific focus.

In this sense, as seen in Fig. 18, independently of the software architecture concepts used for structuring a contextaware system, we have found some important and common elements which should be included in such infrastructure for ensuring a correct context treatment within our system. These components are: *context modelling, context gathering, context inference, context management,* and *context delivery*. Thus, researchers should pay attention to these elements and describe their behaviours and mechanisms for adapting the system. Furthermore, the most advanced software architecture concepts for building context-aware systems are: *multi-layered architecture, component-based architecture, SOA* and *MAS*, and multiple combinations of them. Researchers have a guideline here too that will help them in the task of choosing the most convenient architectural patterns for developing their context-aware systems. As these architectures are more or less equivalent to the state-of-the-art distributed systems architectures in the years of our study, it seems obvious that context-aware systems researchers leverage state-of-the-art distributed systems architectures but do not require specific base-line architectures that are different to the main trends. Consistently in recent years, we see an ongoing trend towards cloud computing, IoT, microservices, etc. which will likely increase both in the distributed systems architectures in general and for context awareness.

With regard to the context of use, researchers here have an extensive analysis of many works which have evidenced the most common terms used for characterizing such context. In addition, we have presented the context dimensions these context terms belong to, such as *User*, *Platform* and *Environment*, as well as other dimensions quite recurrent used in context-aware systems, such as *Historical data*, *User activity* and *Interaction task*. Thus, researchers can take into account this information in order to model context in an appropriate way and to perform adaptations according to these context variables.

Another implication for researchers is the need of providing some quality during the adaptation process. We have detected that there are almost no proposals that focus on evaluating the adaptations applied. This generates strong negative consequences as users may reject the adaptations, and eventually reject the whole system. Hence, more effort should be made at this point by researchers interested in context-aware adaptations with a certain degree of quality.

Apart from evaluating the adaptation, there is still a need to make researchers aware of the task of validating the software architecture concepts they used when building context-aware systems. Especially, if they do not use the most mature ones, i.e. *multi-layered architecture, component-based architecture, SOA* and *MAS*, those validation becomes an indispensable need.

In addition to the above implications, there are other implications derived from this mapping study. For example, researchers and practitioners can pay attention to the application domains identified that make use of context-aware systems (see Section 4.1.3), or the elements most usually adapted by such systems (see Section 4.1.4). Therefore, we encourage both researchers and practitioners to follow the guidelines and the different tips presented here for building context-aware systems and adaptations with a more than acceptable degree of quality.

5.3 Trends

The results of this study indicate a growing interest in research of context-aware systems development, recommendation systems as well as content adaptation. As can be seen, for many sub-fields the four most recent years have more publications than have been published before; while basic topics like context management or service delivery, do rather shrink more advanced topics like decision support or context adaptation, as well as application-focused topics, are on the rise. We also notice that there is a continuous interest in various research topics like context inference and dynamic reconfiguration.

Regarding the application domain we see an ongoing trend in studies investigating mobile environments, contextaware systems research in the area of health-care and studies on smart homes. Smart environments in contrast became less interesting for researchers in recent years. The most interesting finding is a clear trend and growing interest in the domain of Internet-of-Things. Finally, we can also notice the rise of "Other" application domains, indicating the adoption of context-aware systems in various fields of applications and a broadening of the fields of studies in general - which is consistent with the trend towards more application-focused topics noted above.

According to the data, we can infer that an increasing number of studies include usefulness or satisfaction evaluation of context-aware adaptation. Although on a low level, in recent years 28 out of 163 papers published between 2013 and 2017 contributed empirical evidences. Another interesting finding is the shift from software architecture evaluation by surveys or by providing examples to more comprehensive experiments, while case studies remain a proven method to evaluate software architecture aspects of context-aware systems.

6 Validity Threats

This mapping study addresses these validity threats: internal, external, conclusion and construct. They are described in the following sections.

6.1 Internal Validity

Internal validity [50] refers to the use of statistical analysis in order to know how well the data supports the results of the mapping study. In our case, we have only applied basic statistical analysis in the form of bar charts, bubble plots and tag clouds, among others. Moreover, another limitation is related to the likely bias due to the fact that synthesizing the extracted data may be subjective. To mitigate this threat, we defined and revised a review protocol. Additionally, the completeness and performance of the automated search could be also another threat to the internal validity. For this aim, the quasi-sensitivity was calculated and, as shown in Section 3.2.5, the result was 82,4%, higher than the threshold recommended by Zhang et al. [56]. Therefore, we can state that the internal validity of the study is acceptable.

6.2 External Validity

External validity [50] refers to the strength of generalization claims of the study results. The results of this mapping study reflect the state of the art of software architectures for context-aware systems, involving both the HCI and Software Engineering research fields. We have identified 463 relevant papers under these research topics, after an exhaustive search process and after applying a well-defined selection criterion. Therefore, we believe that we have reached an acceptable external validity as our results can be enough generalized, always considering the selected period of time from 1999 to 2017.

6.3 Conclusion Validity

If other researchers replicate the mapping study, the same results would be obtained, that is *Conclusion validity* [50]. We have exhibited very clearly our selection criteria (see Section 3.1) for including or excluding papers in this mapping study. Therefore, we believe that the conclusions and results are valid, and they can be obtained again by using the same research questions. Furthermore, we have followed a systematic mapping study process, along with an exhaustive and well-defined search process based on the Quasi-Gold Standard (see Section 2). The data collection has been also explained in detail, showing all steps followed and the results obtained at every step (see Section 3). Finally, we have clearly described terms and their synonyms in [42].

6.4 Construct Validity

Construct validity [50] refers to the correct interpretation and measurement of the theoretical concepts. In our case, these theoretical concepts are software architecture, context-aware and context dimensions given that we wanted to identify the software architecture concepts used for the development of context-aware systems, and also which context dimensions were involved. We have spent quite a lot of time and effort in studying in detail each concept in order to prepare the most suitable search string for the automated search. We reviewed the literature to find the best words for each theoretical concept, and it was a very challenging task. We had to deal with works from two research fields really different: HCI and Software Architecture. With regard to the concept *software architecture*, we noticed that its name depended on the authors and of course on the research field. The architecture concepts identified ranged from a simple design with not many details to a more complex structure that detail clearly the architectural patterns and styles applied.

In this manner, we had to incorporate to our search string several terms for referring to the architecture of a system in order to cover as many relevant papers as possible.

Something similar happened with other theoretical concepts. For *context-aware* we found also many synonyms in the literature, since it is a popular and widely used concept nowadays, but with different interpretations. We encountered many words, such as adaptive, reactive, responsive and situated, used instead of context-aware, but they were too generic to use them in an isolated way. So that we had to accompany them with other words, like user interface, agent or application, to refine the search. Hence, we included in our search string the most frequently used terms for referring to context-aware.

Finally, regarding the *context dimensions*, we considered the three main ones, accepted by the research community: user, platform and environment. However, we also took into account other widely extended words for referring to context dimensions, such as human, device and task. By doing so, we covered the vast majority of context dimensions that we could find in the literature.

Therefore, we conclude that we have achieved construct validity as the final set of papers is complete thanks to the rigorousness while defining the search string (see Section 3.2.3). We have used not only the theoretical concepts for searching relevant papers, but also similar terms, which provides a high-level of confidence over the search process and its results. Moreover, we have performed the search process based on the Quasi-Gold Standard (see Section 3.2). In our case, we considered 17 papers, the most relevant and referenced ones regarding our research topic, for being part of the QGS. Finally, we verified our results from the automated search against this QGS, obtaining an acceptable quasi-sensitivity (see Section 3.2.5).

7 Conclusions

With this mapping study, we have achieved several goals. We have identified the different *software architecture concepts* proposed for designing and building context-aware systems with the aim of understanding the relationships between them. We have reviewed how *context-based adaptations* are taken into account in those software architecture concepts. Finally, we have also checked the *maturity* of both the existing methods for assessing the context-based adaptations, as well as the maturity of those software architecture concepts. To reach these objectives, we have answered some research questions with the aim of:

- Offering an overview to help researchers and practitioners to analyse which software architecture concepts are the most relevant and the most widely used in context-aware systems.
- Discovering how each dimension of context (User, Platform, Environment and others) is addressed and taken into account within the different architectural proposals to carry out the adaptations based on these dimensions of context.
- Discovering to what extent the evaluation of context-aware adaptations in context-aware systems is covered in the literature.
- Finding out what evaluations have been carried out until now to validate the software architecture concepts proposed for context-aware systems.

This mapping study has shown that there was a real need to analyse which research works highlighted the use of context and its relationship with the software architecture in existing context-aware systems. So, we have provided an overview about the different approaches used in context-aware systems, attending to align the adaptation at both architectural and HCI levels. We have found out the effort invested in improving the use of context in context-aware systems and, especially, within their architecture. Moreover, we have also identified the gaps encountered up to now regarding the deficiencies and weaknesses of systems when dealing with context and its different dimensions. In this sense, it is important to highlight that we have not detected a standard approach for architecting this type of systems. Different commonly used architectural styles are used but without providing guidance about which one should be used and when according to the characteristics of the system to be developed. This gap is highlighted because most of the existing proposals are not published in venues related to software architecture but to pervasive computing and HCI. Therefore, a clear gap exists that the software architecture community should deal with to provide this field with the expected maturity. This need of maturity is also emphasised for the lack of papers that present empirical evaluations that is also an important niche of research for the coming years.

Finally, researchers and practitioners can make use of the results of this mapping study to improve their research and work. They can follow the guidelines proposed for dealing with context and its different dimensions, as well as for knowing how to consider such context for the design of the software architecture. By doing so, they will provide a good adaptation experience to the final user and they will develop context-aware systems and adaptations with higher degree

of quality of adaptation. Further studies could be conducted to complement this mapping with insight about the particularities of the process to design context-aware systems [51], which guide the development of such system.

Acknowledgements

This work was partially supported by Spanish Ministry of Economy, Industry and Competitiveness, State Research Agency / European Regional Development Fund under Vi-SMARt (TIN2016-79100-R) and HA-SYMBIOSIS (TIN2015-72931-EXP). It has also been funded by the Spanish Ministry of Education, Culture and Sport thanks to the FPU scholarship (FPU12/04962). This work was partially supported by Austrian Science Fund (FWF) project ADDCompliance.

References

- G.D. Abowd, A.K. Dey, P.J. Brown, et al., Towards a Better Understanding of Context and Context-Awareness, in: Int. Symp. Handheld Ubiquitous Comput., Springer Berlin / Heidelberg, 1999: pp. 304–307. doi:10.1007/3-540-48157-5_29.
- [2] ACM, ACM Digital Library, (n.d.). http://dl.acm.org/.
- [3] C. Bauer, A Comparison and Validation of 13 Context Meta-Models, in: 20th Eur. Conf. Inf. Syst. (ECIS 2012), 2012. http://aisel.aisnet.org/ecis2012/17.
- [4] A.W. Brown, Model driven architecture : Principles and practice, Softw. Syst. Model. (2004) 314–327. doi:10.1007/s10270-004-0061-2.
- [5] M.G. Brown, Supporting User Mobility, in: Mob. Commun., Springer US, Boston, MA, 1996: pp. 69–77. doi:10.1007/978-0-387-34980-0_8.
- [6] Y. Brun, G. Di Marzo Serugendo, C. Gacek, et al., Engineering Self-Adaptive Systems through Feedback Loops, in: Springer, Berlin, Heidelberg, 2009: pp. 48–70. doi:10.1007/978-3-642-02161-9_3.
- [7] D. Budgen, M. Turner, P. Brereton, B. Kitchenham, Using Mapping Studies in Software Engineering, in: 20th Annu. Work. Psychol. Program. Interes. Group, PPIG 2008, 2008: pp. 195–204. doi:10.1007/978-3-642-02152-7_36.
- [8] A. Cádiz, S. González, K. Mens, Orchestrating context-aware systems: A design perspective, in: 1st Int. Work. Context. Softw. Technol. Appl., 2009: pp. 5–8. doi:10.1145/1595768.1595771.
- [9] F. Carmagnola, F. Cena, L. Console, et al., Tag-based user modeling for social multi-device adaptive guides, User Model. User-Adapt. Interact. 18 (2008) 497–538. doi:10.1007/s11257-008-9052-2.
- [10] D.J. Carmichael, J. Kay, B. Kummerfeld, Consistent modelling of users, devices and sensors in a ubiquitous computing environment, User Model. User-Adapted Interact. 15 (2005) 197–234. doi:10.1007/s11257-005-0001z.
- [11] K. Cheverst, H.E. Byun, D. Fitton, et al., Exploring issues of user model transparency and proactive behaviour in an office environment control system, User Model. User-Adapt. Interact. 15 (2005) 235–273. doi:10.1007/s11257-005-1269-8.
- [12] K. Cooperstock, Jeremy R. Tanikoshi, G. Beirne, T. Narine, W. Buxton, Evolution of a Reactive Environment, in: ACM Conf. Hum. Factors Comput. Syst. (CHI '95), Denver, CO, 1995: pp. 170–177.
- [13] A.K. Dey, G.D. Abowd, D. Salber, A Conceptual Framework and a Toolkit for Supporting the Rapid Prototyping of Context-aware Applications, Human-Computer Interact. 16 (2001) 97–166. doi:10.1207/S15327051HCI16234.
- [14] A.K. Dey, J. Mankoff, Designing mediation for context-aware applications, ACM Trans. Comput. Interact. 12 (2005) 53–80. doi:10.1145/1057237.1057241.
- [15] A.K. Dey, Providing Architectural Support for Building Context-Aware Applications, Georgia Institute of Technology, 2000. doi:10.1207/S15327051HCI16234_02.
- [16] S. Elrod, G. Hall, R. Costanza, M. Dixon, J. Des Rivières, Responsive office environments, Commun. ACM. 36 (1993) 84–85. doi:10.1145/159544.159626.
- [17] Elsevier, Scopus, (n.d.). https://www.scopus.com/.

- [18] S. Fickas, G. Kortuem, Z. Segall, Software Organization for Dynamic and Adaptable Wearable Systems, in: 1st IEEE Int. Symp. Wearable Comput., IEEE, Cambridge, MA, 1997: pp. 56–63.
- [19] Google Inc., Google Scholar, (n.d.). https://scholar.google.com/.
- [20] T. Gu, H.K. Pung, D.Q. Zhang, Toward an OSGi-based infrastructure for context-aware applications, IEEE Pervasive Comput. 3 (2004) 66–74. doi:10.1109/MPRV.2004.19.
- [21] M. Hatala, R. Wakkary, Ontology-based user modeling in an augmented audio reality system for museums, User Model. User-Adapt. Interact. 15 (2005) 339–380. doi:10.1007/s11257-005-2304-5.
- [22] K. Henricksen, J. Indulska, Developing Context-Aware Pervasive Computing Applications: Models and Approach, Pervasive Mob. Comput. 2 (2006) 37–64. doi:10.1016/j.pmcj.2005.07.003.
- [23] J.I. Hong, J.A. Landay, An infrastructure approach to context-aware computing, Human-Computer Interact. 16 (2001) 287–303. doi:10.1207/S15327051HCI16234.
- [24] E. Hudlicka, M.D. McNeese, Assessment of User Affective and Belief States for Interface Adaptation: Application to an Air Force Pilot Task, User Model. User-Adapt. Interact. 12 (2002) 1–47. doi:10.1023/A:1013337427135.
- [25] M.C. Huebscher, J.A. McCann, An adaptive middleware framework for context-aware applications, Pers. Ubiquitous Comput. 10 (2006) 12–20. doi:10.1007/s00779-005-0035-6.
- [26] R. Hull, P. Neaves, J. Bedford-Roberts, Towards situated computing, in: Dig. Pap. First Int. Symp. Wearable Comput., IEEE Comput. Soc, Cambridge, MA, 1997: pp. 146–153. doi:10.1109/ISWC.1997.629931.
- [27] M. Hussein, J. Han, A. Colman, J. Yu, An architecture-based approach to developing context-aware adaptive systems, in: IEEE 19th Int. Conf. Work. Eng. Comput. Syst. ECBS 2012, 2012: pp. 154–163. doi:10.1109/ECBS.2012.13.
- [28] IEEE, IEEE Xplore, (n.d.). http://ieeexplore.ieee.org.
- [29] B. Kitchenham, S. Charters, Guidelines for Performing Systematic Literature Reviews in Software Engineering. Technical Report EBSE 2007-001, 2007. doi:10.1145/1134285.1134500.
- [30] P. Korpipaa, J. Mantyjarvi, J. Kela, H. Keranen, E.-J. Malm, Managing context information in mobile devices, IEEE Pervasive Comput. 2 (2003) 42–51. doi:10.1109/MPRV.2003.1228526.
- [31] C. Krupitzer, F.M. Roth, S. Vansyckel, G. Schiele, C. Becker, A survey on engineering approaches for selfadaptive systems, Pervasive Mob. Comput. 17 (2015) 184–206. doi:10.1016/j.pmcj.2014.09.009.
- [32] M. Kuhrmann, D. Méndez Fernández, M. Tiessler, A mapping study on the feasibility of method engineering, J. Softw. Evol. Process. 26 (2014) 1053–1073. doi:10.1002/smr.1642.
- [33] R. de Lemos, H. Giese, H.A. Müller, et al., Software Engineering for Self-Adaptive Systems: A Second Research Roadmap, in: R. de Lemos, H. Giese, H.A. Müller, M. Shaw (Eds.), Softw. Eng. Self-Adaptive Syst. II Int. Semin. Dagstuhl Castle, Ger. Oct. 24-29, 2010 Revis. Sel. Invit. Pap., Springer Berlin Heidelberg, Berlin, Heidelberg, 2013: pp. 1–32. doi:10.1007/978-3-642-35813-5_1.
- [34] M. Naqvi, Claims and Supporting Evidence for Self-Adaptive Systems A Literature Review, Linnaeus University, 2012. http://lnu.diva-portal.org/smash/record.jsf?pid=diva2%3A512439&dswid=7172.
- [35] D.E. Perry, A.L. Wolf, Foundations for the Study of Software Architecture, ACM Softw. Eng. Notes. 17 (1992) 40–52.
- [36] K. Petersen, R. Feldt, S. Mujtaba, M. Mattsson, Systematic Mapping Studies in Software Engineering, in: 12th Int. Conf. Eval. Assess. Softw. Eng., 2007: pp. 68–77.
- [37] D. Petrelli, E. Not, User-centred design of flexible hypermedia for a mobile guide: Reflections on the hyperAudio experience, User Model. User-Adapt. Interact. 15 (2005) 303–338. doi:10.1007/s11257-005-8816-1.
- [38] H. Pung, T. Gu, W. Xue, et al., Context-aware middleware for pervasive elderly homecare, IEEE J. Sel. Areas Commun. 27 (2009) 510–524. doi:10.1109/JSAC.2009.090513.
- [39] M. Raento, A. Oulasvirta, R. Petit, H. Toivonen, ContextPhone: A prototyping platform for context-aware mobile applications, IEEE Pervasive Comput. 4 (2005) 51–59. doi:10.1109/MPRV.2005.29.
- [40] J. Rekimoto, Y. Ayatsuka, K. Hayashi, Augment-able Reality: Situated Communication through Physical and Digital Spaces, in: 2nd IEEE Int. Symp. Wearable Comput., IEEE, Pittsburgh, PA, 1998: pp. 68–75. doi:10.1109/ISWC.1998.729531.

- [41] D. Riboni, C. Bettini, COSAR: Hybrid reasoning for context-Aware activity recognition, Pers. Ubiquitous Comput. 15 (2011) 271–289. doi:10.1007/s00779-010-0331-7.
- [42] C. Roda, E. Navarro, U. Zdun, V. Lopez-Jaquero, G. Simhandl, Glossary of terms for characterizing contextaware systems, 2018. https://www.dsi.uclm.es/intranet/trep.php?codtrep=DIAB-18-07-1.
- [43] M. Román, C. Hess, R. Cerqueira, et al., A middleware infrastructure for active spaces, IEEE Pervasive Comput. 1 (2002) 74–83. doi:10.1109/MPRV.2002.1158281.
- [44] M. Salehie, L. Tahvildari, Towards a Goal-driven Approach to Action Selection in Self-adaptive Software, Softw.
 Pr. Exper. 42 (2012) 211–233. doi:10.1002/spe.1066.
- [45] A. Schmidt, C. Winterhalter, User context aware delivery of e-learning material: Approach and architecture, J. Univers. Comput. Sci. 10 (2004) 38–46. https://www.scopus.com/inward/record.uri?eid=2-s2.0-13244258859&partnerID=40&md5=be4194c8542459952cd43a7e18ad7e7b.
- [46] A. Schmidt, K.A. Aidoo, A. Takaluoma, et al., Advanced interaction in context, in: Int. Symp. Handheld Ubiquitous Comput., Springer Berlin / Heidelberg, 1999: pp. 89–101.
- [47] M.J. van Sinderen, A.T. van Halteren, M. Wegdam, H.B. Meeuwissen, E.H. Eertink, Supporting context-aware mobile applications: An infrastructure approach, IEEE Commun. Mag. 44 (2006) 96–104. doi:10.1109/MCOM.2006.1705985.
- [48] Springer, Springer Link, (n.d.). http://link.springer.com/.
- [49] O. Stock, M. Zancanaro, P. Busetta, et al., Adaptive, intelligent presentation of information for the museum visitor in PEACH, User Model. User-Adapt. Interact. 17 (2007) 257–304. doi:10.1007/s11257-007-9029-6.
- [50] D. Tofan, M. Galster, P. Avgeriou, W. Schuitema, Past and future of software architectural decisions A systematic mapping study, Inf. Softw. Technol. 56 (2014) 850–872. doi:10.1016/j.infsof.2014.03.009.
- [51] V. Vieira, P. Tedesco, A.C. Salgado, Designing context-sensitive systems: An integrated approach, Expert Syst. Appl. 38 (2011) 1119–1138. doi:10.1016/j.eswa.2010.05.006.
- [52] R. Wieringa, N. Maiden, N. Mead, C. Rolland, Requirements engineering paper classification and evaluation criteria : a proposal and a discussion, Requir. Eng. 11 (2006) 102–107. doi:10.1007/s00766-005-0021-6.
- [53] C. Wohlin, Guidelines for Snowballing in Systematic Literature Studies and a Replication in Software Engineering, in: 18th Int. Conf. Eval. Assess. Softw. Eng. (EASE 2014), 2014: pp. 1–10. doi:10.1145/2601248.2601268.
- [54] C. Wohlin, P. Runeson, M. Höst, et al., Experimentation in Software Engineering, 1st ed., Springer Berlin Heidelberg, Berlin, Heidelberg, 2012. http://link.springer.com/10.1007/978-3-642-29044-2 (accessed February 3, 2014).
- [55] H. Zhang, M. Babar, On Searching Relevant Studies in Software Engineering, in: 14th Int. Conf. Eval. Assess. Softw. Eng., 2010: pp. 1–10. http://www.bcs.org/upload/pdf/ewic_ea10_session5paper1.pdf.
- [56] H. Zhang, M.A. Babar, P. Tell, Identifying Relevant Studies in Software Engineering, Inf. Softw. Technol. 53 (2011) 625–637. doi:10.1016/j.infsof.2010.12.010.

Appendix A. Selected Papers

The 463 selected papers included in this mapping study are listed below (ordered by author). Notice that the papers included in the QGS (see Section 3.2.2) are in bold, e.g. **[P002]**.

- [P001] Abbasi, A.Z., Ahsan, M.U., Shaikh, Z.A., Nasir, Z. (2010), CAWD: A tool for designing context-aware workflows, International Conference on Software Engineering and Data Mining, 128-133.
- [P002] Abowd, G.D., Dey, A.K., Brown, P.J., Davies, N., Smith, M., Steggles, P. (1999), Towards a Better Understanding of Context and Context-Awareness, *International Symposium on Handheld and Ubiquitous Computing*, 304-307, DOI: 10.1007/3-540-48157-5_29.
- [P003] Abusair, M., Sharaf, M., Muccini, H., Inverardi, P. (2017), Adaptation for Situational-aware Cyber-physical Systems Driven by Energy Consumption and Human Safety, *Proceedings of the 11th European Conference on Software Architecture: Companion Proceedings*, 78-84, ACM, DOI: 10.1145/3129790.3129792.

- [P004] Achilleos, A.P., Thoma, M., Kapitsaki, G.M., Mettouris, C., Papadopoulos, G.A. (2015), An automation component for crossplatform, context-aware applications development, *International and Interdisciplinary Conference on Modeling and Using Context*, 63-76, DOI: 10.1007/978-3-319-25591-0_5.
- [P005] Adel, A., Laborie, S., Roose, P. (2014), Semantic context-aware adaptation platform architecture, *Procedia Computer Science*, 32, 959-964, DOI: 10.1016/j.procs.2014.05.518.
- [P006] Adipat, B., Zhang, D. (2005), Developing adaptive and personalized mobile applications: A framework and design issues, *Americas Conference on Information Systems*, 1862-1868.
- [P007] Ahn, C., Nah, Y. (2010), Design of location-based web service framework for Context-Aware applications in ubiquitous environments, *International Conference on Sensor Networks*, *Ubiquitous, and Trustworthy Computing*, 426-433, DOI: 10.1109/SUTC.2010.26.
- [P008] Ajanki, A., Billinghurst, M., Gamper, H., Järvenpää, T., Kandemir, M., Kaski, S., Koskela, M., Kurimo, M., Laaksonen, J., Puolamäki, K., Ruokolainen, T., Tossavainen, T. (2011), An augmented reality interface to contextual information, *Virtual Reality*, 15, (2), 161-173, DOI: 10.1007/s10055-010-0183-5.
- [P009] Alanezi, K., Mishra, S. (2015), Design, implementation and evaluation of a smartphone position discovery service for accurate context sensing, *Computers and Electrical Engineering*, 44, 307-323, DOI: 10.1016/j.compeleceng.2015.01.015.
- [P010] Alférez, G.H., Pelechano, V. (2011), Context-aware autonomous web services in software product lines, International Software Product Line Conference, 100-109, DOI: 10.1109/SPLC.2011.21.
- [P011] Alfonso-Cendón, J., Fernández-de-Alba, J.M., Fuentes-Fernández, R., Pavón, J. (2016), Implementation of context-aware workflows with multi-agent systems, *Neurocomputing*, 176, 91-97, Inderscience Enterprises Ltd., DOI: 10.1016/j.neucom.2014.10.098.
- [P012] Alhamid, M.F., Rawashdeh, M., Al Osman, H., Hossain, M.S., El Saddik, A. (2015), Towards context-sensitive collaborative media recommender system, *Multimedia Tools and Applications*, 74, (24), 11399-11428, DOI: 10.1007/s11042-014-2236-3.
- [P013] Alhamid, M.F., Rawashdeh, M., Dong H., Hossain, M.A., Alelaiwi, A., El Saddik, A. (2016), RecAm: a collaborative contextaware framework for multimedia recommendations in an ambient intelligence environment, *Multimedia Systems*, 22, (5), 587-601, Rinton Press, Incorporated, DOI: 10.1007/s00530-015-0469-2.
- [P014] Alharthi, R., Albalawi, R., Abdo, M., El Saddik, A. (2011), A context-aware e-health framework for students with moderate intellectual and learning disabilities, *International Conference on Multimedia and Expo*, DOI: 10.1109/ICME.2011.6012218.
- [P015] Allègre, W., Burger, T., Antoine, J.Y., Berruet, P., Departe, J.P. (2013), A non-intrusive context-aware system for ambient assisted living in smart home, *Health and Technology*, 3, (2), 129-138, DOI: 10.1007/s12553-013-0050-y.
- [P016] Al-Soh, M., Zualkernan, I.A. (2017), An MQTT-Based Context-Aware Wearable Assessment Platform for Smart Watches, Proceedings - IEEE 17th International Conference on Advanced Learning Technologies, ICALT 2017, 98-100, Elsevier, DOI: 10.1109/ICALT.2017.140.
- [P017] Anand, P. (2016), Enabling Context-Aware Computing in Internet of Things Using M2M, Proceedings 2015 IEEE International Symposium on Nanoelectronic and Information Systems, iNIS 2015, 219-224, Elsevier B.V., DOI: 10.1109/iNIS.2015.13.
- [P018] Anatoli Dimitrov, K. N., Zheng, X., Xiao, F. (2017), CAMDroid: An Adaptation Framework for Android Context-Aware Multitasking, *Proceedings of the First International Workshop on Human-centered Sensing, Networking, and Systems*, 55-60, Springer-Verlag, DOI: 10.1145/3144730.3144740.
- [P019] Anderson, C., Suarez, I., Xu, Y., David, K. (2015), An ontology-based reasoning framework for context-aware applications, International and Interdisciplinary Conference on Modeling and Using Context, 471-476, DOI: 10.1007/978-3-319-25591-0_34.
- [P020] André, F., Daubert, E., Gauvrit, G. (2010), Towards a generic context-aware framework for self-adaptation of service-oriented architectures, *International Conference on Internet and Web Applications and Services*, 309-314, DOI: 10.1109/ICIW.2010.52.
- [P021] Angarita R., Manouvrier M., Rukoz M. (2016), An agent architecture to enable self-healing and context-aware web of things applications, *IoTBD 2016 - Proceedings of the International Conference on Internet of Things and Big Data*, 82-87, Institute of Electrical and Electronics Engineers Inc..
- [P022] Angelaccio M., Buttarazzi B. (2016), STREET web: A framework for "web on the road" smart services, WEBIST 2016 -Proceedings of the 12th International Conference on Web Information Systems and Technologies, 1, 299-302, Springer Verlag.
- [P023] Ardissono, L., Goy, A., Petrone, G., (2008), A framework for the development of distributed, context-aware adaptive hypermedia applications, *International Conference on Adaptive Hypermedia and Adaptive Web-Based Systems*, 259-262, DOI: 10.1007/978-3-540-70987-9_29.
- [P024] Ashibani Y., Kauling D., Mahmoud Q.H. (2017), A context-aware authentication framework for smart homes, *Canadian Conference on Electrical and Computer Engineering*, DOI: 10.1109/CCECE.2017.7946657.
- [P025] Autili, M., Caporuscio, M., Issarny, V., Berardinelli, L. (2014), Model-driven engineering of middleware-based ubiquitous services, *Software and Systems Modeling*, 13, (2), 481-511, DOI: 10.1007/s10270-013-0344-6.
- [P026] Ayala, I., Amor, M., Fuentes, L. (2013), Self-configuring agents for ambient assisted living applications, Personal and Ubiquitous Computing, 17, (6), 1159-1169, DOI: 10.1007/s00779-012-0555-9.
- [P027] Ayed, D., Taconet, C., Bernard, G., Berbers, Y. (2008), CADeComp: Context-aware deployment of component-based applications, *Journal of Network and Computer Applications*, 31, (3), 224-257, DOI: 10.1016/j.jnca.2006.12.002.

- [P028] Babaei, Z., Rahmani, A.M., Rezaei, A. (2016), Real-time reusable event-driven architecture for context aware systems, 2016 24th Iranian Conference on Electrical Engineering, ICEE 2016, 294-299, Emerald Group Publishing Ltd., DOI: 10.1109/IranianCEE.2016.7585535.
- [P029] Babu, A.B.K.A., Sivakumar, R. (2014), Development of ontology based middleware for context awareness in ambient intelligence, *International Conference on Hybrid Intelligent Systems*, 219-224, DOI: 10.1109/HIS.2014.7086202.
- [P030] Badii, A., Crouch, M., Lallah, C. (2010), A Context-Awareness Framework for Intelligent Networked Embedded Systems, International Conference on Advances in Human-Oriented and Personalized Mechanisms, Technologies and Services, 105-110, IEEE Computer Society, DOI: 10.1109/CENTRIC.2010.29.
- [P031] Bajwa, A., Farooq, S., Malik, O., Khalique, S., Suguri, H., Ahmad, H.F., Ali, A. (2006), Persistent architecture for context aware lightweight multi-agent system, *International Workshop on Programming Multi-Agent Systems*, 57-69.
- [P032] Baladrón, C., Aguiar, J.M., Gobernado, J., Carro, B., Sánchez, A. (2010), User-driven context aware creation and execution of home care applications, *Annales des Telecommunications/Annals of Telecommunications*, 65, (9), 545-556, DOI: 10.1007/s12243-010-0164-4.
- [P033] Bandini, S., Sarini, M., Simone, C., Vizzari, G. (2007), WWW in the small : Towards sustainable Adaptivity, World Wide Web, 10, (4), 471-501, DOI: 10.1007/s11280-007-0024-y.
- [P034] Banos, O., Amin, M.B., Khan, W.A., Ali, T., Afzal, M., Kang, B.H. (2015), Mining Minds: An Innovative Framework for Personalized Health and Wellness Support, *International Conference on Pervasive Computing Technologies for Healthcare*, 1-8, ICST (Institute for Computer Sciences, Social-Informatics and Telecommunications Engineering).
- [P035] Barakonyi, I., Schmalstieg, D. (2008), Augmented reality agents for user interface adaptation, *Computer Animation and Virtual Worlds*, 19, (1), 23-35, DOI: 10.1002/cav.220.
- [P036] Bardram, J. E., Hansen, T. R. (2010), Context-Based Workplace Awareness, Computer Supported Cooperative Work, 19, (2), 105-138, DOI: 10.1007/s10606-010-9110-2.
- [P037] Bardram, J.E. (2009), A novel approach for creating activity-aware applications in a hospital environment, *Conference on Human-Computer Interaction*, 731-744, DOI: 10.1007/978-3-642-03658-3_78.
- [P038] Ben Lahmar, I., Belaïd, D., Mukhtar, H. (2012), Middleware for task resolution and adaptation in pervasive environments, *Evolving Systems*, 3, (4), 221-233, DOI: 10.1007/s12530-012-9057-2.
- [P039] Benali A., El Asri B., Kriouile H. (2016), Cloud environment assignment: A context-aware and Dynamic Software Product Lines-based approach, *Proceedings of IEEE/ACS International Conference on Computer Systems and Applications, AICCSA*, 2016-July, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1109/AICCSA.2015.7507225.
- [P040] Bernardos, A.M., Madrazo, E., Casar, J.R. (2010), An embeddable fusion framework to manage context information in mobile devices, *International Conference on Hybrid Artificial Intelligence Systems*, 468-477, DOI: 10.1007/978-3-642-13803-4_58.
- [P041] Bernardos, A.M., Tarrío, P., Casar, J.R. (2009), CASanDRA: A framework to provide context acquisition services ANd reasoning algorithms for ambient intelligence applications, *International Conference on Parallel and Distributed Computing*, *Applications and Technologies*, 372-377, DOI: 10.1109/PDCAT.2009.51.
- [P042] Bettini, C., Maggiorini, D., Riboni, D. (2007), Distributed Context Monitoring for the Adaptation of Continuous Services, World Wide Web, 10, (4), 503-528, DOI: 10.1007/s11280-007-0028-7.
- [P043] Betz, M., Huq, M., Pipek, V., Rohde, M., Stevens, G., Englert, R., Wulf, V. (2007), An architecture for adaptive and adaptable mobile applications for physically handicapped people, *International Conference on Universal Access in Human-Computer Interaction*, 335-344, DOI: 10.1007/978-3-540-73279-2_37.
- [P044] Bezerra Chagas, A., Guimarães Ferraz C. A. (2012), ConnecTV: An Environment to Develop and Run Component-based Customized idTV Applications, *Proceedings of the 18th Brazilian Symposium on Multimedia and the Web*, 83-90, ACM, DOI: 10.1145/2382636.2382657.
- [P045] Bobek, S., Nalepa, G.J., Adrian, W.T. (2013), Mobile Context-Based Framework for Monitoring Threats in Urban Environment, International Conference on Multimedia Communications, Services and Security, 25-35, DOI: 10.1007/978-3-642-38559-9_3.
- [P046] Bongartz S., Jin Y., Paternò F., Rett J., Santoro C., Spano L.D. (2012), Adaptive user interfaces for smart environments with the support of model-based languages, *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, 7683 LNCS, 33-48, Springer Verlag.
- [P047] Bonte, P., Ongenae, F., De Backere, F., Schaballie, J., Arndt, D., Verstichel, S., Mannens, E., Van de Walle, R., De Turck, F. (2017), The MASSIF platform: a modular and semantic platform for the development of flexible IoT services, *Knowledge and Information Systems*, 51, (1), Elsevier B.V., DOI: 10.1007/s10115-016-0969-1.
- [P048] Bortenschlager, M., Castelli, G., Rosi, A., Zambonelli, F. (2009), A Context-sensitive Infrastructure for Coordinating Agents in Ubiquitous Environments, *Multiagent and Grid Systems*, 5, (1), 1-18, IOS Press.
- [P049] Bouguessa, A., Boudaa, B., Mebarki, L.A. (2014), Adaptation of service-based context-aware applications with FraSCAti platform, *International Conference on Advanced Aspects of Software Engineering*, 171-179.
- [P050] Bucchiarone, A., Annapaola, M., Pistore, M., Raik, H. (2017), A context-aware framework for dynamic composition of process fragments in the internet of services, *Journal of Internet Services and Applications*, 8, (1), DOI: 10.1186/s13174-017-0057-0.
- [P051] Buhagiar, A.J., Pace, G.J., Ebejer, J.-P. (2017), Engineering adaptive user interfaces using monitoring-oriented programming, Proceedings - 2017 IEEE International Conference on Software Quality, Reliability and Security, QRS 2017, 200-207, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1109/QRS.2017.30.

- [P052] Bürkle, A., Müller, W., Pfirrmann, U., Dimakis, N., Soldatos, J., Polymenakos, L. (2007), An agent-based framework for context-aware services, *International Conference on Universal Access in Human-Computer Interaction*, 53-62.
- [P053] Burns, W., Chen L., Nugent, C., Donnelly, M., Skillen, K.-L., Solheim, I. (2012), A conceptual framework for supporting adaptive personalized help-on-demand services, *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, 7683 LNCS, 427-432.
- [P054] Cádiz, A., González, S., Mens, K. (2009), Orchestrating context-aware systems: A design perspective, International Workshop on Context-aware software technology and applications, 5-8, DOI: 10.1145/1595768.1595771.
- [P055] Cagalaban, G., Kim, S. (2010), Context-aware service framework for decision-support applications using ontology-based modeling, *Pacific Rim Knowledge Acquisition Workshop*, 103-110, DOI: 10.1007/978-3-642-15037-1_10.
- [P056] Cámara, J., Canal, C., Cubo, J., Murillo, J.M. (2007), Enabling adaptivity in user interfaces, *European Conference on Software Architecture*, 106-114.
- [P057] Cao, D., Nie, W., Li, R., Li, Z., Wang, T., Zeng, C. (2012), Context-aware recommendation in mobile environments: An approach based on interest resonance, *Wuhan University Journal of Natural Sciences*, 17, (5), 400-406, DOI: 10.1007/s11859-012-0861-0.
- [P058] Cao, Z., Zhang, X., Zhang, W., Xie, X., Shi, J., Xu, H. (2015), A context-aware adaptive web service composition framework, *International Conference on Computational Intelligence and Communication Technology*, 62-66, DOI: 10.1109/CICT.2015.68.
- [P059] Capilla, R. (2006), Context-aware architectures for building service-oriented systems, European Conference on Software Maintenance and Reengineering, 300-303, DOI: 10.1109/CSMR.2006.22.
- [P060] Carmagnola, F., Cena, F., Console, L., Cortassa, O., Gena, C., Goy, A., Torre, I., Toso, A., Vernero, F. (2008), Tag-based user modeling for social multi-device adaptive guides, User Modeling and User-Adapted Interaction, 18, (5), 497-538, DOI: 10.1007/s11257-008-9052-2.
- [P061] Carmichael, D.J., Kay, J., Kummerfeld, B. (2005), Consistent Modelling of Users, Devices and Sensors in a Ubiquitous Computing Environment, User Modeling and User-Adapted Interaction, 15, (3), 197-234, DOI: 10.1007/s11257-005-0001-z.
- [P062] Castro Garrido, P., Luque Ruiz, I., Gómez-Nieto, M.Á. (2014), OBCAS: an agent-based system and ontology for mobile context aware interactions, *Journal of Intelligent Information Systems*, 43, (1), 33-57, DOI: 10.1007/s10844-014-0305-8.
- [P063] Celikkan, U., Kurtel, K. (2015), A platform for context-aware application development: PCAD, Federated Conference on Computer Science and Information Systems, 1481-1488, DOI: 10.15439/2015F49.
- [P064] Ceri, S., Daniel, F., Facca, F.M., Matera, M. (2007), Model-driven Engineering of Active Context-awareness, World Wide Web, 10, (4), 387-413, DOI: 10.1007/s11280-006-0014-5.
- [P065] Chaari, T., Ejigu, D., Laforest, F., Scuturici, V.-M. (2007), A comprehensive approach to model and use context for adapting applications in pervasive environments, *Journal of Systems and Software*, 80, (12 SPEC. ISS.), 1973-1992, DOI: 10.1016/j.jss.2007.03.010.
- [P066] Chabridon, S., Conan, D., Abid, Z., Taconet, C. (2013), Building ubiquitous QoC-aware applications through model-driven software engineering, *Science of Computer Programming*, 78, (10), 1912-1929, DOI: 10.1016/j.scico.2012.07.019.
- [P067] Challiol, C., Fortier, A., Gordillo, S., Rossi, G. (2008), Architectural and Implementation Issues for a Context-aware Hypermedia Platform, J. Mob. Multimed., 4, (2), 118-138, Springer-Verlag.
- [P068] Chan, C.S., Ostertag, M.H., Akyürek, A.S., Rosing, T.S. (2017), Context-aware system design, Proceedings of SPIE The International Society for Optical Engineering, 10194, DOI: 10.1117/12.2263232.
- [P069] Chang, C. C., Tseng, J.C.R., Lin, K. J. (2008), A dynamic capability framework for context-aware mobile services, Conference on E-Commerce Technology and Enterprise Computing, E-Commerce and E-Services, 183-189, DOI: 10.1109/CECandEEE.2008.121.
- [P070] Chang, J., Na, S., Yoon, M. (2008), Intelligent context-aware system architecture in pervasive computing environment, International Symposium on Parallel and Distributed Processing with Applications, 745-750, DOI: 10.1109/ISPA.2008.73.
- [P071] Chang, J.W., Lee, H.J. (2007), Context-aware architecture for intelligent application services in ubiquitous computing, International Conference on Semantic Computing, 275-281, DOI: 10.1109/ICSC.2007.65.
- [P072] Chen, P. W., Chen, Y. H., Wu, Y. H. (2017), Constructing a user-centric iot framework for software integration in a smart home environment, 2017 10th International Conference on Ubi-media Computing and Workshops (Ubi-Media), 1-6, Springer Verlag, DOI: 10.1109/UMEDIA.2017.8074147.
- [P073] Chen, X., Chen, S., Zeng X., Zheng, X., Zhang, Y., Rong, C. (2017), Framework for context-aware computation offloading in mobile cloud computing, *Journal of Cloud Computing*, 6, (1), Institute of Electrical and Electronics Engineers Inc., DOI: 10.1186/s13677-016-0071-y.
- [P074] Cheng, S.-W., Sun, S.-Q. (2009), Adaptive 3D navigation user interface design based on rough sets, International Conference on Computer-Aided Industrial Design & Conceptual Design, 1935-1940, DOI: 10.1109/CAIDCD.2009.5375404.
- [P075] Cheverst, K., Byun, H.E., Fitton, D., Sas, C., Kray, C., Villar, N. (2005), Exploring Issues of User Model Transparency and Proactive Behaviour in an Office Environment Control System, User Modeling and User-Adapted Interaction, 15, (3), 235-273, DOI: 10.1007/s11257-005-1269-8.
- [P076] Chien, B.-C., He, S.-Y. (2010), A genetic context interpreter for context-aware systems in pervasive computing environments, International Conference on Grid and Pervasive Computing, 91-100, DOI: 10.1007/978-3-642-13067-0_13.

- [P077] Chien, B.-C., Tsai, H.-C., Hsueh, Y.-K. (2009), CADBA: A context-aware architecture based on context database for mobile computing, Symposia and Workshops on Ubiquitous, Autonomic and Trusted Computing, 367-372, DOI: 10.1109/UIC-ATC.2009.25.
- [P078] Chihani, B., Bertin, E., Crespi, N. (2014), Programmable context awareness framework, *Journal of Systems and Software*, 92, (1), 59-70, DOI: 10.1016/j.jss.2013.07.046.
- [P079] Choi, J. (2008), Software architecture for extensible context-aware systems, International Conference on Convergence and Hybrid Information Technology, 811-816, DOI: 10.1109/ICHIT.2008.236.
- [P080] Choi, J., Moon, J. (2008), MyGuide: A mobile context-aware exhibit guide system, *International Conference on Computational Science and Its Applications*, 348-359, DOI: 10.1007/978-3-540-69848-7_28.
- [P081] Choi, J.H., Choi, S.Y., Shin, D., Shin, D. (2004), Research and implementation of the context-aware middleware based on neural network, *International Conference on AI, Simulation, and Planning in High Autonomy Systems*, 295-303.
- [P082] Chun, I., Park, J., Lee, H., Kim, W., Park, S., Lee, E. (2013), An agent-based self-adaptation architecture for implementing smart devices in Smart Space, *Telecommunication Systems*, 52, (4), 2335-2346, DOI: 10.1007/s11235-011-9547-8.
- [P083] Chun-dong, W.C.W., Xiao-qin, L.X.L., Huai-bin, W.H.W. (2009), A framework of intelligent agent based middleware for context aware computing, *International Conference on Natural Computation*, 107-110, DOI: 10.1109/ICNC.2009.388.
- [P084] Cianciulli, D., Canfora, G., Zimeo, E. (2017), Beacon-based context-aware architecture for crowd sensing public transportation scheduling and user habits, *Procedia Computer Science*, 109, 1110-1115, DOI: 10.1016/j.procs.2017.05.451.
- [P085] Clerckx, T., Vandervelpen, C., Luyten, K., Coninx, K. (2006), A task-driven user interface architecture for ambient intelligent environments, *International Conference on Intelligent User Interfaces*, 309-311, DOI: 10.1145/1111449.1111520.
- [P086] Cobaleda, L.V., Mazo, R., Risco Becerra, J.L., Duitama, J.F. (2017), Reference Software Architecture for Improving Modifiability of Personalised Web Applications - a Controlled Experiment, *Int. J. Web Eng. Technol.*, 11, (4), 351-370, Association for Computing Machinery, DOI: 10.1504/IJWET.2016.081768.
- [P087] Coronato, A., De Pietro, G. (2013), Situation Awareness in Applications of Ambient Assisted Living for Cognitive Impaired People, *Mobile Networks and Applications*, 18, (3), 444-453, DOI: 10.1007/s11036-012-0409-8.
- [P088] Corvetta, F., Matera, M., Medana, R., Quintarelli, E., Rizzo, V., Tanca, L. (2015), Designing and Developing Context-Aware Mobile Mashups: The CAMUS Approach, *International Conference on Engineering the Web in the Big Data Era*, 651-654, Springer-Verlag New York, Inc., DOI: 10.1007/978-3-319-19890-3_49.
- [P089] Coutaz, J. (2010), User Interface Plasticity: Model Driven Engineering to the Limit!, Proceedings of the 2Nd ACM SIGCHI Symposium on Engineering Interactive Computing Systems, 1-8, IEEE Computer Society, DOI: 10.1145/1822018.1822019.
- [P090] Coutaz, J., Balme, L., Alvaro, X., Calvary, G., Demeure, A.,Sottet, J.-S. (2007), An MDE-SOA Approach to Support Plastic User Interfaces in Ambient Spaces, *International Conference on Universal Access in Human-computer Interaction*, 63-72, Springer-Verlag.
- [P091] Cunha, C.R., Peres, E., Morais, R., Oliveira, A.A., Matos, S.G., Fernandes, M.A., Ferreira, P.J.S.G., Reis, M.J.C.S. (2010), The Use of Mobile Devices with Multi-tag Technologies for an Overall Contextualized Vineyard Management, *Computers and Electronics in Agriculture*, 73, (2), 154-164, Elsevier Science Publishers B. V., DOI: 10.1016/j.compag.2010.05.007.
- [P092] Curiel, P., Lago, A.B. (2012), An infrastructure to provide context-aware information and services to mobile users, International Conference on Ubiquitous Computing and Ambient Intelligence, 233-240, DOI: 10.1007/978-3-642-35377-2_32.
- [P093] Cuřín, J., Kleindienst, J. (2008), SitCom: Virtual smart-room environment for multi-modal perceptual systems, *International Conference on Industrial, Engineering and Other Applications of Applied Intelligent Systems*, 476-485, DOI: 10.1007/978-3-540-69052-8_50.
- [P094] Da, K., Roose, P., Dalmau, M., Nevado, J., Karchoud, R. (2014), Kali2Much: A context middleware for autonomic adaptationdriven platform, Workshop on Middleware for Context-Aware Applications in the IoT, 25-30, DOI: 10.1145/2676743.2676748.
- [P095] Daniele, L.M., Silva, E., Pires, L.F., Van Sinderen, M. (2009), A SOA-based platform-specific framework for context-aware mobile applications, *International Workshop on Enterprise Interoperability*, 25-37, DOI: 10.1007/978-3-642-04750-3_3.
- [P096] Davidyuk, O., Riekki, J., Rautio, V.-M., Sun, J. (2004), Context-aware middleware for mobile multimedia applications, International Conference on Mobile and Ubiquitous Multimedia, 213-220, DOI: 10.1145/1052380.1052410.
- [P097] De Alencar, T.S., Machado, L.R., De Oliveira Neris, L., De Almeida Neris, V.P. (2014), Addressing the users' diversity in ubiquitous environments through a low cost architecture, *International Conference on Universal Access in Human-Computer Interaction*, 439-450, DOI: 10.1007/978-3-319-07446-7_43.
- [P098] De Carolis, B., Cozzolongo, G., Pizzutilo, S., Silvestri, V. (2007), MyMap: Generating personalized tourist descriptions, *Applied Intelligence*, 26, (2), 111-124, DOI: 10.1007/s10489-006-0012-4.
- [P099] De Rocha, R.C.A., Endler, M. (2006), Middleware: Context Management in Heterogeneous, Evolving Ubiquitous Environments, *IEEE Distributed Systems Online*, 7, (4), 1-13, IEEE Educational Activities Department, DOI: 10.1109/MDSO.2006.28.
- [P100] De Virgilio, R. (2012), AML: a modeling language for designing adaptive web applications, Personal and Ubiquitous Computing, 16, (5), 527-541, DOI: 10.1007/s00779-011-0418-9.
- [P101] Deborah, R.N., Chitrakala, S. (2016), A context-aware approach based web service recommendation, Proceeding of IEEE -2nd International Conference on Advances in Electrical, Electronics, Information, Communication and Bio-Informatics, IEEE - AEEICB 2016, 188-192, IEEE Computer Society, DOI: 10.1109/AEEICB.2016.7538270.

- [P102] Delicato, F.C., Santos, I.L.A., Pires, P.F., Oliveira, A.L.S., Batista, T., Pirmez, L. (2009), Using aspects and dynamic composition to provide context-aware adaptation for mobile applications, *Symposium on Applied Computing*, 456-460, DOI: 10.1145/1529282.1529381.
- [P103] Desruelle, H., Isenberg, S., Botsikas, A., Vergori, P., Gielen, F. (2016), Accessible user interface support for multi-device ubiquitous applications: architectural modifiability considerations, *Universal Access in the Information Society*, 15, (1), IEEE Early Access Articles, DOI: 10.1007/s10209-014-0373-0.
- [P104] Desruelle, H., Isenberg, S., Lyle, J., Gielen, F. (2013), Multi-device application middleware: leveraging the ubiquity of the Web with webinos, *The Journal of Supercomputing*, 66, (1), 4-20, DOI: 10.1007/s11227-013-0901-3.
- [P105] Deva, B., Garzon, S.R., Schunemann, S. (2016), A Context-Sensitive Privacy-Aware Framework for Proactive Location-Based Services, Proceedings - NGMAST 2015: The 9th International Conference on Next Generation Mobile Applications, Services and Technologies, 138-143, SciTePress, DOI: 10.1109/NGMAST.2015.27.
- [P106] Devaraju, A., Hoh, S., Hartley, M. (2007), A context gathering framework for context-aware mobile solutions, *International Conference on mobile technology, applications, and systems and International symposium on Computer human interaction in mobile technology*, 39-46, DOI: 10.1145/1378063.1378070.
- [P107] Dey, A. K., Mankoff, J. (2005), Designing Mediation for Context-aware Applications, *Transactions on Computer-Human Interaction*, 12, (1), 53-80, ACM, DOI: 10.1145/1057237.1057241.
- [P108] Dey, A.K., Abowd, G.D., Salber, D. (2001), A Conceptual Framework and a Toolkit for Supporting the Rapid Prototyping of Context-Aware Applications, *Human-Computer Interaction*, 16, (2), 97-166, DOI: 10.1207/S15327051HCI16234_02.
- [P109] Dharanyadevi, P., Dhavachelvan, P., Jayakumar, S.K.V., Sujatha, P., Baskaran, R., Venkatachalapathy, V.S.K. (2011), Context aware SOA based discovery in mobile environment by bi-matchmaking technique, *International Conference on Recent Trends* in Information Technology, 618-623, DOI: 10.1109/ICRTIT.2011.5972326.
- [P110] Díaz Rodríguez, N. (2011), A framework for context-aware applications for smart spaces, International Symposium on Applications and the Internet, 218-221, DOI: 10.1109/SAINT.2011.39.
- [P111] Dimakis, N., Soldatos, J., Polymenakos, L., Bürkle, A., Pfirrmann, U., Sutschet, G. (2010), Agent-based architectural framework enhancing configurability, autonomy and scalability of context-aware pervasive services, *Autonomous Agents and Multi-Agent Systems*, 21, (1), 36-68, DOI: 10.1007/s10458-009-9101-1.
- [P112] Ding, J., Hillston, J., Laurenson, D. (2009), Performance Modelling of Content Adaptation for a Personal Distributed Environment, Wireless Personal Communications, 48, (1), 93-112, DOI: 10.1007/s11277-007-9427-4.
- [P113] Dockhorn Costa, P., Ferreira Pires, L., Van Sinderen, M. (2005), Architectural patterns for context-aware services platforms, *International Workshop on Ubiquitous Computing*, 3-18.
- [P114] Dong, Y., Li, Q., Shi, Y. (2007), Research on the architecture of ontology-based context-aware application in pervasive environment, *International Conference on Pervasive Computing and Applications*, 128-132, DOI: 10.1109/ICPCA.2007.4365425.
- [P115] Doukas, C., Fotiou, N., Polyzos, G.C., Maglogiannis, I. (2014), Cognitive and context-aware assistive environments using future internet technologies, *Universal Access in the Information Society*, 13, (1), 59-72, DOI: 10.1007/s10209-013-0299-y.
- [P116] Efstratiou, C., Cheverst, K., Davies, N., Friday, A. (2001), An Architecture for the Effective Support of Adaptive Context-Aware Applications, *International Conference on Mobile Data Management*, 15-26.
- [P117] Elgazzar, K., Martin, P., Hassanein, H. S. (2016), Personal mobile services, Service Oriented Computing and Applications, 10, (1), Springer Verlag, DOI: 10.1007/s11761-014-0164-8.
- [P118] Elmalaki, S., Wanner, L., Srivastava, M. (2015), CAreDroid: Adaptation framework for android context-aware applications, International Conference on Mobile Computing and Networking, 386-399, DOI: 10.1145/2789168.2790108.
- [P119] England, D., Randles, M.J., Taleb-Bendiab, A. (2009), Runtime User Interface Design and Adaptation, British HCI Group Annual Conference on People and Computers, 463-470, British Computer Society.
- [P120] Escolar, J.R., Cachón, C.G., Marín, I., Gutiérrez, N. (2013), MMW-4S: A model-based approach for generating context-aware user interfaces for the mobile Web, Workshop on Context-Aware Adaptation of Service Front-Ends, 28-31.
- [P121] Espada, J.P., Crespo, R.G., Martínez, O.S., Cristina Pelayo G-Bustelo, B., Lovelle, J.M.C. (2012), Extensible architecture for context-aware mobile web applications, *Expert Systems with Applications*, 39, (10), 9686-9694, DOI: 10.1016/j.eswa.2012.02.151.
- [P122] Eugster, P., Garbinato, B., Holzer, A. (2009), Pervaho: A specialized middleware for mobile context-aware applications, *Electronic Commerce Research*, 9, (4), 245-268, DOI: 10.1007/s10660-009-9042-4.
- [P123] Evchina, Y., Dvoryanchikova, A., Lastra, J.L.M. (2012), Ontological framework of context-aware and reasoning middleware for smart homes with health and social services, *International Conference on Systems, Man and Cybernetics*, 985-990, DOI: 10.1109/ICSMC.2012.6377857.
- [P124] Evers, C., Kniewel, R., Geihs, K., Schmidt, L. (2012), Achieving user participation for adaptive applications, *International Conference on Ubiquitous Computing and Ambient Intelligence*, 200-207, DOI: 10.1007/978-3-642-35377-2_28.
- [P125] Falcarin, P., Valla, M., Yu, J., Licciardi, C.A., Frà, C., Lamorte, L. (2013), Context data management: an architectural framework for context-aware services, *Service Oriented Computing and Applications*, 7, (2), 151-168, DOI: 10.1007/s11761-012-0115-1.

- [P126] Fkaier S., Romdhani M., Khalgui M., Frey G. (2017), Enabling reconfiguration of adaptive control systems using real-time context-aware framework, *Proceedings of IEEE/ACS International Conference on Computer Systems and Applications*, AICCSA, Association for Computing Machinery, Inc, DOI: 10.1109/AICCSA.2016.7945781.
- [P127] Floch, J., Frà, C., Fricke, R., Geihs, K., Wagner, M., Lorenzo, J., Soladana, E., Mehlhase, S., Paspallis, N., Rahnama, H., Ruiz, P.A., Scholz, U. (2013), Playing MUSIC - Building context-aware and self-adaptive mobile applications, *Software - Practice* and Experience, 43, (3), 359-388, DOI: 10.1002/spe.2116.
- [P128] Fonteles, A.S., Neto, B.J.A., Maia, M., Viana, W., Andrade, R.M.C. (2013), An adaptive context acquisition framework to support mobile spatial and context-aware applications, *International Symposium on Web and Wireless Geographical Information Systems*, 100-116, DOI: 10.1007/978-3-642-37087-8_8.
- [P129] Fortier, A., Rossi, G., Gordillo, S.E., Challiol, C. (2010), Dealing with variability in context-aware mobile software, *Journal of Systems and Software*, 83, (6), 915-936, DOI: 10.1016/j.jss.2009.11.002.
- [P130] Fraile, J.A., De Paz, Y., Bajo, J., De Paz, J.F., Pérez-Lancho, B. (2014), Context-aware multiagent system: Planning home care tasks, *Knowledge and Information Systems*, 40, (1), 171-203, DOI: 10.1007/s10115-013-0635-9.
- [P131] Frees, S. (2010), Context-driven interaction in immersive virtual environments, *Virtual Reality*, 14, (4), 277-290, DOI: 10.1007/s10055-010-0178-2.
- [P132] Fröhlich, N., Meier, A., Möller, T., Savini, M., Schuldt, H., Vogt, J. (2009), LoCa towards a context-aware infrastructure for eHealth applications, *International Conference on Distributed Multimedia Systems*, 52-57.
- [P133] Fuentes, L., Gámez, N., Sánchez, P. (2009), Aspect-oriented design and implementation of context-aware pervasive applications, *Innovations in Systems and Software Engineering*, 5, (1), 79-93, DOI: 10.1007/s11334-009-0081-8.
- [P134] Gabillon, Y., Biri, N., Otjacques, B. (2015), Designing an adaptive user interface according to software product line engineering, *International Conference on Advances in Computer-Human Interactions*, 86-91.
- [P135] Gacimartín, C., Hernández, J.A., Larrabeiti, D. (2011), A Middleware Architecture for Designing TV-based Adapted Applications for the Elderly, *Proceedings of the 14th International Conference on Human-computer Interaction: Design and Development Approaches - Volume Part I*, 443-449, Elsevier Ltd, DOI: 10.1007/978-3-642-21602-2_48.
- [P136] García Ramirez, A., González-Carrasco, I., Henrique Jasper, E., Lima Lopez, A., Lopez-Cuadrado, J.L., García-Crespo, A., (2017), Towards Human Smart Cities: Internet of Things for sensory impaired individuals, *Computing*, 99, (1), DOI: 10.1007/s00607-016-0529-2.
- [P137] Gaur, S., Rangarajan, R., Tovar, E. (2016), Extending T-Res with mobility for context-aware IoT, Proceedings 2016 IEEE Ist International Conference on Internet-of-Things Design and Implementation, IoTDI 2016, 2016-January, 293-296, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1109/IoTDI.2015.16.
- [P138] Gençer, M., Bilgin, G. Zan, O., Voyvodaoğlu, T. (2013), A New Framework for Increasing User Engagement in Mobile Applications Using Machine Learning Techniques, Proceedings of the Second International Conference on Design, User Experience, and Usability: Web, Mobile, and Product Design - Volume Part IV, 651-659, ACM, DOI: 10.1007/978-3-642-39253-5_72.
- [P139] Ghiani, G., Manca, M., Paternò, F., Rett, J., Vaibhav, A. (2015), Adaptive multimodal web user interfaces for smart work environments, *Journal of Ambient Intelligence and Smart Environments*, 7, (6), 701-717, DOI: 10.3233/AIS-150344.
- [P140] Goh, E., Chieng, D., Mustapha, A.K., Ngeow, Y.C., Low, H.K. (2007), A context-aware architecture for smart space environment, *International Conference on Multimedia and Ubiquitous Engineering*, 908-913, DOI: 10.1109/MUE.2007.10.
- [P141] Gopalan, K.S., Nathan, S., Bhanu Teja, C.H., Channa, A.B., Saraf, P. (2011), A context aware personalized media recommendation system - An adaptive evolutionary algorithm approach, *International Conference on Bio-Inspired Computing: Theories and Applications*, 45-50, DOI: 10.1109/BIC-TA.2011.4.
- [P142] Gopalan, K.S., Nathan, S., Teja, C.H.B., Channa, A.B., Saraf, P., Shanker, G. (2011), A cloud based service architecture for personalized media recommendations, *International Conference on Next Generation Mobile Applications, Services and Technologies*, 19-24, DOI: 10.1109/NGMAST.2011.14.
- [P143] Grigera, J., Fortier, A., Rossi, G., Gordillo, S. (2007), A modular architecture for context sensing, International Conference on Advanced Information Networking and Applications Workshops, 147-152, DOI: 10.1109/AINAW.2007.30.
- [P144] Griol, D., Carbo, J., Molina, J.M. (2013), Bringing context-aware access to the web through spoken interaction, Applied Intelligence, 38, (4), 620-640, DOI: 10.1007/s10489-012-0390-8.
- [P145] Griol, D., Molina, J.M., Callejas, Z. (2014), Modeling the user state for context-aware spoken interaction in ambient assisted living, *Applied Intelligence*, 40, (4), 749-771, DOI: 10.1007/s10489-013-0503-z.
- [P146] Griswold, W.G., Boyer, R., Brown, S.W., Truong, T.M. (2003), A Component Architecture for an Extensible, Highly Integrated Context-aware Computing Infrastructure, *International Conference on Software Engineering*, 363-372.
- [P147] Grundy, J., Hosking, J. (2000), Developing Adaptable User Interfaces for Component-Based Systems, Australasian User Interface Conference, 17-25, DOI: 10.1016/S0953-5438(01)00049-2.
- [P148] Gu, T., Pung, H. K., Zhang, D. Q. (2004), Toward an OSGi-based infrastructure for context-aware applications, *IEEE Pervasive Computing*, 3, (4), 66-74, DOI: 10.1109/MPRV.2004.19.
- [P149] Guermah, H., Fissaa, T., Hafiddi, H., Nassar, M., Kriouile, A. (2014), A semantic approach for service adaptation in contextaware environment, *Procedia Computer Science*, 34, 587-592, DOI: 10.1016/j.procs.2014.07.077.
- [P150] Gui, F., Guillen, M., Rishe, N., Barreto, A., Andrian, J., Adjouadi, M. (2009), A client-server architecture for context-aware search application, *International Conference on Network-Based Information Systems*, 539-546, DOI: 10.1109/NBiS.2009.75.

- [P151] Gui, N., De Florio, V., Sun, H., Blondia, C. (2009), ACCADA: A framework for continuous context-aware deployment and adaptation, *Symposium on Self-Stabilizing Systems*, 325-340, DOI: 10.1007/978-3-642-05118-0_23.
- [P152] Gullà F., Ceccacci S., Menghi R., Germani M. (2016), An adaptive smart system to foster disabled and elderly people in kitchen-related task, ACM International Conference Proceeding Series, 29-June-2016, ACM, DOI: 10.1145/2910674.2910678.
- [P153] Gullà, F., Cavalieri, L., Ceccacci, S., Germani, M., Bevilacqua, R. (2015), Method to design adaptable and adaptive user interfaces, *International Conference on Human-Computer Interaction*, 19-24, DOI: 10.1007/978-3-319-21380-4_4.
- [P154] Guo A., Ma J. (2017), Context-Aware Scheduling in Personal Data Collection from Multiple Wearable Devices, *IEEE Access*, 5, 2602-2614, Springer Verlag, DOI: 10.1109/ACCESS.2017.2666419.
- [P155] Guo K., Huang Y., Kuang L., Zhang Y. (2017), CASP: A Context-Aware Transparent Active Service Provision Architecture in a Mobile Internet Environment, *Computing in Science and Engineering*, 19, (1), 38-45, Association for Computing Machinery, DOI: 10.1109/MCSE.2017.16.
- [P156] Guo, B., Zhang, D., Imai, M. (2011), Toward a cooperative programming framework for context-aware applications, *Personal and Ubiquitous Computing*, 15, (3), 221-233-, DOI: 10.1007/s00779-010-0329-1.
- [P157] Guo, B., Zhang, D., Sun, L., Yu, Z., Zhou, X. (2013), iCROSS: toward a scalable infrastructure for cross-domain context management, *Personal and Ubiquitous Computing*, 17, (3), 591-602, DOI: 10.1007/s00779-012-0564-8.
- [P158] Guo, B., Zhang, D., Yu, Z., Liang, Y., Wang, Z., Zhou, X. (2013), From the internet of things to embedded intelligence, World Wide Web, 16, (4), 399-420, DOI: 10.1007/s11280-012-0188-y.
- [P159] Guo, M., Zhao, Y. (2008), An architecture for digital library information service in an ambient intelligence environment, International Conference on Computer Science and Software Engineering, 444-447, DOI: 10.1109/CSSE.2008.1637.
- [P160] Gupta, S., Sood, S. (2015), Context aware mobile agent for reducing stress and obesity by motivating physical activity: A design approach, *International Conference on Computing for Sustainable Global Development*, 962-966, DOI: NO DOI.
- [P161] Haghighi, P.D., Indrawan-Santiago, M., Perera, A., Huynh, T.M. (2014), Situation-aware mobile health monitoring, International Conference on Mobile and Ubiquitous Systems: Computing, Networking and Services, 248-256, DOI: 10.4108/icst.mobiquitous.2014.257974.
- [P162] Hallsteinsen, S., Geihs, K., Paspallis, N., Eliassen, F., Horn, G., Lorenzo, J., Mamelli, A., Papadopoulos, G.A. (2012), A Development Framework and Methodology for Self-adapting Applications in Ubiquitous Computing Environments, *Journal* of Systems and Software, 85, (12), 2840-2859, Elsevier Science Inc., DOI: 10.1016/j.jss.2012.07.052.
- [P163] Hamann, T., Hübsch, G., Springer, T. (2008), A model-driven approach for developing adaptive software systems, International Conference on Distributed Applications and Interoperable Systems, 196-209, DOI: 10.1007/978-3-540-68642-2_16.
- [P164] Han, M., Kim, G., Park, S., Kim, L., Ha, S. (2009), Process and location-aware information service system for the disabled and the elderly, *International Conference on Universal Access in Human-Computer Interaction*, 207-215, DOI: 10.1007/978-3-642-02713-0_22.
- [P165] Harchay A., Cheniti-Belcadhi L., Braham R. (2014), A context-aware framework to provide personalized mobile assessment, Interaction Design and Architecture(s), 23, (1), 82-97, Springer Verlag.
- [P166] Harumoto, K., Shimojo, S. (2003), A P2P Platform Architecture for Context-Sensitive Applications and Its Implementation Using Web Services, *Pacific Rim Conference on Communications, Computers and signal Processing*, 185-188, DOI: 10.1109/PACRIM.2003.1235748.
- [P167] Hashemi, M., Herbert, J. (2016), A Service-Oriented User Interaction Analysis Framework Supporting Adaptive Applications, Proceedings - International Computer Software and Applications Conference, 2, 546-551, AI Access Foundation, DOI: 10.1109/COMPSAC.2016.36.
- [P168] Hassani, A., Haghighi, P.D., Burstein, F., Davey, S. (2017), Context aware data synchronisation during emergencies, International and Interdisciplinary Conference on Modeling and Using Context (Context 2017), 10257 LNAI, 406-417, IEEE Computer Society, DOI: 10.1007/978-3-319-57837-8_34.
- [P169] Hatala, M., Wakkary, R. (2005), Ontology-Based User Modeling in an Augmented Audio Reality System for Museums, User Modeling and User-Adapted Interaction, 15, (3), 339-380, DOI: 10.1007/s11257-005-2304-5.
- [P170] He, Q. S., Tu, S. L. (2006), A lightweight architecture to support context-aware ubiquitous agent system, Pacific Rim International Workshop on Multi-Agents, 696-701.
- [P171] He, Q.S., Chen, X., Tu, S.L. (2006), Research of a Goal-Driven Architecture in Ubiquitous Environments, International Conference on Computer and Information Technology, 237-, IEEE Computer Society, DOI: 10.1109/CIT.2006.162.
- [P172] He, W., Ren, G., Cui, L., Li, H. (2015), User behavioral context-aware service recommendation for personalized Mashups in pervasive environments, Asia-Pacific Web Conference, 683-694, DOI: 10.1007/978-3-319-25255-1_56.
- [P173] Herbert, J., O'donoghue, J., Chen, X. (2009), A context-sensitive rule-based architecture for a smart building environment, International Journal of Smart Home, 3, (1), 1-6, DOI: 10.1109/FGCN.2008.169.
- [P174] Herdin, C., Märtin, C., Forbrig, P. (2017), SitAdapt: An architecture for situation-aware runtime adaptation of interactive systems, *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, 10271, 447-455, Springer Verlag, DOI: 10.1007/978-3-319-58071-5_33.
- [P175] Hofer, T., Schwinger, W., Pichler, M., Leonhartsberger, G., Altmann, J., Retschitzegger, W. (2003), Context-Awareness on Mobile Devices - The Hydrogen Approach, *Hawaii International Conference on System Sciences*, DOI: 10.1109/HICSS.2003.1174831.

- [P176] Hoh, S., Devaraju, A., Wong, C.C. (2008), A context aware framework for user centered services, International Symposium Human Factors in Telecommunication.
- [P177] Hong, J., Suh, E.-H., Kim, J., Kim, S. (2009), Context-aware system for proactive personalized service based on context history, *Expert Systems with Applications*, 36, (4), 7448-7457, DOI: 10.1016/j.eswa.2008.09.002.
- [P178] Hong, J.I., Landay, J.A. (2001), An infrastructure approach to context-aware computing, *Human-Computer Interaction*, 16, (2), 287-303, DOI: 10.1207/S15327051HCI16234_11.
- [P179] Hong, M., Jung, J. J., Piccialli, F., Chianese, A. (2017), Social recommendation service for cultural heritage, *Personal and Ubiquitous Computing*, 21, (2), DOI: 10.1007/s00779-016-0985-x.
- [P180] Hooper, C.J., Preston, A., Balaam, M., Seedhouse, P., Jackson, D., Pham, C., Ladha, C., Ladha, K., Plötz, T., Olivier, P. (2012), The french kitchen: Task-based learning in an instrumented kitchen, *Conference on Ubiquitous Computing*, 193-202, DOI: 10.1145/2370216.2370246.
- [P181] Hossain, M. A. (2011), A Framework for Adaptive Interaction Support Based on Quality of Context Information, *International Conference on Multimedia and Expo*, 1-6, IEEE Computer Society, DOI: 10.1109/ICME.2011.6012211.
- [P182] Hsu, I.-C. (2011), An architecture of mobile Web 2.0 context-aware applications in ubiquitous Web, *Journal of Software*, 6, (4), 705-715, DOI: 10.4304/jsw.6.4.705-715.
- [P183] Hudlicka, E., McNeese, M. D. (2002), Assessment of User Affective and Belief States for Interface Adaptation: Application to an Air Force Pilot Task, User Modeling and User-Adapted Interaction, 12, (1), 1-47, DOI: 10.1023/A:1013337427135.
- [P184] Huebscher, M.C., McCann, J.A. (2006), An adaptive middleware framework for context-aware applications, *Personal and Ubiquitous Computing*, 10, (1), 12-20, DOI: 10.1007/s00779-005-0035-6.
- [P185] Huigui, S., Xiufen, F., Zhiqing, L., Qunsheng, Y., Shaohua, T. (2006), A service-oriented middleware for pervasive computing environments, *International Symposium on Pervasive Computing and Applications*, 36-41, DOI: 10.1109/SPCA.2006.297471.
- [P186] Hussein, M., Han, J., Colman, A., Yu, J. (2012), An architecture-based approach to developing context-aware adaptive systems, International Conference and Workshops on Engineering of Computer-Based Systems, 154-163, DOI: 10.1109/ECBS.2012.13.
- [P187] Hussein, T., Linder, T., Gaulke, W., Ziegler, J. (2014), Hybreed: A software framework for developing context-aware hybrid recommender systems, User Modeling and User-Adapted Interaction, 24, (1), 121-174, DOI: 10.1007/s11257-012-9134-z.
- [P188] Iglesias, J., Bernardos, A.M., Álvarez, A., Sacristán, M. (2010), A light reasoning infrastructure to enable context-aware mobile applications, *International Conference on Embedded and Ubiquitous Computing*, 386-391, DOI: 10.1109/EUC.2010.64.
- [P189] Iglesias, J., Bernardos, A.M., Tarryio, P., Casar, J.R., Martyin, H. (2012), Design and validation of a light inference system to support embedded context reasoning, *Personal and Ubiquitous Computing*, 16, (7), 781-797, DOI: 10.1007/s00779-011-0447-4.
- [P190] Ikram, A., Baker, N., Knappmeyer, M., Reetz, E.S., Tönjesy, R. (2011), An artificial chemistry based framework for personal and social context aware smart spaces, *International Wireless Communications and Mobile Computing Conference*, 2009-2014, DOI: 10.1109/IWCMC.2011.5982843.
- [P191] Immanuel V.A., Raj P. (2016), Enabling context-awareness: A service oriented architecture implementation for a hospital use case, Proceedings of the 2015 International Conference on Applied and Theoretical Computing and Communication Technology, iCATccT 2015, 224-228, Springer New York LLC, DOI: 10.1109/ICATCCT.2015.7456886.
- [P192] Itao, T., Tanaka, S., Suda, T., Aoyama, T. (2004), A Framework for Adaptive UbiComp Applications Based on the Jack-inthe-Net Architecture, *Wireless Networks*, 10, (3), 287-299, DOI: 10.1023/B:WINE.0000023862.14256.df.
- [P193] Jang, H., Song, C.H., Kim, M.J., Eom, Y.I. (2011), Agent-based intelligent middleware for user-centric services in ubiquitous computing environments, *Journal of Information Science and Engineering*, 27, (5), 1729-1746.
- [P194] Jang, S. I., Kim, J. H., Ramakrishna, R.S. (2001), Framework for Building Mobile Context-Aware Applications, International Conference Human Society@Internet, 139-150, DOI: 10.1007/3-540-47749-7_11.
- [P195] Jaouadi, I., Ben Djemaa, R., Ben-Abdallah, H. (2016), A model-driven development approach for context-aware systems, Software & Systems Modeling, Springer Verlag, DOI: 10.1007/s10270-016-0550-0.
- [P196] Johansson, D., Åhlund, A., Åhlund, C. (2011), A MIP-P2P Based Architecture for Application Mobility, Proceedings of the 10th International Conference on Mobile and Ubiquitous Multimedia, 85-93, ACM, DOI: 10.1145/2107596.2107606.
- [P197] Joo, S.C., Jeong, C.W., Kim, K.H. (2013), A study of Context-Based Adaptive Service Model in home environments, *Telecommunication Systems*, 52, (4), 2375-2386, DOI: 10.1007/s11235-011-9550-0.
- [P198] Jorge, E., Farias, M., Carmo, R., Vieira, W. (2012), A framework for context-aware systems in mobile devices, *International Conference on Computational Science and Its Applications*, 444-456, DOI: 10.1007/978-3-642-31128-4_33.
- [P199] Jung, E.-C., Sato, K. (2005), A framework of context-sensitive visualization for user-centered interactive systems, *International Conference on User Modeling*, 423-427, DOI: 10.1007/11527886_57.
- [P200] Jung, H.-S., Baek, J.-K., Jeong, C.-S., Lee, Y.-W., Hong, P.-D. (2007), Unified ubiquitous middleware for U-city, International Conference on Convergence Information Technology, 2374-2379, DOI: 10.1109/ICCIT.2007.4420607.
- [P201] Kamienski, C., Jentsch, M., Eisenhauer, M., Kiljander, J., Ferrera, E., Rosengren, P., Thestrup, J., Souto, E., Andrade, W.S., Sadok, D. (2017), Application development for the Internet of Things: A context-aware mixed criticality systems development platform, *Computer Communications*, 104, 1-16, Springer Verlag, DOI: 10.1016/j.comcom.2016.09.014.
- [P202] Kang, D.O., Lee, H.J., Ko, E.J., Kang, K., Lee, J. (2006), A wearable context aware system for ubiquitous healthcare, International Conference of the IEEE Engineering in Medicine and Biology Society, 5192-5195.

- [P203] Kang, S.K., Chung, K.Y., Ryu, J.K., Rim, K.W., Lee, J.H. (2013), Bio-Interactive Healthcare Service System Using Lifelog Based Context Computing, *Wireless Personal Communications*, 73, (2), 341-351, DOI: 10.1007/s11277-013-1242-5.
- [P204] Kapitsaki, G.M., Kateros, D.A., Prezerakos, G.N., Venieris, I.S. (2009), Model-driven development of composite contextaware web applications, *Information and Software Technology*, 51, (8), 1244-1260, DOI: 10.1016/j.infsof.2009.03.002.
- [P205] Karchoud, R., Roose, P., Marc D, M., Illarramendi, A., Ilarri, S., (2017), Long-life application: Approach for User Context Management and Situation Understanding, *Personal and Ubiquitous Computing*, 21, (6), DOI: 10.1007/s00779-017-1077-2.
- [P206] Katz, B.F.G., Kammoun, S., Parseihian, G., Gutierrez, O., Brilhault, A., Auvray, M., Truillet, P., Denis, M., Thorpe, S., Jouffrais, C. (2012), NAVIG: augmented reality guidance system for the visually impaired, *Virtual Reality*, 16, (4), 253-269, DOI: 10.1007/s10055-012-0213-6.
- [P207] Kaufmann, O., Lorenz, A., Oppermann, R., Schneider, A., Eisenhauer, M., Zimmermann, A. (2007), Implicit Interaction for Pro-active Assistance in a Context-adaptive Warehouse Application, *International Conference on Mobile Technology*, *Applications, and Systems and International Symposium on Computer Human Interaction in Mobile Technology*, 729-735, ACM, DOI: 10.1145/1378063.1378187.
- [P208] Kawamura, K., Nilas, P., Muguruma, K., Adams, J.A., Zhou, C. (2003), An Agent-Based Architecture for an Adaptive Human-Robot Interface, *Hawaii International Conference on System Sciences*, DOI: 10.1109/HICSS.2003.1174288.
- [P209] Kawanishi, N., Jin, K.S., Si, H., Kawahara, Y., Morikawa, H. (2006), Building context-aware applications and probe space infrastructure, *International Symposium on Intelligent Signal Processing and Communications*, 103-106, DOI: 10.1109/ISPACS.2006.364845.
- [P210] Kawsar, F., Nakajima, T., Park, J.H., Yeo, S.S. (2010), Design and implementation of a framework for building distributed smart object systems, *Journal of Supercomputing*, 54, (1), 4-28, DOI: 10.1007/s11227-009-0323-4.
- [P211] Keeney, J., Cahill, V. (2003), Chisel: A Policy-Driven, Context-Aware, Dynamic Adaptation Framework, International Workshop on Policies for Distributed Systems and Networks, 3-14, DOI: 10.1109/POLICY.2003.1206953.
- [P212] Khattak, A.M., Vinh, L.T., Hung, D.V., Truc, P.T.H., Hung, L.X., Guan, D., Pervez, Z., Han, M., Lee, S., Lee, Y.-K. (2010), Context-aware human activity recognition and decision making, *International Conference on e-Health Networking Applications and Services*, 112-118, DOI: 10.1109/HEALTH.2010.5556585.
- [P213] Khedr, M., Karmouch, A. (2005), ACAI: Agent-based context-aware infrastructure for spontaneous applications, Journal of Network and Computer Applications, 28, (1), 19-44, DOI: 10.1016/j.jnca.2004.04.002.
- [P214] Khoi, N.M., Rodriguez-Pupo, L.E., Casteleyn, S. (2017), Citizense-A generic user-oriented participatory sensing framework, 2017 International Conference on Selected Topics in Mobile and Wireless Networking, MoWNeT 2017, IEEE Conferences, DOI: 10.1109/MoWNet.2017.8045954.
- [P215] Kim, G., Shin, D., Shin, D. (2004), Design of a Middleware and HIML (Human Interaction Markup Language) for Context Aware Services in a Ubiquitous Computing Environment, *International Conference on Embedded and Ubiquitous Computing*, 682-691.
- [P216] Kim, H.W., Hoque, M.R., Seo, H., Yang, S.H. (2016), Development of middleware architecture to realize context-aware service in smart home environment, *Computer Science and Information Systems*, 13, (2), 427-452, AI Access Foundation, DOI: 10.2298/CSIS150701010H.
- [P217] Kim, J., Chung, K.Y. (2014), Ontology-based healthcare context information model to implement ubiquitous environment, *Multimedia Tools and Applications*, 71, (2), 873-888, DOI: 10.1007/s11042-011-0919-6.
- [P218] Kim, J., Kim, J., Lee, D., Chung, K.Y. (2014), Ontology driven interactive healthcare with wearable sensors, *Multimedia Tools and Applications*, 71, (2), 827-841, DOI: 10.1007/s11042-012-1195-9.
- [P219] Kim, J., Park, S.O. (2015), U-Health Smart system architecture and ontology model, *The Journal of Supercomputing*, 71, (6), 2121-2137, DOI: 10.1007/s11227-014-1334-3.
- [P220] Kolamunna, H., Hu, Y., Perino, D., Thilakarathna, K., Makaroff, D., Guan, X., Seneviratne, A. (2016), AFV: Enabling application function virtualization and scheduling in wearable networks, *UbiComp 2016 - Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing*, 981-991, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1145/2971648.2971727.
- [P221] Korpipää, P., Malm, E.J., Salminen, I., Rantakokko, T., Kyllönen, V., Känsälä, I. (2005), Context Management for End User Development of Context-aware Applications, *Proceedings of the 6th International Conference on Mobile Data Management*, 304-308, Association for Computing Machinery, DOI: 10.1145/1071246.1071296.
- [P222] Korpipää, P., Mantyjarvi, J., Kela, J., Keranen, H., Malm, E.-J. (2003), Managing Context Information in Mobile Devices, IEEE Pervasive Computing, 2, (3), 42-51, DOI: 10.1109/MPRV.2003.1228526.
- [P223] Korzun, D., Varfolomeyev, Yalovitsyna, S., Volokhova, V. (2017), Semantic infrastructure of a smart museum: toward making cultural heritage knowledge usable and creatable by visitors and professionals, *Personal and Ubiquitous Computing*, 21, (2), DOI: 10.1007/s00779-016-0996-7.
- [P224] Kostelník, P., Sarnovský, M., Furdík, K. (2011), The semantic middleware for networked embedded systems applied in the internet of things and services domain, *Scalable Computing: Practice and Experience*, 12, (3), 307-315.
- [P225] Koufi, V., Malamateniou, F., Vassilacopoulos, G. (2010), A system for the provision of medical diagnostic and treatment advice in home care environment, *Personal and Ubiquitous Computing*, 14, (6), 551-561, DOI: 10.1007/s00779-009-0275-y.

- [P226] Kropf, J., Kadyrov, S., Roedl, L. (2017), Modular IoT platform for AAL and home care using bluetooth low energy, International Conference on Wireless Mobile Communication and Healthcare (MobiHealth 2016), 192, 132-137, DOI: 10.1007/978-3-319-58877-3_17.
- [P227] Kubo, Y., Shizuki, B., Takada, R., Takahashi, S. (2017), SynCro: Context-aware user interface system for smartphonesmartwatch cross-device interaction, *Conference on Human Factors in Computing Systems - Proceedings*, Part F127655, 1794-1801, Inderscience Publishers, DOI: 10.1145/3027063.3053088.
- [P228] Kuo, Z., Yanni, W., Zhenkun, Z., Xiaoge, W. and Yu, C. (2005), A component-based reflective middleware approach to context-aware adaptive systems, *International Conference on Web Engineering*, 429-434.
- [P229] Kurmis, M., Andziulis, A., Dzemydiene, D., Jakovlev, S., Voznak, M., Gricius, G. (2015), Cooperative Context Data Acquisition and Dissemination for Situation Identification in Vehicular Communication Networks, Wireless Personal Communications, 85, (1), 49-62, DOI: 10.1007/s11277-015-2727-1.
- [P230] Kurschl, W., Mitsch, S., Schoenboeck, J. (2009), Modeling situation-aware ambient assisted living systems for eldercare, International Conference on Information Technology: New Generations, 1214-1219, DOI: 10.1109/ITNG.2009.25.
- [P231] Kyriazakos, S., Mihaylov, M., Anggorojati, B., Mihovska, A., Craciunescu, R., Fratu, O., Prasad, R. (2016), eWALL: An Intelligent Caring Home Environment Offering Personalized Context-Aware Applications Based on Advanced Sensing, *Wireless Personal Communications*, 87, (3), 1093-1111, Association for Computing Machinery, Inc, DOI: 10.1007/s11277-015-2779-2.
- [P232] Lachhab, F., Essaaidi M., Bakhouya M., Ouladsine R. (2016), Towards a context-aware platform for complex and stream event processing, 2016 International Conference on High Performance Computing and Simulation, HPCS 2016, 961-966, Springer Verlag, DOI: 10.1109/HPCSim.2016.7568438.
- [P233] Laga, N., Bertin, E., Crespi, N., Bedini, I., Molina, B., Zhao, Z. (2013), A flexible service selection for executing virtual services, World Wide Web, 16, (3), 219-245, DOI: 10.1007/s11280-012-0184-2.
- [P234] Laitakari, J., Pakkala, D. (2008), Dynamic context monitoring service for adaptive and context-aware applications, International Workshop on Applications and Services in Wireless Networks, 11-19, DOI: 10.1109/ASWN.2008.6.
- [P235] Lalanda, P., Escoffier C. (2017), Resource-oriented framework for representing pervasive context, Proceedings 2017 IEEE 2nd International Congress on Internet of Things, ICIOT 2017, 155-158, DOI: 10.1109/IEEE.ICIOT.2017.27.
- [P236] Lavirotte, S., Rey, G., Rocher, G., Tigli, J.-Y., (2015), A generic service oriented software platform to design ambient intelligent systems, *International Joint Conference on Pervasive and Ubiquitous Computing and International Symposium on Wearable Computers*, 281-284, DOI: 10.1145/2800835.2800843.
- [P237] Lee, D.G., Park, J.H., Kim, T.H., Yang, L.T. (2008), U-multimedia framework: a secure and intelligent multimedia service framework based on context information in U-home, *The Journal of Supercomputing*, 45, (1), 88-104, DOI: 10.1007/s11227-008-0202-4.
- [P238] Lee, J.Y., Rhee, G. (2008), Context-aware 3D visualization and collaboration services for ubiquitous cars using augmented reality, *International Journal of Advanced Manufacturing Technology*, 37, (5), 431-442, DOI: 10.1007/s00170-007-0996-x.
- [P239] Lee, J.Y., Seo, D.W. (2005), A context-aware and augmented reality-supported service framework in ubiquitous environments, International Conference on Embedded and Ubiquitous Computing, 258-267.
- [P240] Lee, J.Y., Seo, D.W., Kim, K., Kim, H. (2005), A Ubiquitous and Context-Aware Framework for Supporting Virtual Engineering Services, *Computer-Aided Design and Applications*, 2, (6), 769-776.
- [P241] Lee, Y., Min, C., Ju, Y., Pushp, S., Song, J. (2011), A mobile context monitoring platform for pervasive computing environments, *International Conference on Digital Ecosystems and Technologies*, 345-348, DOI: 10.1109/DEST.2011.5936591.
- [P242] Lehmann, G., Rieger, A., Blumendorf, M., Albayrak, S. (2010), A 3-layer architecture for smart environment models: A modelbased approach, *International Conference on Pervasive Computing and Communications Workshops*, 636-641, DOI: 10.1109/PERCOMW.2010.5470513.
- [P243] Lewis, G., Novakouski, M., Sánchez, E. (2012), A reference architecture for group-context-aware mobile applications, International Conference on Mobile Computing, Applications, and Services, 44-63, DOI: 10.1007/978-3-642-36632-1_3.
- [P244] Lézoray, J.B., Segarra, M.T., Phung-Khac, A., Thépaut, A., Gilliot, J.M., Beugnard, A. (2011), A design process enabling adaptation in pervasive heterogeneous contexts, *Personal and Ubiquitous Computing*, 15, (4), 353-363, DOI: 10.1007/s00779-010-0356-y.
- [P245] Li G., Su L., Chen X. (2017), Pragmatic Web Oriented Context-Aware System Design, Proceedings 9th International Conference on Intelligent Human-Machine Systems and Cybernetics, IHMSC 2017, 1, 288-293, DOI: 10.1109/IHMSC.2017.73.
- [P246] Li, H., Wang, J. (2011), Application architecture for ambient intelligence systems based on context ontology modeling, International Conference on Internet Technology and Applications, 1-4, DOI: 10.1109/ITAP.2011.6006387.
- [P247] Li, X., Tao, X., Lu, J. (2017), Towards a programming framework for activity-oriented context-aware applications, *Frontiers of Computer Science*, 11, (6), DOI: 10.1007/s11704-016-5399-4.
- [P248] Lin, C.-H., Ho, P.-H., Lin, H.-C. (2014), Framework for NFC-based intelligent agents: A context-awareness enabler for social internet of things, *International Journal of Distributed Sensor Networks*, 10, (2), DOI: 10.1155/2014/978951.
- [P249] Liu, S., Cheng, L. (2011), A context-aware reflective middleware framework for distributed real-time and embedded systems, *Journal of Systems and Software*, 84, (2), 205-218, DOI: 10.1016/j.jss.2010.09.049.

- [P250] Liu, X. (2007), A multi-agent-based architecture for enterprise customer and supplier cooperation context-aware information systems, *International Conference on Autonomic and Autonomous Systems*, DOI: 10.1109/CONIELECOMP.2007.46.
- [P251] Lopes, J., Gusmão, M., Souza, R., Davet, P., Souza, A., Costa, C., Barbosa, J., Pernas, A., Yamin, A., Geyer, C. (2013), Towards a distributed architecture for context-aware mobile applications in UbiComp, *Brazilian Symposium on Multimedia and the Web*, 43-49, DOI: 10.1145/2526188.2526209.
- [P252] López-Jaquero, V., Montero, F., González, P. (2009), AB-HCI: An interface multi-agent system to support human-centred computing, *IET Software*, 3, (1), 14-25, DOI: 10.1049/iet-sen:20070108.
- [P253] Luo, J., Dong, F., Cao, J., Song, A. (2010), A context-aware personalized resource recommendation for pervasive learning, *Cluster Computing*, 13, (2), 213-239, DOI: 10.1007/s10586-009-0113-z.
- [P254] Maamar, Z., Benslimane, D., Mrissa, M., Ghedira, C. (2006), CooPS Towards a method for coordinating personalized services, *Software & Systems Modeling*, 5, (2), Association for Computing Machinery, Inc, DOI: 10.1007/s10270-006-0006z.
- [P255] Macik, M., Cerny, T., Basek, J., Slavik, P. (2013), Platform-aware rich-form generation for adaptive systems through codeinspection, *International Conference on Human Factors in Computing and Informatics*, 768-784, DOI: 10.1007/978-3-642-39062-3_55.
- [P256] Madni, A.M., Rahnama, H., Sadeghian, A. (2010), Self-adaptive middleware for the design of context-aware software applications in public transit systems, *International Symposium on Communications, Control and Signal Processing*, DOI: 10.1145/2030112.2030186.
- [P257] Magableh, B., Barrett, S. (2011), Adaptive context oriented component-based application middleware (COCA-middleware), International Conference on Ubiquitous Intelligence and Computing, 137-151, DOI: 10.1007/978-3-642-23641-9_13.
- [P258] Malandrino, D., Mazzoni, F., Riboni, D., Bettini, C., Colajanni, M., Scarano, V. (2010), MIMOSA: context-aware adaptation for ubiquitous web access, *Personal and Ubiquitous Computing*, 14, (4), 301-320, DOI: 10.1007/s00779-009-0232-9.
- [P259] Malatras, A., Hirsbrunner, B. (2009), A context-aware framework to enable adaptation in pervasive computing environments, International Conference on Network-Based Information Systems, 182-187, DOI: 10.1109/NBiS.2009.11.
- [P260] Mallik, A., Tripathi A., Kumar R., Chaudhury S., Sinha K. (2015), Ontology based context aware situation tracking, *IEEE World Forum on Internet of Things*, WF-IoT 2015 Proceedings, 687-692, Springer-Verlag London Ltd, DOI: 10.1109/WF-IoT.2015.7389137.
- [P261] Mamo, B., Ejigu, D. (2014), A generic layered architecture for context aware applications, *Procedia Computer Science*, 34, 619-624, DOI: 10.1016/j.procs.2014.07.085.
- [P262] Mantoro, T., Johnson, C.W., Ayu, M.A. (2009), A framework in ubiquitous computing environment for providing intelligent responses, *International Conference on Mobile Ubiquitous Computing, Systems, Services, and Technologies*, 289-294, DOI: 10.1109/UBICOMM.2009.59.
- [P263] Marsa-Maestre, I., Lopez-Carmona, M.A., Velasco, J.R. (2008), A hierarchical, agent-based service oriented architecture for smart environments, *Service Oriented Computing and Applications*, 2, (4), 167-185, DOI: 10.1007/s11761-008-0030-7.
- [P264] Martín, D., Lamsfus, C., Alzua, A. (2011), Mobile context data management framework, International Conference on Multimedia and Ubiquitous Engineering, 73-78, DOI: 10.1109/MUE.2011.24.
- [P265] Martín, D., Lamsfus, C., Alzua-Sorzabal, A. (2016), A cloud-based platform to develop context-aware mobile applications by domain experts, *Computer Standards and Interfaces*, 44, 177-184, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1016/j.csi.2015.08.009.
- [P266] Marucci, L., Paternò, F. (2002), Design and Evaluation of an Adaptive Virtual Guide for Web Applications, Universal Access in the Information Society, 1, (3), 163-176, DOI: 10.1007/s102090100016.
- [P267] McHeick, H., Sbeity, H., Hazimeh, H., Naim, J., Alameh, M. (2014), Context aware mobile application architecture (CAMAA) for health care systems: Standardization and abstraction of context aware layers, 2014 IEEE Canada International Humanitarian Technology Conference, IHTC 2014, IGI Global, DOI: 10.1109/IHTC.2014.7147523.
- [P268] McNaull, J., Augusto, J., Mulvenna, M., McCullagh, P. (2014), Flexible context aware interface for ambient assisted living, *Human-centric Computing and Information Sciences*, 4, (1), 1-41, DOI: 10.1186/2192-1962-4-1.
- [P269] Meere, D., Ganchev, I., O'Dróma, M., Stojanov, S., Valkanova, V. (2010), An enhanced context-sensitive InfoStation-based mLearning architecture, *International Conference on Digital Ecosystems and Technologies*, 347-352, DOI: 10.1109/DEST.2010.5610623.
- [P270] Mens, K., Cardozo, N., Duhoux, B. (2016), A context-oriented software architecture, Proceedings of the 8th International Workshop on Context-Oriented Programming, COP 2016, 7-12, Elsevier, DOI: 10.1145/2951965.2951971.
- [P271] Merkle N., Kämpgen B., Zander S. (2016), Self-service ambient intelligence using web of things technologies, CEUR Workshop Proceedings, 1588, 1-10, Institute of Electrical and Electronics Engineers Inc..
- [P272] Mets, K., Nelis, J., Verslype, D., Leroux, P., Haerick, W., De Turck, F., Develder, C. (2009), Design of a context aware multimedia management system for home environments, *Computation World: Future Computing, Service Computation, Cognitive, Adaptive, Content, Patterns*, 49-54, DOI: 10.1109/ComputationWorld.2009.111.
- [P273] Mihaylov, M., Mihovska, A., Kyriazakos, S., Prasad, R. (2015), Interoperable eHealth platform for personalized smart services, International Conference on Communication Workshop, 240-245, DOI: 10.1109/ICCW.2015.7247185.

- [P274] Mitrović, N., Bobed, C., Mena, E. (2017), Dynamic user interface architecture for mobile applications based on mobile agents, Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 10034 LNCS, 282-292, Springer Verlag, DOI: 10.1007/978-3-319-55961-2_29.
- [P275] Mizouni, R., Matar, M.A., Mahmoud, Z.A., Alzahmi, S., Salah, A. (2014), A framework for context-aware self-adaptive mobile applications SPL, *Expert Systems with Applications*, 41, (16), 7549-7564, DOI: 10.1016/j.eswa.2014.05.049.
- [P276] Mohomed, I., Misra, A., Ebling, M., Jerome, W. (2008), Context-aware and personalized event filtering for low-overhead continuous remote health monitoring, *International Symposium on a World of Wireless, Mobile and Multimedia Networks*, DOI: 10.1109/WOWMOM.2008.4594820.
- [P277] Mongiello, M., di Noia, T., Nocera, F., di Sciascio, E., Parchitelli, A. (2016), Context-aware design of reflective middleware in the internet of everything, *Federation of International Conferences on Software Technologies: Applications and Foundations* (STAF 2016), 9946 LNCS, 423-435, Association for Computing Machinery, DOI: 10.1007/978-3-319-50230-4_33.
- [P278] Moon, A., Kim, M., Kim, H., Lee, K.-W., Kim, H. (2007), Development of CAMUS based context-awareness for pervasive home environments, *International Journal of Smart Home*, 1, (1), 49-56.
- [P279] Moreira, R. S., Morla, R. S., Moreira, L. P. C., Soares, C. (2016), A behavioral reflective architecture for managing the integration of personal ubicomp systems: automatic SNMP-based discovery and management of behavior context in smartspaces, *Personal and Ubiquitous Computing*, 20, (2), Institute of Electrical and Electronics Engineers Inc., DOI: 10.1007/s00779-016-0901-4.
- [P280] Morrison, J. (2016), Context integrity measurement architecture: A privacy-preserving strategy for the era of ubiquitous computing, 2016 IEEE 7th Annual Ubiquitous Computing, Electronics and Mobile Communication Conference, UEMCON 2016, IEEE Computer Society, DOI: 10.1109/UEMCON.2016.7777816.
- [P281] Motti, V.G., Vanderdonckt, J. (2013), A computational framework for context-aware adaptation of user interfaces, International Conference on Research Challenges in Information Science, DOI: 10.1109/RCIS.2013.6577709.
- [P282] Moussa, F., Ismail, I., Jarraya, M. (2015), Towards a runtime evolutionary model of user-adapted interaction in a ubiquitous environment: the RADEM formal model, *Cognition, Technology and Work*, 17, (3), 391-415, DOI: 10.1007/s10111-014-0288-9.
- [P283] Mowafi, Y., Zhang, D. (2007), A user-centered approach to context-awareness in mobile computing, *International Conference on Mobile and Ubiquitous Systems: Computing, Networking and Services*, DOI: 10.1109/MOBIQ.2007.4450989.
- [P284] Muro, M., Amoretti, M., Zanichelli, F., Conte, G. (2012), Towards a flexible middleware for context-aware pervasive and wearable systems, *Medical and Biological Engineering and Computing*, 50, (11), 1127-1136, DOI: 10.1007/s11517-012-0905-9.
- [P285] Myrhaug, H., Whitehead, N., Goker, A., Faegri, T.E., Lech, T.C. (2004), AmbieSense A System and Reference Architecture for Personalised Context-Sensitive Information Services for Mobile Users, *European Symposium on Ambient Intelligence*, 327-338.
- [P286] Nava-Muñoz, S., Morán, A.L., Favela, J. (2010), Providing awareness of elder's situations of care through a context-aware notification environment: A preliminary evaluation, *International Conference on Intelligent Environments*, 134-139, DOI: 10.1109/IE.2010.32.
- [P287] Nawrocki P., Sniezynski B. (2017), Autonomous Context-Based Service Optimization in Mobile Cloud Computing, Journal of Grid Computing, 15, (3), 343-356, CEUR-WS, DOI: 10.1007/s10723-017-9406-2.
- [P288] Ngo, H.Q., Shehzad, A., Liaquat, S., Riaz, M., Lee, S. (2004), Developing Context-Aware Ubiquitous Computing Systems with a Unified Middleware Framework, *International Conference on Embedded and Ubiquitous Computing*, 672-681.
- [P289] Nguyen, T., Loke, S.W., Torabi, T., Lu, H. (2011), PlaceComm: A Framework for Context-aware Applications in Place-based Virtual Communities, *Journal of Ambient Intelligence and Smart Environments*, 3, (1), 51-64, IOS Press.
- [P290] Nickelsen, A., Paternò, F., Grasselli, A., Schmidt, K.-U., Martin, M., Mureddu, F. (2012), OPEN Platform for Migration of Interactive Services: Architecture and Evaluation, *Int. J. Adapt. Resilient Auton. Syst.*, 3, (2), 18-43, Springer-Verlag, DOI: 10.4018/jaras.2012040102.
- [P291] Nicklas, D., Mitschang, B. (2004), On building location aware applications using an open platform based on the NEXUS Augmented World Model, *Software and Systems Modeling*, 3, (4), 303-313, DOI: 10.1007/s10270-004-0055-0.
- [P292] Nilsson, E.G., Floch, J., Hallsteinsen, S., Stav, E. (2006), Using a patterns-based modelling language and a model-based adaptation architecture to facilitate adaptive user interfaces, *International Workshop on Design, Specification, and Verification* of Interactive Systems, 234-247, DOI: 10.1007/978-3-540-69554-7_19.
- [P293] Nino, C.P., Marques, J., Barbosa, D.N.F., Geyer, C.F.R., Barbosa, J.L.V., Augustin, I. (2007), Context-aware model in a ubiquitous learning environment, *International Conference on Pervasive Computing and Communications Workshops*, 182-186, DOI: 10.1109/PERCOMW.2007.30.
- [P294] Norrie, M.C., Signer, B., Grossniklaus, M., Belotti, R., Decurtins, C., Weibel, N. (2007), Context-aware platform for mobile data management, *Wireless Networks*, 13, (6), 855-870, DOI: 10.1007/s11276-006-9858-y.
- [P295] Nunes, V.T., Werner, C.M.L., Santoro, F.M. (2011), Dynamic process adaptation: A context-aware approach, International Conference on Computer Supported Cooperative Work in Design, 97-104, DOI: 10.1109/CSCWD.2011.5960061.
- [P296] Oh, M., Lee, J., Chang, B.-M., Ahn, J., Doh, K.-G. (2007), A programming environment for ubiquitous computing environment, SIGPLAN Notices, 42, (4), 14-22.

- [P297] Oh, Y., Choi, A., Woo, W. (2010), u-BabSang: a context-aware food recommendation system, *The Journal of Supercomputing*, 54, (1), 61-81, DOI: 10.1007/s11227-009-0314-5.
- [P298] Oh, Y., Han, J., Woo, W. (2010), A context management architecture for large-scale smart environments, *Communications Magazine*, 48, (3), 118-126, DOI: 10.1109/MCOM.2010.5434383.
- [P299] Olaru, A., Florea, A.M., El Fallah Seghrouchni, A. (2013), A Context-Aware Multi-Agent System as a Middleware for Ambient Intelligence, *Mobile Networks and Applications*, 18, (3), 429-443, DOI: 10.1007/s11036-012-0408-9.
- [P300] Ongenae, F., Famaey, J., Verstichel, S., De Zutter, S., Latré, S., Ackaert, A., Verhoeve, P., De Turck, F. (2014), Ambientaware continuous care through semantic context dissemination, *BMC Medical Informatics and Decision Making*, 14, (1), 1-47, DOI: 10.1186/1472-6947-14-97.
- [P301] Opu, Md.F.R., Biswas, E., Habiba, M., Kim, C. (2013), Architecture of a context aware framework for automated mobile device configuration, *International Conference on Grid and Pervasive Computing*, 299-308, DOI: 10.1007/978-3-642-38027-3_32.
- [P302] Ortmann, S., Langendörfer, P., Maaser, M. (2007), A Self-configuring Privacy Management Architecture for Pervasive Systems, Proceedings of the 5th ACM International Workshop on Mobility Management and Wireless Access, 184-187, Springer Verlag, DOI: 10.1145/1298091.1298125.
- [P303] Osullivan, M.J., Grigoras, D. (2016), Context aware mobile cloud services: A user experience oriented middleware for mobile cloud computing, *Proceedings - 2016 4th IEEE International Conference on Mobile Cloud Computing, Services, and Engineering, MobileCloud 2016*, 67-72, Elsevier B.V., DOI: 10.1109/MobileCloud.2016.13.
- [P304] Otebolaku, A.M., Andrade, M.T. (2015), Context-aware media recommendations for smart devices, Journal of Ambient Intelligence and Humanized Computing, 6, (1), 13-36, DOI: 10.1007/s12652-014-0234-y.
- [P305] Otebolaku, A.M., Iyilade, J.S., Adigun, M.O. (2008), CAAM: A context aware adaptation model for mobile grid service infrastructure, *International Conference on Computational Science and Engineering*, 419-425, DOI: 10.1109/CSEW.2008.37.
- [P306] Pantsar-Syväniemi, S. (2011), Adaptable context-aware micro-architecture, International Conference on Grid and Pervasive Computing, 125-132, DOI: 10.1007/978-3-642-27916-4_14.
- [P307] Papakonstantinou, S., Brujic-Okretic, V (2009), Framework for context-aware smartphone applications, *The Visual Computer*, 25, (12), 1121-1132, DOI: 10.1007/s00371-009-0391-8.
- [P308] Pardo, E., Espes, D., Le-Parc, P. (2016), A Framework for Anomaly Diagnosis in Smart Homes Based on Ontology, Procedia Computer Science, 83, 545-552, Springer Verlag, DOI: 10.1016/j.procs.2016.04.255.
- [P309] Park, H.-S., Choi, D.-H., Jeong, Y.-H., Choi, T.-U. (2006), A CMQ Middleware Architecture for Multimedia Application in Ubiquitous Environment, *International Conference on Advanced Communication Technology*, 1341-1345, DOI: 10.1109/ICACT.2006.206219.
- [P310] Park, J., Lee, J.-K., Paik, Y., Lee, M., Narayanaswami, C. (2006), Device context discovery system for context-aware services in ubiquitous device collaboration environment, *International Workshop on System Support for Future Mobile Computing Applications*, 25-34, DOI: 10.1109/FUMCA.2006.4.
- [P311] Park, N.-S., Lee, K.-W., Kim, H. (2005), A Middleware for Supporting Context-Aware Services in Mobile and Ubiquitous Environment, *International Conference on Mobile Business*, 694-697, DOI: 10.1109/ICMB.2005.7.
- [P312] Park, S.-K., Kim, D.-C., Kang, B.-S. (2006), Context-aware middleware architecture for intelligent service in mobile environment, *International Conference on Computer and Information Technology*, DOI: 10.1109/CIT.2006.65.
- [P313] Paspallis, N., Papadopoulos, G.A. (2014), A pluggable middleware architecture for developing context-aware mobile applications, *Personal and Ubiquitous Computing*, 18, (5), 1099-1116, DOI: 10.1007/s00779-013-0722-7.
- [P314] Pederson, T., Ardito, C., Bottoni, P., Costabile, M. (2008), A General-Purpose Context Modeling Architecture for Adaptive Mobile Services, *ER Workshops on Advances in Conceptual Modeling: Challenges and Opportunities*, 208-217, Springer-Verlag, DOI: 10.1007/978-3-540-87991-6_26.
- [P315] Pederson, T., Surie, D. (2007), Towards an activity-aware wearable computing platform based on an egocentric interaction model, *International Symposium on Ubiquitious Computing Systems*, 211-227.
- [P316] Peleg, M., Shahar, Y., Quaglini, S., Fux, G., García-Sáez, G., Goldstein, A., Hernando, M.E., Klimov, D., Martínez-Sarriegui, I., Napolitano, C., Parimbelli, E., Rigla, M., Sacchi, L., Shalom, E., Soffer, P., (2017), MobiGuide: a personalized and patientcentric decision-support system and its evaluation in the atrial fibrillation and gestational diabetes domains, *User Modeling and User-Adapted Interaction*, 27, (2), DOI: 10.1007/s11257-017-9190-5.
- [P317] Pernas, A.M., Oliveira, J.P.M.D. (2012), Enabling situation-aware behavior in web-based learning systems, International Conference of the Chilean Computer Science Society, 153-160, DOI: 10.1109/SCCC.2011.21.
- [P318] Petersen, F., Bartolomeo, G., Pluke, M., Kovacikova, T. (2009), An architectural framework for context sensitive personalization, *International Conference on Mobile Technology*, *Application & Systems*, DOI: 10.1145/1710035.1710072.
- [P319] Petrelli, D., Not, E. (2005), User-Centred Design of Flexible Hypermedia for a Mobile Guide: Reflections on the HyperAudio Experience, User Modeling and User-Adapted Interaction, 15, (3), 303-338, DOI: 10.1007/s11257-005-8816-1.
- [P320] Pfautz, S.L., Ganberg, G., Fouse, A., Schurr, N. (2015), A general context-aware framework for improved human-system interactions, AI Magazine, 36, (2), 42-49, Springer Verlag.
- [P321] Pittarello, F. (2006), A Simple Story: Using an Agents' Based Context-aware Architecture for Storytelling, Proceedings of the Third International Conference on Technologies for Interactive Digital Storytelling and Entertainment, 25-36, Inderscience Publishers, DOI: 10.1007/11944577_3.

- [P322] Pittarello, F. (2011), Designing a context-aware architecture for emotionally engaging mobile storytelling, Conference on Human-Computer Interaction, 144-151, DOI: 10.1007/978-3-642-23774-4_14.
- [P323] Pombinho, P., Afonso, A.P., Carmo, M.B. (2011), Chameleon A context adaptive visualization framework for a mobile environment, *International Conference on Information Visualisation*, 151-157, DOI: 10.1109/IV.2011.34.
- [P324] Popovici, D., Desertot, M., Lecomte, S., Delot, T. (2013), A framework for mobile and context-aware applications applied to vehicular social networks, *Social Network Analysis and Mining*, 3, (3), 329-340, DOI: 10.1007/s13278-012-0073-9.
- [P325] Pötter, H.B., Sztajnberg, A. (2016), Adapting heterogeneous devices into an IoT context-aware infrastructure, Proceedings -11th International Symposium on Software Engineering for Adaptive and Self-Managing Systems, SEAMS 2016, 64-74, Association for Computing Machinery, Inc, DOI: 10.1145/2897053.2897072.
- [P326] Poulcheria, B., Costas, V. (2012), A context management architecture for m-commerce applications, *Central European Journal of Computer Science*, 2, (2), 87-117, DOI: 10.2478/s13537-012-0010-z.
- [P327] Preuveneers, D., Berbers, Y. (2007), Towards context-aware and resource-driven self-adaptation for mobile handheld applications, *Symposium on Applied Computing*, 1165-1170, DOI: 10.1145/1244002.1244255.
- [P328] Preuveneers, D., Berbers, Y., Joosen, W. (2016), SAMURAI: A batch and streaming context architecture for large-scale intelligent applications and environments, *Journal of Ambient Intelligence and Smart Environments*, 8, (1), 63-78, DOI: 10.3233/AIS-150357.
- [P329] Prost, S., Schrammel, J., Röderer, K., Tscheligi, M. (2013), Contextualise! Personalise! Persuade!: A Mobile HCI Framework for Behaviour Change Support Systems, *Proceedings of the 15th International Conference on Human-computer Interaction* with Mobile Devices and Services, 510-515, IGI Global, DOI: 10.1145/2493190.2494434.
- [P330] Pu, F., Liu, L., Shen, Y., Cao, Q. (2010), P2P-based semantic policy infrastucture for pervasive computing, *International Conference on Computational Intelligence and Security*, 541-545, DOI: 10.1109/CIS.2010.124.
- [P331] Pung, H., Gu, T., Xue, W., Palmes, P., Zhu, J., Ng, W., Tang, C., Chung, N. (2009), Context-aware middleware for pervasive elderly homecare, *Journal on Selected Areas in Communications*, 27, (4), 510-524, DOI: 10.1109/JSAC.2009.090513.
- [P332] Quade, M., Lehmann, G., Engelbrecht, K.-P., Roscher, D., Albayrak, S. (2015), Automated usability evaluation of model-based adaptive user interfaces for users with special and specific needs by simulating user interaction, *User Modeling and Adaptation* for Daily Routines: Providing Assistance to People with Special Needs, 219-247, IOS Press, DOI: 10.1007/978-1-4471-4778-7_9.
- [P333] Raento, M., Oulasvirta, A., Petit, R., Toivonen, H. (2005), ContextPhone: a prototyping platform for context-aware mobile applications, *IEEE Pervasive Computing*, 4, (2), 51-59, DOI: 10.1109/MPRV.2005.29.
- [P334] Rahman, M.A., Kim, H.N., El Saddik, A., Gueaieb, W. (2014), A context-aware multimedia framework toward personal social network services, *Multimedia Tools and Applications*, 71, (3), 1717-1747, DOI: 10.1007/s11042-012-1302-y.
- [P335] Rangel, P., De Carvalho, Jr. J.G., Ramirez, M.R., De Souza, J.M. (2011), RT-MLR: A hybrid framework for context-aware systems, *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, 6679 LNAI, 92-99, DOI: 10.1007/978-3-642-21222-2_12.
- [P336] Rasch, K., Li, F., Sehic, S., Ayani, R., Dustdar, S. (2011), Context-driven personalized service discovery in pervasive environments, *World Wide Web*, 14, (4), 295-319, DOI: 10.1007/s11280-011-0112-x.
- [P337] Reiff-Marganiec, S., Truong, H.-L., Casella, G., Dorn, C., Dustdar, S., Moretzky, S. (2008), The inContext pervasive collaboration services architecture, *European Conference on a Service-Based Internet*, 134-146, DOI: 10.1007/978-3-540-89897-9_12.
- [P338] Repo, P. (2004), Facilitating user interface adaptation to mobile devices, Nordic Conference on Human-Computer Interaction, 433-436, DOI: 10.1145/1028014.1028088.
- [P339] Repo, P., Riekki, J. (2004), Middleware support for implementing context-aware multimodal user interfaces, *International Conference on Mobile and Ubiquitous Multimedia*, 221-227, DOI: 10.1145/1052380.1052411.
- [P340] Riboni, D., Bettini, C. (2011), COSAR: hybrid reasoning for context-aware activity recognition, *Personal and Ubiquitous Computing*, 15, (3), 271-289, DOI: 10.1007/s00779-010-0331-7.
- [P341] Righetti, X., Peternier, A., Hopmann, M., Thalmann, D. (2008), Design and implementation of a wearable, context-aware MR framework for the Chloe@University application, *International Conference on Emerging Technologies and Factory Automation*, 1362-1369, DOI: 10.1109/ETFA.2008.4638576.
- [P342] Roda, C., Rodríguez A.C., López-Jaquero, V., Navarro, E., González, P. (2017), A Multi-Agent System for Acquired Brain Injury rehabilitation in Ambient Intelligence environments, *Neurocomputing*, 231, 11-18, Slovak Academy of Sciences, DOI: 10.1016/j.neucom.2016.04.066.
- [P343] Roda, C., Rodríguez, A., López-Jaquero, V., González, P., Navarro, E. (2015), A multi-agent system in ambient intelligence for the physical rehabilitation of older people, *Conference on Practical Applications of Agents and Multi-Agent Systems*, 113-123, DOI: 10.1007/978-3-319-19629-9_13.
- [P344] Rodríguez-Hernández, M. C., Ilarri, S. (2014), Towards a context-aware mobile recommendation architecture, *International Conference on Mobile Web and Information Systems*, 56-70, DOI: 10.1007/978-3-319-10359-4_5.
- [P345] Rodríguez-Hernández, M.D.C., Ilarri, S. (2016), Pull-based recommendations in mobile environments, *Computer Standards and Interfaces*, 44, 185-204, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1016/j.csi.2015.08.002.
- [P346] Roggen, D., Tröster, G., Lukowicz, P., Ferscha, A., Del R. Millán, J., Chavarriaga, R. (2013), Opportunistic Human Activity and Context Recognition, *Computer*, 46, (2), 36-45, IEEE Computer Society Press, DOI: 10.1109/MC.2012.393.

- [P347] Román, M., Hess, C., Cerqueira, R., Ranganathan, A., Campbell, R.H., Nahrstedt, K. (2002), A middleware infrastructure for active spaces, *IEEE Pervasive Computing*, 1, (4), 74-83, DOI: 10.1109/MPRV.2002.1158281.
- [P348] Rosaci, D., Sarné, G.M.L. (2006), MASHA: A multi-agent system handling user and device adaptivity of Web sites, User Modeling and User-Adapted Interaction, 16, (5), 435-462, DOI: 10.1007/s11257-006-9015-4.
- [P349] Rothenpieler, P., Carlson, D., Schrader, A. (2013), SmartAssist: Open infrastructure and platform for AAL services, International Conference on Distributed, Ambient, and Pervasive Interactions, 495-504, DOI: 10.1007/978-3-642-39351-8_54.
- [P350] Roy, N., Pallapa, G., Das, S.K. (2007), A middleware framework for ambiguous context mediation in smart healthcare application, *International Conference on Wireless and Mobile Computing, Networking and Communications*, DOI: 10.1109/WIMOB.2007.4390866.
- [P351] Sabagh, A.A.A., Al-Yasiri, A. (2015), GECAF: a framework for developing context-aware pervasive systems, *Computer Science Research and Development*, 30, (1), 87-103, DOI: 10.1007/s00450-013-0248-2.
- [P352] Salmeri, A., Licciardi, C. A., Lamorte, L., Valla, M., Giannantonio, R., Sgroi, M. (2009), An Architecture to Combine Context Awareness and Body Sensor Networks for Health Care Applications, *International Conference on Smart Homes and Health Telematics*, 90-97, Springer-Verlag, DOI: 10.1007/978-3-642-02868-7_12.
- [P353] Sánchez-Garzón, I., González-Ferrer, A., Fernández-Olivares, J. (2014), A knowledge-based architecture for the management of patient-focused care pathways, *Applied Intelligence*, 40, (3), 497-524, DOI: 10.1007/s10489-013-0466-0.
- [P354] Sannino, G., De Pietro, G. (2011), An evolved ehealth monitoring system for a nuclear medicine department, *International Conference on Developments in eSystems Engineering*, 3-6, DOI: 10.1109/DeSE.2011.67.
- [P355] Santos, A.C., Diniz, P.C., Cardoso, J.M.P., Ferreira, D.R. (2011), A domain-specific language for the specification of adaptable context inference, *International Conference on Embedded and Ubiquitous Computing*, 268-273, DOI: 10.1109/EUC.2011.4.
- [P356] Santos, D.F.S., Gorgônio, K.C., Perkusich, A., Almeida, H.O. (2016), A Standard-Based and Context-Aware Architecture for Personal Healthcare Smart Gateways, *Journal of Medical Systems*, 40, (10), Springer-Verlag, DOI: 10.1007/s10916-016-0580-8.
- [P357] Saremi, A. (2009), CaDSS: A framework for enhancing decision making in pervasive computing environment, UKSim European Symposium on Computer Modelling and Simulation, 17-22, DOI: 10.1109/EMS.2009.105.
- [P358] Satoh I. (2016), Self-Adaptively Auto-scaling for Mobile Cloud Applications, *Procedia Computer Science*, 94, 9-16, Springer Verlag, DOI: 10.1016/j.procs.2016.08.006.
- [P359] Satoh, I. (2009), A context-aware service framework for large-scale ambient computing environments, *International Conference on Pervasive Services*, 199-208, DOI: 10.1145/1568199.1568233.
- [P360] Scerri, S., Debattista, J., Attard, J., Rivera, I. (2015), A semantic infrastructure for personalizable context-aware environments, *AI Magazine*, 36, (2), 73-84, Springer International Publishing.
- [P361] Schmidt, A., Aidoo, K.A., Takaluoma, A., Tuomela, U., Van Laerhoven, K., Van de Velde, W. (1999), Advanced interaction in context, *International Symposium on Handheld and Ubiquitous Computing*, 89-101, DOI: 10.1007/3-540-48157-5_10.
- [P362] Schmidt, A., Winterhalter, C. (2004), User context aware delivery of e-learning material: Approach and architecture, *Journal of Universal Computer Science*, 10, (1), 38-46.
- [P363] Schmohl, R., Baumgarten, U. (2008), A generalized context-aware architecture in heterogeneous mobile computing environments, *International Conference on Wireless and Mobile Communications*, 118-124, DOI: 10.1109/ICWMC.2008.59.
- [P364] Schrader, A., Carlson, D. V., Busch, D. (2008), Modular framework support for context-aware mobile cinema, *Personal and Ubiquitous Computing*, 12, (4), 299-306, DOI: 10.1007/s00779-007-0151-6.
- [P365] Schroeder, A., Van Zwaag, M.D., Hammer, M. (2008), A middleware architecture for human-centred pervasive adaptive applications, *International Conference on Self-Adaptive and Self-Organizing Systems Workshops*, 138-143, DOI: 10.1109/SASOW.2008.38.
- [P366] Sebek, J., Cerny, T., Richta, K. (2016), Adaptive application structure design for Java EE applications, *Proceedings of the 2016 Research in Adaptive and Convergent Systems, RACS 2016*, 159-164, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1145/2987386.2987417.
- [P367] Seidel, M., Zug, S. (2016), Context aware architecture for distributed robotics, *IEEE International Conference on Emerging Technologies and Factory Automation, ETFA*, 2016-November, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1109/ETFA.2016.7733646.
- [P368] Seiger, R., Huber, S., Heisig, H., Aßmann, U., (2017), Toward a framework for self-adaptive workflows in cyber-physical systems, *Software & Systems Modeling*, DOI: 10.1007/s10270-017-0639-0.
- [P369] Selviandro, N., Sabariah, M.K., Novandy Purna, D.R.D. (2016), Enhancing QoS context-Aware ubiquitous learning by utilizing logical and physical characteristic of device, 2016 4th International Conference on Information and Communication Technology, ICoICT 2016, IEEE Computer Society, DOI: 10.1109/ICoICT.2016.7571963.
- [P370] Sendín, M. (2006), Implicit plasticity framework: A client-side generic framework for context-awareness, International Conference on Ubiquitous Computing.
- [P371] Serral, E., Valderas, P., Pelechano, V. (2015), Addressing the evolution of automated user behaviour patterns by runtime model interpretation, *Software and Systems Modeling*, 14, (4), 1387-1420, DOI: 10.1007/s10270-013-0371-3.
- [P372] Serrano, J. M., Justo, J., Marín, R., Serrat, J. (2006), Context architecture as a solution for managing pervasive applications, *Network Operations and Management Symposium*, 1-6.

- [P373] Shakshuki, E.M., Alghamdi, W., Sheltami, T. (2013), Integrated context-aware driver assistance system architecture, *Mobile Information Systems*, 9, (3), 225-240, DOI: 10.3233MIS-130159.
- [P374] Shakshuki, E.M., Reid, M., Sheltami, T.R. (2015), An adaptive user interface in healthcare, *Procedia Computer Science*, 56, (1), 49-58, DOI: 10.1016/j.procs.2015.07.182.
- [P375] Shams, H., Zamanifar, K. (2014), MVCC: An architectural pattern for developing context-aware frameworks, *Procedia Computer Science*, 34, 344-351, DOI: 10.1016/j.procs.201407.035.
- [P376] Shi, L., Lin, F., Yang, T., Qi, J., Ma, W., Xu, S. (2014), Context-based Ontology-driven Recommendation Strategies for Tourism in Ubiquitous Computing, *Wireless Personal Communications*, 76, (4), 731-745, DOI: 10.1007/s11277-013-1550-9.
- [P377] Shrestha, A., Biel, B., Griebe, T., Gruhn, V. (2011), A framework for building and operating context-aware mobile applications, International Conference on Mobile Wireless Middleware, Operating Systems, and Applications, 135-142, DOI: 10.1007/978-3-642-30607-5_13.
- [P378] Sielis, G., Tzanavari, A., Papadopoulos, G. A. (2017), ArchReco: a software tool to assist software design based on context aware recommendations of design patterns, *Journal of Software Engineering Research and Development*, 5, (1), DOI: 10.1186/s40411-017-0036-y.
- [P379] Smirnov, A. V., Kashevnik, A. M., Ponomarev, A. (2017), Context-based infomobility system for cultural heritage recommendation: Tourist Assistant-TAIS, *Personal and Ubiquitous Computing*, 21, (2), DOI: 10.1007/s00779-016-0990-0.
- [P380] Smirnov, A.V., Shilov, N.G., Ponomarev, A.V., Kashevnik, A.M., Parfenov, V.G. (2014), Group context-aware recommendation systems, *Scientific and Technical Information Processing*, 41, (5), 325-334, DOI: 10.3103/S0147688214050050.
- [P381] Soldatos, J., Dimakis, N., Stamatis, K., Polymenakos, L. (2007), A breadboard architecture for pervasive context-aware services in smart spaces: middleware components and prototype applications, *Personal and Ubiquitous Computing*, 11, (3), 193-212, DOI: 10.1007/s00779-006-0102-7.
- [P382] Song, C.W., Lee, D., Chung, K.Y., Rim, K.W., Lee, J.H. (2014), Interactive middleware architecture for lifelog based context awareness, *Multimedia Tools and Applications*, 71, (2), 813-826, DOI: 10.1007/s11042-013-1362-7.
- [P383] Sottet, J.-S., Ganneau, V., Calvary, G., Coutaz, J., Demeure, A., Favre, J.-M., Demumieux, R. (2007), Model-driven adaptation for plastic user interfaces, *Conference on Human-Computer Interaction*, 397-410.
- [P384] Sridevi, S., Sayantani, B., Amutha, K.P., Mohan, C.M., Pitchiah, R. (2010), Context aware health monitoring system, International Conference on Medical Biometrics, 249-257, DOI: 10.1007/978-3-642-13923-9_27.
- [P385] Srirama, S. N. (2017), Mobile web and cloud services enabling Internet of Things, CSI Transactions on ICT, 5, (1), DOI: 10.1007/s40012-016-0139-3.
- [P386] Stavroulaki, V., Koutsouris, N., Kritikou, Y., Demestichas, P. (2014), Cognitive Management of Devices in the Wireless World, Wireless Personal Communications, 75, (4), 2289-2322, DOI: 10.1007/s11277-013-1468-2.
- [P387] Stellingwerff, L., Pazienza, G.E. (2014), An agent-based architecture to model and manipulate context knowledge, International Conference on Practical Applications of Agents and Multi-Agent Systems, 256-267, DOI: 10.1007/978-3-319-07551-8_22.
- [P388] Stock, O., Zancanaro, M., Busetta, P., Callaway, C., Krüger, A., Kruppa, M., Kuflik, T., Not, E., Rocchi, C. (2007), Adaptive, intelligent presentation of information for the museum visitor in PEACH, *User Modeling and User-Adapted Interaction*, 17, (3), 257-304, DOI: 10.1007/s11257-007-9029-6.
- [P389] Su, J.M., Huang, C.F. (2014), An easy-to-use 3D visualization system for planning context-aware applications in smart buildings, *Computer Standards and Interfaces*, 36, (2), 312-326, DOI: 10.1016/j.csi.2012.07.004.
- [P390] Su, J.M., Tseng, S.S., Lin, H.Y., Chen, C.H (2011), A personalized learning content adaptation mechanism to meet diverse user needs in mobile learning environments, User Modeling and User-Adapted Interaction, 21, (1), 5-49, DOI: 10.1007/s11257-010-9094-0.
- [P391] Sun, J., YongPing, Z., Fan, J. (2010), Towards a context-aware middleware in smart car space, International Conference on Genetic and Evolutionary Computing, 276-279, DOI: 10.1109/ICGEC.2010.75.
- [P392] Sun, W., Zhang, X., Yuan, Y., Han, T. (2013), Context-aware web service composition framework based on agent, International Conference on Information Technology and Applications, 30-34, DOI: 10.1109/ITA.2013.13.
- [P393] Tadj, C., Ngantchaha, G. (2006), Context handling in a pervasive computing system framework, International Conference on Mobile Technology, Applications and Systems.
- [P394] Taentzer G., Vaupel S. (2016), Model-Driven development of mobile applications: Towards context-aware apps of high quality ?, *CEUR Workshop Proceedings*, 1591, 17-29, Institute of Electrical and Electronics Engineers Inc..
- [P395] Takatsuka, H., Saiki, S., Matsumoto, S., Nakamura, M. (2014), Design and implementation of rule-based framework for context-aware services with web services, *International Conference on Information Integration and Web-based Applications* & Services, 233-242, DOI: 10.1145/2684200.2684310.
- [P396] Tang, L., Ni, Z., Xiong, H., Zhu, H. (2015), Locating targets through mention in Twitter, World Wide Web, 18, (4), 1019-1049, DOI: 10.1007/s11280-014-0299-8.
- [P397] Tegegne, T., Kanagwa, B., Van Der Weide, T. (2010), eHealth service discovery framework for a low infrastructure context, International Conference on Computer Technology and Development, 606-610, DOI: 10.1109/ICCTD.2010.5645971.
- [P398] Temdee, P. (2014), Ubiquitous Learning Environment: Smart Learning Platform with Multi-Agent Architecture, Wireless Personal Communications, 76, (3), 627-641, DOI: 10.1007/s11277-014-1730-2.

- [P399] Testoni, A., Di Felice, M. (2017), A software architecture for generic human activity recognition from smartphone sensor data, 2017 IEEE International Workshop on Measurement and Networking, M and N 2017 - Proceedings, Springer Verlag, DOI: 10.1109/IWMN.2017.8078368.
- [P400] Thiele, O., Knapp, H., Schader, M., Prat, N. (2006), A context-aware architecture for mobile knowledge management, International Conference on E-Business and Telecommunication Networks, 359-370.
- [P401] Thyagaraju, G.S., Kulkarni, U.P. (2012), Design and implementation of user context aware recommendation engine for mobile using Bayesian network, fuzzy logic and rule base, *International Journal of Pervasive Computing and Communications*, 8, (2), 133-157, DOI: 10.1108/17427371211245364.
- [P402] Tiberghien, T., Mokhtari, M., Aloulou, H., Biswas, J., Zhu, J., Lee, V.Y. (2011), Handling user interface plasticity in assistive environment: UbiSMART framework, *International Conference on Smart Homes and Health Telematics*, 256-260, DOI: 10.1007/978-3-642-21535-3_37.
- [P403] Tokunaga, S., Horiuchi, H., Takatsuka, H., Saiki, S., Matsumoto, S., Nakamura, M., Yasuda, K. (2016), Towards personalized and context-aware reminder service for people with dementia, *Proceedings of the International Joint Conference on Neural Networks*, 2016-October, 2946-2953, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1109/IJCNN.2016.7727572.
- [P404] Tortorella, R. A. W., Kinshuk, D., Chen, N. S., (2017), Framework for designing context-aware learning systems, *Education and Information Technologies*, DOI: 10.1007/s10639-017-9591-4.
- [P405] Tortorella, R. A.W., Kinshuk (2017), A mobile context-aware medical training system for the reduction of pathogen transmission, *Smart Learning Environments*, 4, (1), DOI: 10.1186/s40561-017-0043-9.
- [P406] Tortorella, R.A.W., Graf, S. (2015), Considering learning styles and context-awareness for mobile adaptive learning, *Education and Information Technologies*, 22, (1), 297-315, DOI: 10.1007/s10639-015-9445-x.
- [P407] Tosin, C.E., Lima, L. (2009), A reflexive infrastructure for context-aware applications, International Conference on Mobile Ubiquitous Computing, Systems, Services, and Technologies, 102-107, DOI: 10.1109/UBICOMM.2009.14.
- [P408] Tran, M.H., Han, J., Colman, A. (2009), Social context: Supporting interaction awareness in ubiquitous environments, International Conference on Mobile and Ubiquitous Systems: Networking and Services, 1-10, DOI: 10.4108/ICST.MOBIQUITOUS2009.6835.
- [P409] Trullemans, S., Signer, B. (2016), A multi-layered context modelling approach for end users, expert users and programmers, CEUR Workshop Proceedings, 1602, 36-41, CEUR-WS.
- [P410] Tu, M., Chang, Y.K., Chen, Y.T. (2016), A context-aware recommender system framework for iot based interactive digital signage in urban space, ACM International Conference Proceeding Series, 24-25-May-2016, 39-42, Elsevier, DOI: 10.1145/2962735.2962736.
- [P411] Uhm, Y., Hwang, Z., Lee, M., Kim, Y., Kim, G., Park, S. (2007), A context-aware dynamic service-oriented system for reconstructing adaptive services by the management of rule and ontology, *International Conference on Convergence Information Technology*, 83-88, DOI: 10.1109/ICCIT.2007.4420241.
- [P412] Urbieta, A., González-Beltrán, A., Ben Mokhtar, S., Anwar Hossain, M., Capra, L. (2017), Adaptive and context-aware service composition for IoT-based smart cities, *Future Generation Computer Systems*, 76, 262-274, DOI: 10.1016/j.future.2016.12.038.
- [P413] Vahdat-Nejad, H., Zamanifar, K., Nematbakhsh, N. (2013), Context-Aware middleware architecture for smart home environment, *International Journal of Smart Home*, 7, (1), 77-86.
- [P414] Vale, S., Hammoudi, S. (2009), ODM-based architecture for the development of mobile context-aware applications, International Conference on PErvsive Technologies Related to Assistive Environments, 1-4, DOI: 10.1145/1579114.1579123.
- [P415] Van Kranenburg, H., Bargh, M.S., Iacob, S., Peddemors, A. (2006), A context management framework for supporting contextaware distributed applications, *Communications Magazine*, 44, (8), 67-74, DOI: 10.1109/MCOM.2006.1678112.
- [P416] Van Sinderen, M.J., Van Halteren, A.T., Wegdam, M., Meeuwissen, H.B., Eertink, E.H. (2006), Supporting context-aware mobile applications: An infrastructure approach, *IEEE Communications Magazine*, 44, (9), 96-104, DOI: 10.1109/MCOM.2006.1705985.
- [P417] Vaupel, S., Wlochowitz, D., Taentzer, G. (2016), A generic architecture supporting context-aware data and transaction management for mobile applications, *Proceedings - International Conference on Mobile Software Engineering and Systems,* MOBILESoft 2016, 111-122, Elsevier B.V., DOI: 10.1145/2897073.2897091.
- [P418] Vecchia, G. D., Esposito, M. (2010), A Pervasive System for Nuclear Medicine Department, Wireless Personal Communications, 55, (1), 105-120, DOI: 10.1007/s11277-009-9789-x.
- [P419] Vecchiato, D., Araújo, N., Maciel, C., Viterbo, J., El, A. (2014), Using agents towards providing security on a context-aware architecture, *International Workshop on Agents and CyberSecurity*, 1-4, DOI: 10.1145/2602945.2602954.
- [P420] Venkatesh, J., Akşanli, B., Chan, C.S., Akyürek, A.S., Rosing, T.Š. (2017), Scalable-Application Design for the IoT, *IEEE Software*, 34, (1), 62-70, IEEE Conferences, DOI: 10.1109/MS.2017.4.
- [P421] Viterbo, J., Sacramento, V., Rocha, R., Baptista, G., Malcher, M., Endler, M. (2008), A middleware architecture for contextaware and location-based mobile applications, *Software Engineering Workshop*, 52-61, DOI: 10.1109/SEW.2008.7.
- [P422] Vogt, J., Meier, A. (2010), An adaptive user interface framework for eHealth services based on UIML, *BLED Proceedings*, 409-422.

- [P423] Vuković, M., Kotsovinos, E., Robinson, P. (2007), An architecture for rapid, on-demand service composition, Service Oriented Computing and Applications, 1, (4), 197-212, DOI: 10.1007/s11761-007-0016-x.
- [P424] Wang, B., Tang, X. (2014), Designing a self-adaptive and context-aware service composition system, Computers, Communications and IT Applications Conference, 155-160, DOI: 10.1109/ComComAp.2014.7017188.
- [P425] Wang, H.M., Ding, B., Shi, D.X., Cao, J.N., Chan, A.T.S. (2015), Auxo: an architecture-centric framework supporting the online tuning of software adaptivity, *Science China Information Sciences*, 58, (9), DOI: 10.1007/s11432-015-5307-9.
- [P426] Wang, P., Li, H., Zhang, B. (2014), A context-aware workflow system framework and scheduling algorithm, Journal of Information and Computational Science, 11, (18), 6689-6701, DOI: 10.12733/jics20105082.
- [P427] Wang, P., Zhao, J., Liao, Q. (2012), A cross-platform context-aware application developing framework for mobile terminals, International Conference on Cloud Computing and Intelligence Systems, 3, 1272-1276, DOI: 10.1109/CCIS.2012.6664589.
- [P428] Wei, E.J.Y., Chan, A.T.S. (2013), CAMPUS: A middleware for automated context-aware adaptation decision making at run time, *Pervasive and Mobile Computing*, 9, (1), 35-56, DOI: 10.1016/j.pmcj.2011.10.002.
- [P429] Williams, E., Gray, J. (2014), Contextion: A framework for developing context-aware mobile applications, *International Workshop on Mobile Development Lifecycle*, 27-31, DOI: 10.1145/2688412.2688416.
- [P430] Wohltorf, J., Cissée, R., Rieger, A. (2005), BerlinTainment: An agent-based context-aware entertainment planning system, *Communications Magazine*, 43, (6), 102-109, DOI: 10.1109/MCOM.2005.1452837.
- [P431] Wójtowicz, A., Chmielewski, J. (2017), Technical feasibility of context-aware passive payment authorization for physical points of sale, *Personal and Ubiquitous Computing*, 21, (6), DOI: 10.1007/s00779-017-1035-z.
- [P432] Wongpatikaseree, K., Kim, J., Makino, Y., Lim, A. O., Tan, Y. (2013), Architecture for Organizing Context-Aware Data in Smart Home for Activity Recognition System, *International Conference on Distributed, Ambient, and Pervasive Interactions*, 173-182, Springer-Verlag New York, Inc., DOI: 10.1007/978-3-642-39351-8_20.
- [P433] Wu, S.-Y., Wu, K.-T. (2006), Effective Location Based Services with Dynamic Data Management in Mobile Environments, Wireless Networks, 12, (3), 369-381, DOI: 10.1007/s11276-005-5280-0.
- [P434] Wutzler, M., Weissbach, M., Springer, T. (2017), Role-Based Models for Building Adaptable Collaborative Smart Service Systems, 2017 IEEE International Conference on Smart Computing, SMARTCOMP 2017, Institute of Electrical and Electronics Engineers Inc., DOI: 10.1109/SMARTCOMP.2017.7947041.
- [P435] Xuwei, P., Li, Z. (2009), A service-oriented middleware architecture for building context-aware personalized information service, *International Symposium on Intelligent Ubiquitous Computing and Education*, 457-460, DOI: 10.1109/IUCE.2009.89.
- [P436] Yang, Y., Huang, Y., Ma, X., Lu, J. (2016), Enabling context-awareness by predicate detection in asynchronous environments, *IEEE Transactions on Computers*, 65, (2), 522-534, Association for Computing Machinery, Inc, DOI: 10.1109/TC.2015.2424879.
- [P437] Yau, J., Joy, M. (2007), A context-aware and adaptive learning schedule framework for supporting learners' daily routines, International Conference on Systems, DOI: 10.1109/ICONS.2007.3.
- [P438] Yau, S.S., Karim, F. (2004), An Adaptive Middleware for Context-Sensitive Communications for Real-Time Applications in Ubiquitous Computing Environments, *Real-Time Systems*, 26, (1), 29-61, DOI: 10.1023/B:TIME.0000009305.62647.ee.
- [P439] Yazidi, A., Granmo, O.C., Oommen, B.J., Gerdes, M., Reichert, F. (2011), A User-Centric Approach for Personalized Service Provisioning in Pervasive Environments, *Wireless Personal Communications*, 61, (3), 543-566, DOI: 10.1007/s11277-011-0387-3.
- [P440] Yen, C. K., Wang, K. S., Ko, R. S. (2014), AMDWAF: A context-aware web application framework, *International Conference on Information Science, Electronics and Electrical Engineering*, 1085-1089, DOI: 10.1109/InfoSEEE.2014.6947837.
- [P441] Yigitbas, E., Sauer, S. (2014), Flexible & adaptive UIs for self-service systems, Proceedings ABIS 2014 20th International Workshop on Adaptivity and User Modeling, Institute of Electrical and Electronics Engineers Inc..
- [P442] Yigitbas, E., Sauer, S., Engels, G. (2015), A model-based framework for multi-adaptive migratory user interfaces, *International Conference on Human-Computer Interaction*, 563-572, DOI: 10.1007/978-3-319-20916-6_52.
- [P443] Yilmaz, Ö., Erdur, R.C. (2012), IConAwa An intelligent context-aware system, *Expert Systems with Applications*, 39, (3), 2907-2918, DOI: 10.1016/j.eswa.2011.08.152.
- [P444] Yoon, C., Chon, Y., Cha, H. (2012), PION: Human mobility-based service provisioning framework for smartphone users, Communications Society Conference on Sensor, Mesh and Ad Hoc Communications and Networks, 174-181, DOI: 10.1109/SECON.2012.6275774.
- [P445] You, I., Choi, J., Choi, C., Kim, P. (2014), Intelligent healthcare service based on context inference using smart device, Soft Computing, 18, (12), 2577-2586, DOI: 10.1007/s00500-014-1420-8.
- [P446] Yu, J., Huang, Y., Cao, J., Tao, X. (2010), Middleware support for context-awareness in asynchronous pervasive computing environments, *International Conference on Embedded and Ubiquitous Computing*, 136-143, DOI: 10.1109/EUC.2010.29.
- [P447] Yu, J., Reddy, Y.V.R., Selliah, S., Bharadwaj, V., Reddy, S., Kankanahalli, S. (2005), The design of a workflow-centric, context-aware framework to support heterogeneous computing environments in collaboration, *International Conference on Cooperative Design, Visualization, and Engineering*, 22-29, DOI: 10.1007/11555223_3.
- [P448] Yu, Z., Zhou, X., Nakamura, Y. (2008), Semantic learning space: An infrastructure for context-aware ubiquitous learning, International Conference on Ubiquitous Intelligence and Computing, 131-142, DOI: 10.1007/978-3-540-69293-5_12.
- [P449] Yu, Z., Zhou, X., Yu, Z., Zhang, D., Chin, C.-Y. (2006), An OSGI-based infrastructure for context-aware multimedia services, *Communications Magazine*, 44, (10), 136-142, DOI: 10.1109/MCOM.2006.1710425.

- [P450] Yuan, B., Herbert, J. (2014), Context-aware hybrid reasoning framework for pervasive healthcare, *Personal and Ubiquitous Computing*, 18, (4), 865-881, DOI: 10.1007/s00779-013-0696-5.
- [P451] Yue, W., Wang, H., Wang, G. (2007), Designing transparent interaction for ubiquitous computing: Theory and application, International Conference on Human-Computer Interaction, 331-339.
- [P452] Zarikas, V. (2007), Modeling decisions under uncertainty in adaptive user interfaces, Universal Access in the Information Society, 6, (1), 87-101, DOI: 10.1007/s10209-007-0072-1.
- [P453] Zhang, B., Yin C., David, B., Chalon, R., Xiong, Z. (2016), A context-aware mobile system for work-based learning, *Computer Applications in Engineering Education*, 24, (2), 263-276, World Scientific Publishing Co. Pte Ltd, DOI: 10.1002/cae.21704.
- [P454] Zhang, L., Liu, Y., Zhan, X., Yang, X., Chi, X., Zhao, S. (2012), Campus View: An Innovative Location and Context-Aware Video Sharing Application on Smart Phone, *Wireless Personal Communications*, 66, (3), 493-509, DOI: 10.1007/s11277-012-0737-9.
- [P455] Zhang, S., McCullagh, P., Nugent, C., Zheng, H., Black, N. (2013), An ontological framework for activity monitoring and reminder reasoning in an assisted environment, *Journal of Ambient Intelligence and Humanized Computing*, 4, (2), 157-168, DOI: 10.1007/s12652-011-0063-1.
- [P456] Zhang, W., Guan, H., Li, M., Wu, M.-Y., Zhang, C., Tang, F. (2006), Context-aware adaptation for media delivery in pervasive computing environment, *International Conference on Grid and Pervasive Computing*, 634-643.
- [P457] Zhang, Z., Wang, N., Chen, Y., Lu, Y., Shi, Y., Wang, R., Lu, X. (2011), SREngine: An OSGi-based Context-aware inference engine for smart room, *International Conference on Pervasive Computing and Applications*, 267-271, DOI: 10.1109/ICPCA.2011.6106515.
- [P458] Zhao, M., Xiang, H., Chang, J. (2011), A context-aware architecture for intelligent home environment, *International Conference on Pervasive Computing and Applications*, 208-211, DOI: 10.1109/ICPCA.2011.6106506.
- [P459] Zheng, D., Jia, Y., Zhou, P., Han, W.-H. (2007), Context-aware middleware support for component based applications in pervasive computing, *International Workshop on Advanced Parallel Processing Technologies*, 161-171.
- [P460] Zheng, D., Wang, J., Kerong, B. (2014), Research of context-aware component adaptation model in pervasive environment, International Conference on Dependable, Autonomic and Secure Computing, 496-501, DOI: 10.1109/DASC.2014.95.
- [P461] Zhou, F., Jiao, J., Chen, S., Zhang, D. (2011), A case-driven ambient intelligence system for elderly in-home assistance applications, *Transactions on Systems, Man and Cybernetics*, 41, (2), 179-189, DOI: 10.1109/TSMCC.2010.2052456.
- [P462] Zhu, J., Ong, S. K., Nee, A. Y. C. (2013), An authorable context-aware augmented reality system to assist the maintenance technicians, *The International Journal of Advanced Manufacturing Technology*, 66, (9), 1699-1714, DOI: 10.1007/s00170-012-4451-2.
- [P463] Zimmermann, A., Lorenz, A. (2008), LISTEN: a user-adaptive audio-augmented museum guide, User Modeling and User-Adapted Interaction, 18, (5), 389-416, DOI: 10.1007/s11257-008-9049-x.